

Grose, Derek B. (2016) *Comorbidity in lung cancer: influence on treatment and survival.* MD thesis.

http://theses.gla.ac.uk/7079/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Glasgow Theses Service http://theses.gla.ac.uk/ theses@gla.ac.uk

# COMORBIDITY IN LUNG CANCER: INFLUENCE ON TREATMENT AND SURVIVAL

by

### DEREK B GROSE

MB.CHB, MRCP, FRCR

# SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF MEDICINE

to

### THE UNIVERSITY OF GLASGOW

February 2016

© Derek B Grose

## DEDICATION

I dedicate this thesis to my wife Pauline and to my children, Cameron and Emma, for their love and support.

### ACKNOWLEDGEMENTS

I owe a significant debt of gratitude to the following people for their help, advice and encouragement

Dr Robert Milroy, Department of Respiratory Medicine, Royal Infirmary, Glasgow

Professor Donald C McMillan, University Department of Surgery, Royal Infirmary, Glasgow

Dr David S Morrison, Department of Public Health, University of Glasgow

Professor Graham Devereux, Department of Respiratory Medicine, University of Aberdeen

The West of Scotland Lung Cancer Group for its support, both clinical and financial, of this research.

## DECLARATION

I declare that all work in this thesis was performed personally unless otherwise acknowledged.

### TABLE OF CONTENTS

Declaration	4
List of Tables	10
List of Figures	13
Publications	14
Abstract	16

### CHAPTER 1: INTRODUCTION

1.1	Lung cancer in Scotland	19
1.2	Incidence	20
1.3	Lung cancer risk factors	23
1.4	Presentation	24
1.5	Diagnosis	24
1.6	Staging	25
1.7	Treatment of NSCLC	28
1.8	Treatment of SCLC	29
1.9	Mortality	30
1.10	Survival	32

### CHAPTER 2: REVIEW OF THE CURRENT LITERATURE

2.1	Materials and methods	36
2.2	Tools available to assess comorbidity in cancer	36
2.3	Tools available to assess comorbidity specifically in lung cancer	39
2.4	Association between comorbidity and cancer survival	40
2.5	Association between comorbidity and lung cancer outcomes	40
2.6	Studies comparing comorbidity and survival in lung cancer	42
2.7	Conclusion	45

### **CHAPTER 3: METHODS**

3.1	Introduction	46
3.2	Data Collection	46
3.3	Socio-economic Status	47
3.4	Comorbidity Severity Scoring	48
3.5	Survival Data	51
3.6	Ethics	51

## CHAPTER 4: VARIATION IN COMORBIDITY AND CLINICAL MANAGEMENT IN NEWLY DIAGNOSED LUNG CANCER PATIENTS IN FOUR SCOTTISH CENTRES

4.1	Introduction	52
4.2	Methods	54
4.3	Results	55
4.4	Discussion	72

## CHAPTER 5: COMORBIDITIES IN LUNG CANCER: PREVALENCE, SEVERITY AND LINKS WITH SOCIO-ECONOMIC STATUS AND TREATMENT.

5.1	Introduction	77
5.2	Methods	79
5.3	Statistical Analysis	79
5.4	Results	80
5.5	Discussion	87

## CHAPTER 6: SIMPLE AND OBJECTIVE PREDICTION OF SURVIVAL IN PATIENTS WITH LUNG CANCER; STAGING THE HOST SYSTEMIC INFLAMMATORY RESPONSE.

6.1	Introduction	89
6.2	Methods	91
6.3	Statistics	92
6.4	Results	93
6.5	Discussion	109

## CHAPTER 7: THE IMPACT OF COMORBIDITY UPON DETERMINANTS OF OUTCOME IN PATIENTS WITH LUNG CANCER.

7.1	Introduction	112
7.2	Methods	115
7.3	Statistics	115
7.4	Results	116
7.5	Discussion	123

## CHAPTER 8: IMPACT OF THE TREATMENT DECISION PROCESS UPON SURVIVAL IN LUNG CANCER

8.1	Introduction	126
8.2	Methods	128
8.3	Statistics	128
8.4	Results	129
8.5	Discussion	139

### CHAPTER 9: CONCLUSION AND FUTURE WORK

9.1	Conclusions	141
9.2	Utilisation of the SCSS and Proposed Future Work	146

### REFERENCES

149

### **APPENDIX I**

10.1	Data Collection Proforma	170

### LIST OF TABLES

- Table 1.1International Lung Cancer Staging (6th edition)
- Table 1.2Stage Grouping
- Table 1.3 Age-Standardised One, Five and Ten Year Relative Survival Rates, Adults Aged 15-99, England 2005-2009, England and Wales 2007
- Table 3.1Comorbidities and severity scales assessed for each subject in<br/>the study
- Table 4.1Between centre differences in patient profile
- Table 4.2Between centre differences in investigation and treatment
- Table 4.3Results of univariable analysis relating performance status0/1 to centre
- Table 4.4 Results of univariable analysis relating decline in performance status from 0/1 to 2/3/4 in the 6 months prior to presentation to centre
- Table 4.5Results of multiple logistic regression relating decision to<br/>treat non-small cell lung cancer by surgical resection
- Table 4.6 Results of univariable analysis relating decision to treat nonsmall cell lung cancer by high dose radical radiotherapyto centre
- Table 4.7Results of univariable analysis decision to treat small cell<br/>lung cancer with combination chemotherapy or concurrent<br/>chemoradiotherapy to centre
- Table 4.8Results of univariable analysis relating decision to treat by<br/>best supportive/palliative care to centre

- Table 5.1Distribution of comorbidities among 868 newly diagnosedlung cancer patients
- Table 5.2Concurrence between individual comorbid conditions.
- Table 5.3Cumulative comorbidity scoring index.
- Table 5.4The relationship between comorbidity grouping,demographics, socio-economic status, stage and treatment.
- Table 6.1 The Glasgow Prognostic Scores
- Table 6.2The relationship between the mGPS and clinicopathological<br/>characteristics in patients with lung cancer
- Table 6.3The relationship between parameters and survival in patientswith mGPS = 0
- Table 6.4The relationship between parameters and survival inpatientswith mGPS = 1
- Table 6.5The relationship between parameters and survival in patientswith mGPS = 2
- Table 6.6The relationship between mGPS and PS and 12 month survivalrate
- Table 6.7The relationship between mGPS and TNM Stage (NSCLC only)and 12 month survival rate
- Table 6.8The Relationship between mGPS and PS and 3 month survivalrate in patients with TNM stage IIIb/IV NSCLC
- Table 6.9 The Relationship between mGPS and PS and 3 month survival rate in patients with TNM stage IIIb/IV NSCLC and receiving palliative chemotherapy
- Table 7.1Median Cumulative Comorbidity Score as determined by PSand mGPS in the early group (Stage I-IIIa NSCLC).
- Table 7.2Median Cumulative Comorbidity Score as determined by PSand mGPS in the advanced group.
- Table 7.3The relationship between parameters and survival inthe Early<br/>stage group (Stage I, II and IIIa NSCLC)

- Table 7.4The relationship between parameters and survival in theAdvanced group (Stage IIIb/IV NSCLC, SCLC and not staged)
- Table 8.1
   Baseline characteristics of patients with lung cancer
- Table 8.2Breakdown of determining factors for sub-optimal therapy in<br/>Stage I-IIIa NSCLC.
- Table 8.3Comparison of factors between those being offered radical<br/>and palliative therapy in Stage I-IIIa NSCLC.
- Table 8.4Breakdown of determining factors for sub-optimal therapyinStage IIIb/IV NSCLC + SCLC
- Table 8.5 Comparison of factors between those being offered chemotherapy and supportive care in Stage IIIb/IV NSCLC + SCLC.

### LIST OF FIGURES

- Figure 1.1 European Age-Standardised Incidence Rate per 100,000 Population, by Sex, Great Britain
- Figure 1.2 European Age-Standardised Incidence Rates by Deprivation Quintile, England
- Figure 1.3 European Age-Standardised Mortality Rates by Deprivation Quintile, England
- Figure 1.4 One-Year Relative Survival (%) by Stage, Adults 15-99, Former Anglia Cancer Network
- Figure 1.5 Five-Year Relative Survival (%) by Stage, Adults 15-99, Former Anglia Cancer Network
- Figure 3.1 Flowchart of patient numbers in Study
- Figure 4.1 Bar chart illustrating between centre differences in severity scores for (a) chronic obstructive pulmonary disease (COPD) and (b) ischaemic heart disease (IHD).
- Figure 5.1 Histogram showing number of comorbid conditions per patient. 868 lung cancer patients in 4 Scottish centres, 2005-08.
- Figure 6.1 Flow chart of patient selection process
- Figure 6.2 The relationship between the Glasgow Prognostic Score (GPS) and survival.
- Figure 6.3 The relationship between the modified Glasgow Prognostic Score (mGPS) and survival.

### PUBLICATIONS

The work presented in this thesis has resulted in the following publications:

Grose D, Devereux G, Brown L, Jones R, Sharma D, Selby C, Morrison DS, Docherty K, McIntosh D, Louden G, Downer P, Nicolson M, Milroy R. Variation in comorbidity and clinical management in patients newly diagnosed with lung cancer in four Scottish centres. J ThoracOncol. 2011 Mar;6(3):500-9

Grose D, Devereux G, Milroy R. Comorbidity in lung cancer: important but neglected. A review of the current literature.Clin Lung Cancer. 2011 Jul;12(4):207-11

Grose D, Milroy R. Chronic obstructive pulmonary disease: a complex comorbidity of lung cancer. Jrnl Comorbidity 2011;1:45-50

Grose D, Devereux G, Brown L, Jones R, Sharma D, Selby C, Morrison DS, Docherty K, McIntosh D, Louden G, McElhinney P, Nicolson M, McMillan DC Milroy R.Simple and objective prediction of survival in patients with lung cancer; staging the host systemic inflammatory response. Lung Ca Int 2014, Article ID 731925

Derek Grose, David S Morrison, Graham Devereux, Richard Jones, Dave Sharma, Colin Selby, Kirsty Docherty, David McIntosh, Greig Louden, Marianne Nicolson, Donald C McMillan, Robert Milroy, on behalf of the Scottish Lung Cancer Forum. Comorbidities in lung cancer: prevalence, severity and links with socioeconomic status and treatment. Postgrad Med J 2014;90:1064 305-310 Grose D, Morrison DS, Devereux G, Jones R, Sharma D, Selby C, Docherty K, McIntosh D, Nicolson M, McMillan DC, Milroy R; Scottish Lung Cancer Forum. The impact of comorbidity upon determinants of outcome in patients with lung cancer. Lung Cancer. 2015 Feb;87(2):186-92.

#### ABSTRACT

Lung cancer is the commonest cancer in Scotland and survival rates for patients in Scotland appear lower than in many other European countries. Although this variation in survival is usually interpreted as evidence of variation in facilities, access to care and clinical practice it is possible that the increased comorbidity and poor performance status of the Scottish population may contribute to the observed disparities in treatment and outcomes, although this has never been proven.

The overall aim of the Thesis was to examine the impact of comorbidity in lung cancer, to attempt to quantify the extent and severity of comorbidity and to explore its relationship with treatment and survival.

Between 2005 and 2008 all newly diagnosed lung cancer patients coming through the Multi-Disciplinary Teams (MDTs) in four Scottish Centres were included in the demographics, study. Patient World Health Organization/Eastern Cooperative Oncology Group performance status (PS), clinic-pathological features, stage, comorbidity, markers of systemic inflammation and proposed primary treatment modality were all recorded. Information on date of death was obtained via survival analysis undertaken by the Information Service Division (ISD) of NHS Scotland. Death records were complete until 1 June 2011, which served as the censor date for those alive.

Chapter 4 examines the variations in demographics and baseline characteristics seen between the centres and reveals significant differences between the centres such as deprivation, stage at presentation, PS and treatments offered.

Chapter 5 explores the relationship between comorbidity and the patient cohort. It shows that comorbidity can be quantified using a scoring index (the Scottish Comorbidity Scoring System (SCSS)) and that increasing

16

comorbidity is associated with treatment centre and socio-economic status, with the most deprived patients having increased levels of comorbidity. It also demonstrates that comorbidity appears to have an impact on treatment offered.

Chapter 6 examines the relationship between systemic inflammation (utilizing the well established modified Glasgow Prognostic Score (mGPS)) and outcome in the patient cohort. It confirms previous work supporting the use of the mGPS in predicting lung cancer survival and also shows how it might be used to provide more objective risk stratification in patients diagnosed with lung cancer.

Chapter 7 explores the relationship between a novel comorbidity scoring system (SCSS) and the already established Charlson Comorbidity Index (CCI) and the modified Glasgow Prognostic Score (mGPS). This study aimed to determine which of these factors provided the most accurate information on survival. The novel comorbidity scoring system, the SCSS compares very favourably with the more established CCI. In addition this study demonstrates clear differences between patients having potentially radically treatable disease (NSCLC stage I - IIIa) and disease which would generally be considered incurable (NSCLC IIIb/IV and SCLC).

Chapter 8 examines the reasons for the clinician decision-making process and if these reasons do indeed mirror the individual patient's demographics, fitness and stage. In the majority of patients, both in the early and advanced stage at presentation, the treatment decision appears to be appropriate given the recorded fitness, PS and comorbidity. However in a small but significant number of patients there did appear to be discrepancies between the clinician's reasons for sub-optimal therapy and the recorded objective assessment of the patient in question.

The work presented in this thesis has demonstrated the significant extent of comorbidity in lung cancer and the important role it appears to play (along with systemic inflammation) in determining treatment choice and survival.

### CHAPTER 1: INTRODUCTION

#### 1.1 Lung Cancer in Scotland

Lung cancer is the commonest cancer in Scotland and the second commonest in the UK (Office for National Statistics, 2006; ISD, 2009). In 2011 5,096 cases of lung cancer were recorded in Scotland. 2,601 were in males with 2,495 being in females. The incidence rate in Scotland was 97.0 per 100,000 in comparison to 68.7 per 100,000 across the UK (ISD Scotland, 2013). The incidence rates in Scotland remain amongst the highest in the world (Parkin et al., 2002) reflecting the history of high smoking prevalence.

Treatment and survival rates for patients with lung cancer in Scotland appear lower than in many other European countries. Five-year survival is quoted at 6-7% (Gregor et al., 2001) compared with 8-15% in other European countries and America (Janssen-Heijnen et al., 1998). Erridge et al showed significant differences in survival between Scotland and British Columbia with 2 yr. OS being 10% vs. 23% (Erridge et al., 2009). There also appear to be variations in treatment rates within Scotland (Gregor et al., 2001).

Although this variation in treatment and survival is usually interpreted as evidence of variation in facilities, access to care and clinical practice it is possible that the increased co-morbidity and poor performance status of the Scottish population may contribute to the observed disparities in treatment and outcomes, although this has never been proven. In a recent Lung Cancer paper Erridge et al suggesting that suboptimal treatment is not the only reason for poor outcome, even when treatment rates were taken into account, also showed a significant hazard ratio for death of 1.5. The majority of patients with lung cancer will have smoked and will, in general, have a less favourable socio-economic status, two factors known to be associated with an increased likelihood of co-morbid conditions e.g. Chronic Obstructive Pulmonary Disease (COPD), Ischaemic Heart Disease, that influence treatment options. Co-existent comorbidities may also have an adverse effect on the rate of deterioration of performance status when lung cancer develops, with obvious consequence for treatment options and survival. The demonstration that comorbidity influences treatment and survival rates will have important implications for the targeting of health services resources, screening, interpretation of cancer statistics and the assessment and management of patients with lung cancer.

#### 1.2 Incidence

Lung cancer incidence is strongly related to age. In the UK more than four in ten cases are in those aged 75+ (Office for National Statistics, 2013; ISD Scotland, 2013).

Male lung cancer incidence rates have gradually decreased over the last twenty years. Unfortunately the rates for females continue to rise indicating the changing frequency of smoking between the sexes (Office for National Statistics, 2013;ISD, 2013)



Year of Diagnosis

Figure 1.1 European Age-Standardised Incidence Rates per 100,000 Population, by Sex, Great Britain

There also appears to be an association between lung cancer incidence and deprivation. Data from England in 2006 -2010 shows significantly higher rates for those living in the most deprived areas compared to the least (CRUK, 2012).



Figure 1.2 European Age-Standardised Incidence Rates by Deprivation Quintile, England

#### 1.3 Lung cancer risk factors

Smoking remains the commonest cause of lung cancer with an estimated 86% of lung cancer deaths in the UK being caused by tobacco smoking (Parkin et al., 2011). In addition lung cancer death risk is around 15 times higher in current smokers compared with never-smokers. (Doll et al., 2005)

An estimated 0.5% of lung cancer in the UK is linked to radon gas exposure. (Parkin et al., 2011). Approximately 1% of the total is linked to previous radiation exposure (either radiotherapy or diagnostic radiation) (Parkin et al., 2011)

A number of occupational exposures are classified as causes of lung cancer. This is seen in an estimated 21% of men with lung cancer and 5% of women. This is commonest in occupations linked to asbestos exposure (typically construction and shipyard workers) with an estimated 6-8% of all lung cancer in the UK being associated with asbestos exposure. (Brown et al., 2012).

Lung cancer risk is increased in survivors of several cancers. In particularly previous breast cancer (Lorigan et al., 2010; Maddams et al 2011) and Hodgkin's lymphoma (Ibrahim et al., 2013). This risk is related to previous thoracic radiotherapy.

#### 1.4 Presentation

Lung cancer often produces very few symptoms until the disease is well advanced (over 50% will have incurable disease at diagnosis (NLCA, 2013). In addition lung cancer can present as an incidental finding on chest radiological imaging undertaken for other reasons. (Spiro et al., 2007). When lung cancer presents in this way it is more often diagnosed at an early and potentially curable stage).

Cough is the most commonly reported symptom along with dyspnoea, chest pain and haempotysis due to chest disease. Non-specific symptoms such as weight loss, fatigue and cachexia are commonly seen due to metastatic disease. Specific symptoms due to site of metastatic spread such as headaches with intracranial deposits and localised pain due to bone metastasis are also commonly seen.

#### 1.5 Diagnosis

A chest x-ray is the first investigation for all patients being investigated for the possibility of lung cancer (Detterbeck et al., 2001). A contrast enhanced CT scan of the chest and abdomen is recommended in all suspected lung cancer cases, regardless of CXR findings (SIGN 137). Results from CT scanning are variable but a high sensitivity and relatively lower specificity is the norm (Yankelevitz et al., 1999). This should be performed prior to further diagnostic investigation, including bronchoscopy, as the result of the CT scan is likely to help guide the investigation most likely to provide a diagnosis (Laroche et al., 2000).

If not contraindicated either by patient fitness or wishes then a tissue diagnosis should be obtained both to confirm diagnosis and help determine most appropriate treatment (SIGN 137). Bronchoscopy has a good diagnostic sensitivity for central lesions (Schreiber et al., 2003). However this is lower for peripheral lesions (Schreiber et al., 2003) in which case percutaneous biopsy should be considered (Detterbeck et al., 2001).

All pathological samples should be, if possible, classified as Small Cell Lung Cancer (SCLC) or Non Small Cell Lung Cancer (NSCLC). NSCLC tumours should then be subtyped and genetic mutations for EGFR and ALK examined for (SIGN 137).

#### 1.6 Staging

Lung cancer is currently staged using the 7<sup>th</sup> edition of TNM staging system, which covers NSCLC and SCLC. At the time of data collection of this research the 6<sup>th</sup> edition of TNM was the current version and for all future references during this work it will be the 6<sup>th</sup>edition, which will be used. It is to be acknowledged that the TNM staging update, in particular the refined use of tumour size, may make a potential difference to interpretation of the results. However the data recorded would not allow for formal "restaging" of patients.

## Table 1.1International Lung Cancer Staging (6th edition)

T1	≤3cm, surrounded by lung or visceral pleura, without bronchscopic evidence of	
	invasion of more proximal than lobar bronchus	
T2	Any of the following features:	
	≥3cm	
	involves main bronchus. 2cm or more distal to the carina	
	invades visceral pleura	
	associated with atelectoric or obstructive preumonitis that extends to the hilar	
	radion but door not involve the entire lung	
	region but does not involve the entire tung	
T2	Tumour of any cize that directly invades any of the following:	
15	short well	
	chest wall	
	diaphragm	
	mediastinal pleura	
	parietal pericardium	
	tumour in main bronchus <2cm distal to carina	
	associated with atelectasis or obstructive pneumonitis that involves the entire	
	lung	
T4	Tumour of any size that invades any of the following:	
	mediastinum	
	heart	
	great vessels	
	trachea	
	oesophagus	
	vertebral bod	
	carina	
	Or separate tumour nodules in the same lobe	
	Or tymour with malignant pleural effusion	
NO	Na involved lymph nodes	
NU	No involved tymph hodes.	
N1	Involved ipsilateralhilar lymph nodes.	
N2	Involved mediastinal lymph nodes.	
N3	Involved contralateral lymph nodes.	
M1	Distant metastasis present	

T = Primary tumour, N = Regional Lymph Nodes, M = Distant Metastasis

Stage	ТИМ
la	Τ1
lb	T2
lla	T1 N1
IIb	T2 N1 or T3 N0
Illa	T1 N2, T2 N2, T3 N1 or T3 N2
IIIb	Any T4 or any N3
IV	Any M1

Table 1.2 Stage Grouping

CT scanning is poor at predicting more advanced T staging e.g. T3/4 (Detterbeck et al., 2001). FDG PET-CT is more accurate in detecting mediastinal nodal metastases in patients with NSCLC (De Leyn et al., 2007). In addition FDG PET-CT has been shown to identify unsuspected metastases in 10-15% of patients with NSCLC (Pieterman et al., 2000). As such, all patients with NSCLC being considered for radical treatment should have a staging FDG PET-CT.

The majority of patients have advanced stage disease (NSCLC IIIb/IV or SCLC) at time of diagnosis (national lung cancer audit report 2012). In the most recent national lung cancer audit 12% of new lung cancer cases were classified as SCLC. The remaining cases were either NSCLC (87%) or carcinoid (1%). Of the different subtypes of NSCLC squamous (35%) and adenocarcinoma (30-40%) are the most common.

#### 1.7 Treatment of NSCLC

Treatment options are dependent upon pathology, TNM staging, patient fitness and choice. Radical (i.e. potentially curable) treatment options are surgery or radical radiotherapy (+/- chemotherapy).

#### Surgery

Patients with Stage I + II NSCLC should be considered for curative surgery whenever possible as it confers the highest chance of cure. 5 Year survival of 54-80% for patients with St 1a and 38-65% for patients with Stage 1b has been reported (Suzuki et al., 1999; Makitaro et al., 2002; Inoue et al., 1998). In patients with good performance status (PS 0/1) that have completely resected NSCLC (stage II to IIIa) should be offered platinum based postoperative systemic anticancer therapy (NSCLC Meta-analyses Collaborative Group. 2010).

#### Radical Radiotherapy

In patients with localized NSCLC not fit / suitable for surgery, radical radiotherapy (+/- chemotherapy) is an appropriate option. Meta-analysis of retrospective studies in patients with Stage I/II has shown overall survival from 50-93% at one year and 0-42% at 5 years (Rowell et al., 2004). In recent years the option of Stereotactic Ablative Radiotherapy has become a very reasonable alternative for those with early stage disease and in many institutions has replaced conventional radiotherapy in this setting. (Palma et al., 2011; Lagerwald FJ., 2012)

#### Palliative Chemotherapy

In patients with advanced disease and a PS <2 then systemic chemotherapy with a platinum based doublet has been shown to confer a survival benefit of approximately 2 - 4 months (Goffin et al., 2010). Recent advances including tyrosine kinase inhibitors (in those with an EGFR mutation) (Rosell et al., 2012) and pemetrexed/platinum doublets in adenocarcinoma (Scagliotti et al., 2008) have shown further improvement in survival but these treatments were not routinely available at the time of this research.

#### Symptomatic Therapy

Palliative radiotherapy to areas causing symptoms such as pain is very effective at reducing such symptoms. In addition all patients should be considered for referral to specialist palliative care services (SIGN 137). There has also been some suggestion of a modest improvement in survival with early integrated palliative care input (Temel et al., 2010)

#### 1.8 <u>Treatment of SCLC</u>

#### Limited disease SCLC

Systemic chemotherapy with a platinum based doublet in patients of good PS has been shown to confer a survival benefit over single agent therapy (Simon et al., 2003). Radiotherapy delivered either concurrently or sequentially to the primary site appears to derive a benefit in survival, although absolute figures are difficult to quantify (De Ruysscher et al., 2000). In patients who have achieved remission after systemic chemotherapy, prophylactic cranial irradiation should be considered as it has been shown to confer a survival benefit (Arrigada et al., 2002; Auperin et al., 1999).

However even with optimal combination therapy two year survival is only around 25% (SIGN)

#### Extensive disease SCLC

Durable responses to chemotherapy are seldom seen. However there is often a useful symptomatic response (Girling. 1996).

#### 1.9 Mortality

Lung cancer accounts for 6% of all deaths in the UK and is also the commonest cause of cancer death in the UK (2011), accounting for 22% of all cancer deaths. It is the commonest cause of death for both males (23%) and females (21%). Lung cancer is also the most common cause of cancer death worldwide, with around 1,590,000 deaths from lung cancer in 2012 (19% total) (Office for National statistics, 2012).

In Scotland there were 4,178 deaths due to lung cancer in 2011. 2,200 deaths were in males with 1,978 in females. The European agestandardised mortality rates (AS rates) are significantly higher in Scotland compared with the remainder of the UK. In Scotland they are 53.1 / 100,000 compared to 38.2 / 100,000 for the UK (ISD Scotland, 2013).

Data from the UK also appears to show an association between mortality and deprivation with European AS rates being approximately 170% higher in those living in the most deprived areas compared to the least deprived (CRUK, 2013).



Figure 1.3 European Age-Standardised Mortality Rates by Deprivation Quintile, England

#### 1.10 <u>Survival</u>

Lung cancer has one of the lowest survival outcomes of any cancer because the majority of patients are diagnosed when the disease is already advanced and there is no curative option. In addition, the majority of patients are older (approx. 70 years) and have high incidence of co morbidities (NLCA Report, 2005).

Current 1 year survival in England is 29.4% for men and 33% in woman. This falls to 7.8% for men and 9.3% at 5 years respectively (Office National Statistics, 2010; ISD, 2011).

Table 1.3Age-Standardised One, Five and Ten Year Relative SurvivalRates, Adults Aged 15-99, England 2005-2009, England and Wales 2007

	Relative Survival (%)		
	1 Year	5 Year	10 Year
Sex	2005-2009	2005-2009*	2007**
Male	29.4	7.8	4.9
Female	33	9.3	5.9

#### By age

Lung cancer survival has been show to worsen as age increases for both men and women. For example five-year relative survival rates for men in England during 2005-2009 ranged from 35% in those diagnosed before 40 to 3% in those over 80.

#### By Stage

The majority of patients are diagnosed at an advanced stage with two thirds of NSCLC having either stage IIIb / IV (ISD Scotland). One and five year survival is strongly linked to stage at diagnosis. In NSCLC the one year survival varied from 71% in patients with stage I to 14% in those with stage IV. Five year survival was 35% and 1% respectively.



Figure 1.4 One-Year Relative Survival (%) by Stage, Adults 15-99, Former Anglia Cancer Network



Figure 1.5 Five-Year Relative Survival (%) by Stage, Adults 15-99, Former Anglia Cancer Network

#### **By Deprivation**

Analyses of lung cancer survival rates by socio-economic status in England and Wales in the late 1990s has shown a small but significant difference of 1.4% between men in the most and least affluent groups (Coleman et al., 2004). Lung cancer is the commonest cancer in Scotland and survival rates for patients in Scotland appear lower than in many other European countries. Although this variation in survival is usually interpreted as evidence of variation in facilities, access to care and clinical practice it is possible that the increased comorbidity and poor performance status of the Scottish population may contribute to the observed disparities in treatment and outcomes, although this has never been proven.

The overall aim of the Thesis was to examine the impact of comorbidity in lung cancer, to attempt to quantify the extent and severity of comorbidity and to explore its relationship with treatment and survival.
#### CHAPTER 2: REVIEW OF THE LITERATURE

#### 2.1 <u>Materials and Methods</u>

Literature searches were undertaken via PubMed and Google Scholar using various arrays of the following keywords; Cancer, Comorbidity, Lung Cancer, Performance Status, Survival and Tools. The search was limited to published articles in peer-review journals with English as the language. No Abstract or posters were included in the review. Results were separated into several sections. Tools available to assess comorbidity, association between comorbidity and outcome and finally the impact upon survival. The search was completed in March 2010. An extension of the literature review to include up until December 2015 and how recent papers relate to the work in this thesis is presented in Chapter 9.

#### 2.2 Tools available to assess co-morbidity in cancer

The evaluation of comorbidity in Lung Cancer is a relatively new area of research. In recent years a number of studies have attempted to assess the impact of comorbidity, performance status and age on treatment and survival in lung cancer. A variety of tools have been used to quantify comorbidity primarily in breast and prostate cancer (Extermann. 2000) although none of these have been validated in lung cancer. In addition the majority of lung cancer patients are from lower socio-economic classes with significant factors which may influence ability to deliver curative treatment e.g. Chronic Obstructive Pulmonary Disease and Ischaemic Heart Disease. The biology of lung cancer is obviously significantly different from the afore-mentioned tumour types in terms of treatment options, stage at time of presentation and median survival.

Extermann reviewed 4 tools currently in place. Although these have all been validated against cancer outcomes, none have been specifically validated in lung cancer.

The most widely quoted tool is the Charlson Comorbidity Index (Charlson et al., 1987) (CCI). This was designed in 1987 and assigned 19 conditions with a weighting index of 1 - 6 in attempt to quantify the likelihood of impact upon survival. Data was acquired via patients being admitted with medical conditions to a Washington Hospital. This tool was initially validated in breast cancer patients with 10-year mortality as an endpoint. It has also been validated in predicting progression-free survival (Chen et al., 1999) in a variety of diseases such as breast and prostate cancer. However limitations with using this tool for lung cancer patients include absence of some potentially relevant diseases such as pulmonary fibrosis, the weighting of HIV is probably now less significant due to improved treatment and the lack of grading of severity of the specific disease(s).

The second method reviewed was the Cumulative Illness Rating Scale (CIRS). Designed in 1968 by Linn et al, this scale scores 14 organ systems with severity between 0-4. Since its conception it has been modified on a number of occasions and recently modified to cirs-geriatric that is specifically designed to assess geriatric oncology patients. It has been validated previously in elderly cancer patients using mortality rates as an end-point. Due to the large number of comorbidities it becomes more unwieldy and therefore has not been used widely in clinical practice. There is no published validation of the CIRS in lung cancer.

The Index of Coexistent Disease (ICED) was developed by Greenfield et al in 1987. Two subscales attempt to measure physical and functional comorbid status. The physical subscale incorporates 14 categories with grade of severity 0-4. The functional subscale has 12 domains each rated 0-2. Overall severity is then summarised 0-3. This is the first such system to attempt to gain an understanding of the functional impact of comorbidity rather than relying on physiological results and as such has much to recommend it. The ICED has been used in cancer studies with good validation but not in lung cancer (Bennett et al., 1991; Guadagnoli et al., 1997) In addition this is a very complicated system and it is unlikely that the information could easily be collected at the time of the decision making process. Thus its greatest attribute is possibly its major failing.

The Kaplan-Feinstein Index (Kaplan et al., 1974) was developed in 1974 and comprises 12 categories graded 0-3. It has been shown to be predictive of outcome in prostate and head and neck cancer (Clemens et al., 1986; Piccirillo, 1995) but not validated in lung cancer.

A more recent tool validated in head and neck cancer patients was the Adult Co-morbidity Evaluation 27 (ACE-27) (Piccirillo et al., 2004). This graded 27 specific conditions from mild to severe. A prospective cohort study including over 17,000 patients with all major cancer types was undertaken. This showed that worsening comorbidity significantly impacted negatively on survival. Unfortunately sub-group analysis identified comorbidity as having least prognostic information in lung cancer patients. A clear reason for this was not identified. It may be related to the finding that comorbidity had greatest impact on survival in indolent cancers with long survival time. This is an interesting view point but fails to address the affect that comorbidity has upon ability to deliver radical treatment.

The simplest tool, superficially at least, is the American Society of Anesthesiologists (ASA), which initially used a five-category physical status classification system; a sixth category was later added (ASA Relative value Guide. 2002). This system was used for assessing the fitness of a patient prior to surgery. It ranged from a healthy patient with no systemic illness to a moribund state. It has been used in assessing comorbidity and is simple to use (Riechelmann et al., 2006). Limitations of the system include the transferability of a fitness for surgery scale to patients with lung

cancer, for the majority of whom surgery is not a therapeutic option. In addition there is a well documented inter-user variability and lack of clear guidance for how to score a variety of significant diseases (Haynes et al., 1995)

None of the available tools for assessing comorbidity have been designed with lung cancer as the main focus of data collection. Thus researchers are faced with difficulty in identifying which system to use in studies of comorbidity in lung cancer patients.

#### 2.3 Tools available to assess comorbidity specifically in lung cancer

Currently, there is only one tool that has been validated in lung cancer. Colinet et al in 2005 studied the outcome in Non Small Cell Lung Cancer (NSCLC) within 5 French territorial divisions. A prospective study of 735 consecutive patients between 1998-2003 evaluated treatment and They used a Simplified Comorbidity Score (SCS) and also prognosis. assessed Performance Status (PS), weight loss, age and the Charlson Comorbidity Index (CCI). It was shown that worsening stage, PS and SCS (>9) or CCI (>5) were independent adverse prognostic factors in terms of survival. Hazard ratio was 1.36 [1.09-1.69] Criticisms of this tool include the fact that SCS does not give any grading of disease severity or allow for more than one disease within a group e.g. a patient with both a stroke and ischaemic heart disease. In addition the reason for the extremely high weighting given to diabetes mellitus and smoking in comparison to respiratory and cardiac comorbidity is not made clear by the authors. Our own (published in abstract only) work (Milroy et al., 2005; Milroy et al., 2009) suggests that diabetes mellitus has minimal if any impact on the treatment of patients with lung cancer. However it is certainly a straightforward tool to use in clinical practice.

#### 2.4 Association between comorbidity and cancer survival

The possible prognostic impact of comorbidity on cancer patient survival was previously studied by Read et al (Read et al., 2004). A prospective cohort study was undertaken using the local cancer registry and 11,558 patients with a variety of cancers were identified. The severity of patient comorbidity was assessed using the Adult Co-morbidity Evaluation 27 (Piccirillo. 1995). The 1 year survival was determined for each tumour site and stage combination. The results suggested that the impact of comorbidity was greatest among tumour types with the longest survival i.e. those with the most indolent course such as prostate cancer. This suggests that comorbidity should have less effect on treatment or survival in aggressive tumour types such as lung cancer.

These findings are not to be questioned. However this study was primarily evaluating inter-tumour type variations in survival. It did not compare survival within a heterogeneous group of lung cancer patients or attempt to identify causes for such variations.

#### 2.5 Association between comorbidity and lung cancer outcome

In most cancer trials significant comorbidity is an exclusion criteria. Consequently trial data may be of limited relevance to every day clinical practice with patients, the majority of whom will have co-existing comorbidities, and would therefore have been excluded from the original trial upon which their proposed treatment may be based. It is therefore essential to assess what, if any, relationship has been shown in the literature between comorbidity and outcome in lung cancer. Although sparse, a number of studies have suggested that comorbidity has a significant impact on both treatment and outcome in lung cancer. However these studies suffer from a number of limitations. Potosky et al (Potosky et al., 2004) retrospectively evaluated 898 patients with Non-small Cell Lung Cancer (NSCLC) and variations in initial treatment. Using National Guidelines, recommended treatment strategies were determined. Information that might influence treatment was recorded. This included stage, age, and comorbidity (using the Charlson Comorbidity Index, smoking status and race. Overall only 52% of patients received recommended therapy. The use of recommended therapy declined with increasing age and tumour stage. There was also a reported detrimental association with increasing comorbidity. This would be in keeping with most general clinical experience although there has been criticism of the effect of co-morbidity upon treatment in at least one previous article (Janssen-Heijnen et al., 2004) which is described in the next section

A previous study by Earle (Earle et al., 2000) evaluated chemotherapy utilisation for patients with stage IV NSCLC and demonstrated that younger patients and those with fewer comorbid conditions were more likely to receive chemotherapy. It was however a retrospective analysis.

An analysis from Wisconsin, USA(Firat et al., 2002) showed that KPS and comorbidity (recorded yes/no) were important independent prognostic factors in Stage III NSCLC. The limitations of this study include the focus on one tumour stage, and the fact it was a retrospective analysis of four Radiation Therapy Oncology Group studies evaluating survival benefit with radiotherapy alone. It was not a prospective study and did not have comorbidity as a primary explanatory variable.

Kates et al (2009) recently retrospectively reviewed their localised lung cancer patients (Stage I - IIIA) and attempted to predict those likely to suffer from perioperative mortality (POM). It was shown that risk of death was significantly increased with a Charlson Comorbidity Index Score of greater than 4. Acute Myocardial Infarction, perhaps not surprisingly, had

the strongest association. Other factors associated with increased POM included increasing age, male gender and extent of surgical resection.

Thus there is an indication that comorbidity has a relevance to management decisions in lung cancer. However, these studies did not explore actual survival.

#### 2.6 Studies comparing comorbidity and survival in lung cancer

A number of articles have attempted to clarify the impact of comorbidity on lung cancer survival.

Tammemagi et al (2004) evaluated comorbidity and its association with smoking on lung cancer survival. One thousand one hundred and fifty five patients who were diagnosed with lung cancer between 1995 and 1998 were identified and retrospective analysis was performed to assess the impact of smoking on survival. It was shown that current smokers' hazard ratio for death was significantly increased compared to former/never smokers 1.37 (95% confidence interval 1.18-1.59; p<0.001). Sixteen significant comorbid conditions were recorded. Sub-group analysis of these was also performed. A number of these comorbid factors were found to be independent predictors of survival. However, it should be noted that this was not the primary aim of the study and it was retrospective.

Janssen-Heijnen (Janssen-Heijnen et al., 2004) evaluated 4072 patients with NSCLC. The authors assessed a number of comorbidities using a very slightly adapted version of the Charlson Comorbidity Index along with age, tumour size and proposed treatment. This study was, however, retrospective and used a comorbidity index not validated in lung cancer. Univariate and multivariate analyses of survival were performed. One year survival decreased significantly with age (p<0.0001). Comorbidity was not

42

found to have any influence on survival; however comorbidity was recorded as actual number of diseases ranging from 0 to 2 with no attempt to quantify the severity of comorbidity.

Birim et al (2005) reported the assessment of the impact of the Charlson Comorbidity Index (CCI) in patients with NSCLC treated with surgery and the impact of the CCI score on survival. With multivariate analysis it was shown that for a CCI score >3 the relative risk of death was 2.2 (95% CI 1.5-3.1). However with a score of 1 or 2 the relative risk was 1.4 (95% CI1.0-1.8). This showed a significant negative correlation on overall survival, which is decreased with increasing CCI score, suggesting that worsening comorbidity did impact negatively on survival. Limitations of this study include the absence of a grade of disease severity and that the study period was rather prolonged. In addition the authors failed to assess early stage NSCLC patients deemed ineligible for surgery.

Imperatori et al (2006) compared management and survival between Teeside (UK), which has very similar population demographics to Scotland, and Varese (Italy), which does not. Data were recorded prospectively. The study revealed that Teeside had statistically significantly more advanced disease, higher comorbidity at presentation and lower survival rates. Comorbidity was recorded simply as yes/no with no grading of severity. It did not show any impact on survival. Only surgical resection was shown in multivariate analysis to be a positive predictor of survival (HR 0.46; p= 0.0016). The overall 2 year survival was 14% for Italy and 7% in Teeside (p< 0.001). Interestingly survival was relatively low by any European standard in both Italy and England (Janssen-Heijnen et al., 1998)

Asmis et al (2008) retrospectively assessed two other studies recently undertaken by the National Cancer Institute of Canada Clinical Trials Group. The paper by Asmis evaluated age, comorbidity and performance status as factors that might influence survival. The Charlson Comorbidity Index assessed comorbidity. A CCI score of 1 was associated with a poorer overall survival compared to 0 (hazard ratio [HR 1.28; 95% CI 1.09-1.5; p= 0.003). For a CCI >1 there was no additional effect seen. It must be noted that very few patients fell into the category of having CCI>1. Age was not shown to be an independent factor. As with most clinical trials these did not include unfit, poor performance status patients. It must also be noted that the comorbidity data was collected retrospectively. This was a highly selected group of patients as only relatively fit, good performance status patients were eligible for trial enrolment. In addition previous studies have shown that higher social class patients are more likely to be offered trial enrolment (Mosenifar et al., 2007). It would, therefore, be inappropriate to make significant judgement about the impact of comorbidity based on this report alone.

#### 2.7 Conclusion

Lung cancer is a major cause of death and imposes a significant burden on health care systems throughout the developed world. Nearly 40,000 new cases were reported in 2008 in the UK (Ferlay et al., 2007). Furthermore, with the increase in smoking and increasing life expectancy in the developing world, lung cancer is likely to place an increasing burden on the health services of developing countries in the future.

Factors such as age, performance status and social class are often seen as correlates for comorbidity. Therefore any attempts to both evaluate and quantify comorbidity will require taking into account these additional factors if we are to determine the underlying reasons for the poor prognosis in lung cancer and regional variations in survival.

At present there is little published work in the field of comorbidity in lung cancer. None of the previous comorbidity tools were designed specifically for lung cancer, with the exception of the study by Colinet et al and most have been adapted from other uses. No published studies have definitively demonstrated and prospectively quantified the impact of comorbidity on treatment decision making and on overall survival. There is a clear need for further prospective studies to attempt to clarify further the impact of comorbidity in lung cancer patients. Finally a specific tool designed for, and validated in the lung cancer setting will be essential if we are to truly understand the multi-factorial nature of variations in treatment and survival within the lung cancer population.

#### CHAPTER 3: METHODS

#### 3.1 Introduction

Four Scottish centres were included in the study. These non-tertiary centres routinely investigate, diagnose and treat patients with lung cancer who live in demographically contrasting areas of Scotland (Grose et al., 2011). Healthcare coverage in Scotland is universal, free at the point of need and highly centralised. National guidelines state that all newly diagnosed lung cancer patients should be referred to a Respiratory Physician and discussed at a Multi-Disciplinary Meeting (MDT) (SIGN, 2005). All four centres involved in this study adhere to this with a greater than 90% success (Managed Clinical Network audit report, WOSCAN, 2011) with the majority of referrals being made by General Practitioners or via unscheduled acute hospital admission with a minority from radiologists and surgeons. This ensured that patients included in the study would be representative of the underlying lung cancer population. Consecutive patients diagnosed with lung cancer were included in the study but because of local factors the centres did not recruit for the same periods of time. The study periods for each centre were Aberdeen, October 2005 to February 2007; Stobhill, December 2005 to April 2008; Inverclyde, October 2005 to December 2007 and Dunfermline, June 2007 to April 2008. All newly diagnosed lung cancer patients, whose care was discussed at the Multi-Disciplinary Team (MDT) meeting, were included in the study.

#### 3.2 Data Collection

At the time of the patient's case being discussed at the MDT, anoymised details were entered into a specifically designed Microsoft Access database (The proforma is presented in Appendix 1). Patient demographics and baseline characteristics (age, sex, postcode and smoking history), PS, weight loss, laboratory parameters (C-reactive protein, albumin, creatinine and ventilatory function, wherever possible this was based on

full lung function testing, if unavailable spirometry was used), comorbidites (including severity), tumour stage (Detterbeck et al., 2009) and histology and primary treatment proposed by the MDT were all recorded.

Those team members involved in the decision-making processes were also recorded. The doctor delivering the treatment (e.g. thoracic surgeon or clinical oncologist) would have the final say on treatment, thus limiting any potential discrepancy between recommended and delivered treatment. If the primary treatment decision differed from that recommended by the 2005 Scottish Intercollegiate Guidelines Network (SIGN) guidelines on the management of patients with lung cancer (SIGN, 2005) the reason(s) were recorded (for example age, poor performance status, comorbidity, patient choice). If a histological diagnosis was not achieved this was recorded with reason(s) (for example failed procedure, age, poor performance status, comorbidity, non-diagnostic investigations, patient choice). All investigations attempted, clinical/radiological stage and primary treatment plan were itemised.

All four MDTs had input from both a clinical oncologist and thoracic surgeon. Assessment of patient including PS was made by a Respiratory Physician at point of diagnosis prior to MDT treatment plan. All data was recorded in real time at the MDT and at first clinic attendance by one of the clinical staff (either a respiratory physician or clinical nurse specialist), taking on average 2 -3 minutes to enter the data.

#### 3.3 <u>Socioeconomic Status</u>

Information on patients' individual educational or occupational social class was not available, and we therefore used their postcode of residence as a proxy indicator of their socioeconomic status. Using the 2006 Scottish Index of Multiple Deprivation (SIMD) ranking (Office of Chief Statistician, 2006) (see box 1) the postcode enabled us to group patients into one of 5 quintiles.

#### Box 1 - Scottish Index of Multiple Deprivation (SIMD)

The 2006 SIMD is a validated area-based index that uses 37 indicators in seven domains to rank 6505 small geographic areas in Scotland (data zones) from 1 (most deprived) to 6505 (least deprived). These can be subsequently grouped into quintiles. These split up the datazones into 5 groups, each containing 20% of Scotland's datazones. The first quintile contains the 20% most deprived datazones with the fifth representing the 20% least deprived.

#### 3.4 Comorbidities and Severity Scores

Comorbidities present in each patient, with severity of each graded on a four point scale (0, 1, 2, 3) using a novel severity scales (Table 3.1) were recorded, in real time, by a clinician at time of review. The severity index was developed using, where possible, previously validated scoring systems (GOLD Guidelines, 2004; Little et al., 1994; Lyden et al., 2001; Morris, 1993;). This included use of the British Thoracic Society guidelines for Chronic Obstructive Pulmonary Disease, the Canadian Cardiovascular Society Classification for Ischaemic Heart Disease, the New York Heart Association classification for Heart Failure, the National Institutes of Health Stroke Scale for Cerebrovascular disease and the Clinical Dementia Rating. For comorbidities without a validated severity scale we used a scale based on local discussion with eminent regional experts in that particular field.

# Table 3.1Comorbidities and severity scales assessed for eachsubject in the study

Comorbidity	Severity scale	Severity Score			
		0	1	2	3
COPD	BTS/GOLD guidelines	No disease	FEV1 > 60%	FEV1 40-60%	FEV1 < 40%
lschaemic heart disease	Canadian CV Society Classification	No disease	Angina with strenuous/ prolonged exertion	Angina after walking 200 hundred yards flat/flight stairs	Inability to carry out any level of exertion/an gina at rest
Heart failure	NYHA classification	No disease	Slight limitation of physical activity due to dyspnoea	Comfortable at rest, less than ordinary activity causes dyspnoea	Dyspnoea at rest
Cerebrovascular disease	National Institutes of Health Stroke Scale	No detectable weakness/ sensory (incl visual/speech) impairment	Mild weakness/ deficit	Moderate weakness/ deficit	Severe weakness/ deficit
Dementia	Clinical Dementia Rating	No disease	Mild, able to carry out normal activity	Moderate, requires assistance in activities	Severe, unable to manage any activity. Full time care
Diabetes mellitus		No disease	HbA1C < 7	HbA1C 7.1 - 10	HbA1C> 10
Renal function		eGFR> 90ml/min	eGFR 60-89 ml/min	eGFR 30-59 ml/min	eGFR <30ml/min or dialysis
Previous malignancy		No disease or Basal Cell Carcinoma	Previous cancer, no evidence active disease	Active, unlikely to cause death	Active, likely to cause death before lung cancer
Peripheral Vascular Disease		None	Claudication at > 200 yards	Claudication at < 200 yards	Rest pain
Alcohol		<25 Units/week	25-50 units/week	>50 units/week	Establishd alcohol related illness or end organ failure



Figure 3.1: Flowchart of patient numbers in study

#### 3.5 <u>Survival data</u>

Information on date of death was obtained via survival analysis undertaken by the Information Service Division (ISD) of NHS Scotland. Death records were complete until 1 June 2011, which served as the censor date for those alive.

#### 3.6 Ethics

The audit was discussed with the local ethics department and as it was health service clinical practice audit formal ethical approval was deemed not to be necessary.

### CHAPTER 4: VARIATION IN COMORBIDITY AND CLINICAL MANAGEMENT IN NEWLY DIAGNOSED LUNG CANCER PATIENTS IN FOUR SCOTTISH CENTRES

#### 4.1 Introduction

Lung cancer is the commonest cause of cancer death for both men and women in Scotland (Office for National Statistics, 2006; ISD, 2009). Although survival has improved over the past decade, it remains significantly poorer in Scotland than in comparable Western European countries or the USA, with 5-year relative survivals of 8.0%, 10.2% and 16.3%, respectively. Variations in treatment may explain some of the observed differences in lung cancer survival between countries (Berrino et al., 2007; Horner et al., 2009). Surgical resection rates in the United Kingdom ( $\leq 10\%$ ) are consistently reported to be lower than those in Europe and North America, which are in excess of 20% (Fry et al., 1999; Laroche et al., 1998; Cartman et al., 2002; Damhuis et al., 1996). There also appear to be variations in treatment rates between healthcare sites within Scotlandthat might give insights into inequalities in survival (Gregor et al., 2001; Janssen-Heijnen et al., 1998). However, variations in treatment do not wholly explain survival differences between countries and other patient characteristics, including comorbidities, may need to be considered. A number of studies have indicated that comorbid factors influence both choice of therapy, as well as directly affecting survival (Potosky et al., 2004; Earle et al., 2000; Extermann, 2000; Charlson et al., 1987; Tammemagi et al., 2004; Janssen-Heijnen et al., 2004; Colinet et al., 2005; Birim et al., 2005; Imperatori et al., 2006; Asmis et al., 2008; Erridge et al., 2009). However, the investigation of comorbidities has not been the primary aim of any of these studies and the quantification of comorbidities has lacked precision, having been either dichotomised into yes/no categories or expressed as the crude number of co-existent diseases.

To date, no study of lung cancer has attempted to investigate specifically the association between inter-hospital variations in investigation and treatment, and the type and severity of comorbidities. In a pilot study we screened 50 newly diagnosed lung cancer patients presenting sequentially to the Lung Cancer Service at Stobhill General Hospital, Glasgow (Milroy et al., 2005). This pilot study suggested a relationship between increasing comorbidity, worsening performance status and clinicians' inability to offer either potentially curative treatment or optimal palliative treatment. Building on the results of the pilot study we have conducted a study in four Scottish centres to identify prospectively any difference in the investigation and treatment of patients diagnosed with lung cancer. In addition we have quantified the number and severity of co-morbid conditions to investigate how different comorbidity contributes to observed variation in clinical management.

#### 4.2 <u>Methods</u>

Previously described in chapter 3

#### Statistical methods

Statistical analyses were performed using SPSS v17.0 (SPSS Inc., Chicago, USA). The primary outcome variable of interest was the primary treatment option decided by the MDT. Univariable analyses using  $\chi^2$ , t-tests and ANOVA were used to explore the associations between treatment options and potential explanatory variables such as tumour stage, performance status, age, sex, SIMD, co-morbidities (expressed as ordinal scores) and FEV1. Potential explanatory variables with p<0.1 on univariable testing were included in a multiple logistic regression model to identify associations between treatment options and co-morbidities after adjustment for other potentially influential factors.

#### 4.3 <u>Results</u>

#### Between Centre Differences

In total 882 patients were included in the study, comprising 297 from Aberdeen, 136 from Dunfermline, 285 from Glasgow and 164 from Inverclyde. The mean age of participants was 70.4 years and 55.2% were male. There were marked differences between centres in the socioeconomic status profiles of the patients with the majority of patients at Glasgow, but none in Dunfermline, living in the most deprived circumstances, and less marked, but significant differences in age, sex and smoking history profiles (Table 4.1). There were clear differences in the comorbidity severity score profiles between centres for COPD (Fig 4.1a), ischaemic heart disease (Fig 4.1b), congestive cardiac failure ( $\chi^2$ ,p<0.001), dementia ( $\chi^2$ , p=0.008), diabetes mellitus ( $\chi^2$ , p=0.001), renal function  $(\chi^2, p<0.001)$ , weight loss  $(\chi^2, p<0.001)$ , alcohol intake  $(\chi^2, p<0.001)$ , but there were no significant between centre differences in the severity score profiles for cerebrovascular disease, previous malignancies or peripheral vascular disease. Performance status also varied between centres (Table 4.1), as did rate of deterioration in performance status. There was suggestion of between centre differences in the proportion of patients diagnosed with small cell cancer, with no differences in the stage profile (Table 4.1).

Table 4.1	Between centre	differences in	patient profile
-----------	----------------	----------------	-----------------

	All centres	Aberdeen	Dunfermline	Glasgow	Inverclyde	p value*
	n=882	n=297	n=136	n=297	n=164	(between
					_	centres)
Age Mean (95% CI)	70	69	71	71	73	0.001
··· <b>5</b> ······(·····)	(70-71)	(68-70)	(69-73)	(69-72)	(71-74)	
% male	55%	60%	58%	49%	54%	0.049
Ever smoked	94%	94%	97%	93%	94%	0.35
Mean pack year	44	46	38	44	44	0.023
consumption (95%	(42-46)	(42-49)	(35-41)	(41-47)	(39-49)	
CI)	(,	(	(00)	()	(0, 1,)	
Socio-economic	30%	5%	0	64%	42%	<0.001
status % in most						
deprived quintile of						
SIMD						
Socio-economic	7%	19%	0	0	4%	<0.001
status % in most			-	-		
affluent guintile of						
SIMD						
% Performance	79%	81%	82%	76%	78%	0.47
status 0/1 6 months						
prior to						
presentation						
% Performance	47%	58%	53%	43%	28%	<0.001
status 0/1 at						
presentation						
% Performance	41%	28%	36%	44%	65%	<0.001
status fell from 0/1						
to 2/3/4 in 6						
months prior to						
presentation						
FEV1 % predicted	64%	62%	65%	<b>67</b> %	<b>59</b> %	0.018
Mean (95% CI)	(62-65)	(60-65)	(61-69)	(64-69)	(55-63)	
Serum creatinine	93	93	100	87	97	0.008
Mean (95% CI)	(90-96)	(89-97)	(92-107)	(83-91)	(89-105)	
nmol/l	-	-			-	
Serum albumin	37	40	36	33	37	<0.001
Mean (95% CI) g/l	(36-37)	(39-41)	(35-37)	(32-34)	(36-37)	
CRP mg/l	48	46	62	45	56	0.21
Mean (95% CI)	(44-53)	(39-53)	(39-74)	(38-52)	(45-67)	
% Non Small Cell	62%	58%	60%	<b>59</b> %	76%	0.001
% Small Cell	13%	15%	7%	16%	12%	0.077
% Small Cell Limited	29%	22%	20%	34%	37%	0.47
Disease	(33/115)	(10/45)	(2/10)	(14/41)	(7/19)	
% Non Small Cell	21%	27%	22% (18/81)	18%	14%	0.028
Stage I or II	(109/531)	(47/173)		(27/153)	(17/124)	
% No histology Stage	18%	12%	20%	21%	0%	0.57
lorll	(31/173)	(6/50)	(8/41)	(17/81)	(0/1)	

\*ANOVA,  $\chi^2$  tests of statistical significance



Figure 4.1 Bar chart illustrating between centre differences in severity scores for (a) chronic obstructive pulmonary disease (COPD) and (b) ischaemic heart disease (IHD). Between centre p values 0.017 for COPD and <0.001 for IHD,

There were marked differences between centres in the investigations performed to stage and characterise lung cancer. In Dunfermline, for local reasons, measurement of CRP was about a third that of other sites  $(\chi^2, p<0.001)$  and histological diagnosis was lower  $(\chi^2, p<0.001)$  (Table 4.1). The majority of the patients had NSCLC (62.0%), 13.4% had SCLC and the remainder of patients had no pathological confirmation. There was variation in non-small cell stage at presentation between centres with the incidence of stage I or II disease ranging from 13.7% of patients at Inverclyde to 27.2% of patients in Aberdeen ( $\chi^2, p=0.028$ ) (Table 4.1). This is clinically important, as stage at presentation is one of the most important factors in determining treatment and survival (Mountain, 1986). The proportion of patients presenting with limited stage SCLC varied between centres from 36.8% at Inverclyde to 20.0% in Dunfermline, but probably because of low numbers this difference was not statistically significant (Table 4.1).

Between centre differences were observed in the rates of radical radiotherapy, the proportion of patients with locally advanced/metastatic NSCLC having chemotherapy and the proportion of patients for who best supportive care with no surgical/oncological intervention was recommended. The differences between sites in proportions of patients referred for surgical resection were small and not statistically significant (Table 4.2).

	All	Aberdeen	Dunfer	Glasgow	Inverclyd	p value*
	centres	n=297	mline	n=297	e	(betwee
	n=882		n=136		n=164	n
						centres)
Measurement of CRP	86.3%	96.6%	37.5%	94.7%	93.3%	<0.001
Obtained histological	75.5%	73.4%	<b>66.9</b> %	74.9%	87.2%	<0.001
diagnosis						
Surgical resection	6.3%	7.4%	6.6%	5.6%	5.5%	0.79
Radical Radiotherapy	4.0%	4.0%	8.8%	3.1%	0.6%	0.004
Chemotherapy NSCLC	4.0%	9.8%	0%	2.8%	0%	<0.001
stage IIIb/IV						
No	20.6%	10.8%	36.8%	22.5%	22.0%	<0.001
surgery/chemo/radiot						
herapy						

## Table 4.2Betweencentredifferencesininvestigationandtreatment

 $^{*}\chi^{2}$  or Fishers exact tests of statistical significance

#### Associations with Performance Status

Of the 882 patients 46.8% (413) had a performance status of 0 or 1 at time of diagnosis. Univariable analysis demonstrated that a performance status of 0/1 was associated with centre (Aberdeen 58.2%, Fife 52.9%, Stobhill 42.8%, Inverclyde 28.0%, Table 4.3 model 1,  $\chi^2$ ,p<0.001), younger age (67.6 years 95% CI 66.5-68.7 vs. 72.4 (71.2-73.5), t test, p<0.001), those in more affluent areas as defined by SIMD, (highest quintile of affluence 62.9% PS 0/1 vs. lowest quintile 40.4%,  $\chi^2$ , p<0.001) and those with a confirmed pathological (i.e. either cytological or histological) diagnosis (SCLC 48.7%, NSCLC 53.3%, no histology 32.4%,  $\chi^2$ , p<0.001). Univariable analysis demonstrated that performance status was adversely associated with increasing severity of the comorbidities: COPD, IHD, CCF, cerebrovascular disease, dementia, renal impairment, weight loss, peripheral vascular disease and alcohol history. There were no associations with sex, or severity of diabetes mellitus. Multivariable modeling demonstrated that a performance status of 0/1 was less likely with increasing age, some centres (Glasgow and Inverclyde), and increasing

severity of the comorbidities COPD, CCF, cerebrovascular disease and dementia (Table 4.3). Reduced performance status at presentation (PS 2/3/4) was associated with NSCLC stage IV, extensive SCLC and failure to achieving positive tumour histology. The predictive power of the final logistic regression model to identify correctly individuals with a performance status of 0/1 was assessed using receiver operating characteristic (ROC) curve analysis. The area under the ROC curve for the final logistic model was 0.79 indicating fair to good discrimination (although as this was tested on the same cohort of patients from which it was derived this may appear artificially good).

Table 4.3 Results of univariable analysis relating performance status 0/1 to centre (model 1). Model 2 is the results of multiple logistic regression modeling relating performance status 0/1 to centre, age, comorbidity scores (for COPD, CCF, cerebrovascular disease, dementia, weight loss), and tumour stage, and histology,

	Model 1		Model 2	
	Odds ratio (95% CI)	p value	Odds ratio (95% CI)	p value
Aberdeen(n=173)Dunfermline(n=72)Glasgow(n=122)Inverclyde(n=46)	1 0.87 (0.54-1.21) 0.54 (0.39-0.75) 0.28 (0.19-0.42)	0.30 <0.001 <0.001	1 1.50 (0.90-2.49) 0.47 (0.32-0.71) 0.29 (0.18-0.46)	0.12 <0.001 <0.001
Age (years) <60 (n=80) 60-70 (n=142) 70-80 (n=149) >80 (n=42)			1 0.85 (0.51-1.41) 0.58 (0.35-0.95) 0.25 (0.14-0.45)	<0.001*
COPD (FEV <sub>1</sub> % predicted) >80% (n=252) 60-80% (n=80) 40-60% (n=64) <40% (n=42)			1 0.93 (0.59-1.48) 0.54 (0.35-0.83) 0.22 (0.10-0.48)	<0.001*
CCF (severity score) 0 (n=369) 1 (n=35) 2 (n=6) 3 (n=1)			1 0.67 (0.38-1.17) 0.21 (0.08-0.59) 0.33 (0.03-3.63)	0.002*
Cerebrovascular disease (severity score) 0 (n=800) 1 (n=52) 2 (n=18) 3 (n=8)			1 0.58 (0.29-1.16) 0.12 (0.02-0.62) 0.00 (0.00-)+	<0.001*
Dementia (severity score) 0 (n=827) 1 (n=24) 2 (n=14) 3 (n=1)			1 0.59 (0.19-1.86) 0.34 (0.08-1.50) 0.00 (0.00-)+	0.002*
Weight loss None (n=416) <5% (n=131) 5-10% (n=74) >10% (n=258)			1 0.47 (0.28-0.78) 0.49 (0.27-0.91) 0.20 (0.14-0.30)	<0.001*
NSCLC stage I/II (n=136) NSCLC stage III (n=233) NSCLC stage IV (n=311) SCLC limited (n=33) SCLC extensive (n=82) No histology (n=26)			1 0.67 (0.40-1.13) 0.37 (0.22-0.62) 0.52 (0.22-1.28) 0.47 (0.25-0.91) 0.42 (0.20-0.89)	0.13 <0.001 0.15 0.026 0.023

\*: p value for trend across categories

+: upper 95 CI not computable

#### Predictors of decline in performance status.

Of the 694 patients with a retrospectively estimated performance status of 0 or 1 six months prior to diagnosis, 40.9% had declined by the time of presentation. Univariable analysis demonstrated that a decline in performance status from 0/1 to 2/3/4 in the 6 months prior to presentation was associated with centre (Aberdeen 28.2%, Dunfermline 35.5%, Glasgow 43.8%, Inverclyde 65.1%,  $\chi^2$ ,p<0.001, Table 4.4 model 1), increasing age, increasing deprivation (highest quintile of affluence 18.8%, lowest quintile of affluence 46.8%), COPD, IHD, CCF, cerebrovascular disease, dementia, tumour histology and stage. Multivariable modeling (Table 4.4 model 2) demonstrated that a decline in performance status from 0/1 to 2/3/4 in the 6 months prior to presentation was associated with centre (more likely in Inverclyde), increasing age, decreasing affluence, COPD, CCF, cerebrovascular disease, tumour stage and tumour histology.

Table 4.4: Results of univariable analysis relating decline in performance status from 0/1 to 2/3/4 in the 6 months prior to presentation to centre (model 1). Model 2 is the results of multiple logistic regression modeling relating decline in performance status to centre, age, SIMD, and comorbidity scores (for COPD, CCF, cerebrovascular disease), and tumour stage and histology.

	Model 1		Model 2	
	Odds ratio (95%	p value	Odds ratio (95%	p value
	CI)		CI)	
Aberdeen (n=68)	1		1	
Dunfermline (n=39)	1.40 (0.86-2.26)	0.17	1.07(0.62-1.83)	0.82
Glasgow (n=95)	1.98 (1.34-2.92)	0.001	1.48 (0.87-2.52)	0.15
Inverclyde (n=82)	4.74 (2.99-7.52)	<0.001	3.87 (2.27-6.60)	<0.001
Age (years)				
<60 (n=36)			1	<0.001*
60-70 (n=77)			1.15 (0.68-1.97)	
70-80 (n=102)			1.47 (0.87-2.49)	
>80 (n=69)			3.33 (1.77-6.28)	
Quintile affluence (SIMD)				
Q1: Least affluent(n=94)			1	0.047*
Q2(n=49)			1.39 (0.76-2.55)	
Q3 (n=90)			1.06 (0.65-1.73)	
Q4 (n=42)			0.77 (0.41-1.47)	
Q5: most affluent (n=9)			0.34 (0.13-0.88)	
COPD (FEV <sub>1</sub> % predicted)				
>80% (n=154)			1	<0.001*
60-80% (n=44)			1.03 (0.62-1.70)	
40-60% (n=63)			1.71 (1.07-2.73)	
<40% (n=22)			4.12 (1.80-9.45)	
CCF (severity score)				
0 (n=243)			1	0.008*
1 (n=31)			1.97 (1.08-3.59)	
2 (n=8)			3.46 (1.01-11.9)	
3 (n=1)			2.76 (0.16-46.8)	
Cerebrovascular disease (severity				
score)				
0 (n=257)			1	0.007*
1 (n=20)			1.84 (0.85-3.99)	
2 (n=6)			7.70 (1.27-46.9)	
3 (n=0)			++	
NSCLC stage I/II (n=21)				
NSCLC stage III (n=72)			2.63 (1.43-4.84)	0.002
NSCLC stage IV (n=119)			5.60 (3.09-10.1)	<0.001
SCLC limited (n=11)			3.22 (1.22-8.47)	0.018
SCLC extensive (n=28)			3.98 (1.90-8.32)	<0.001
No histology (n=23)			4.37 (1.94-9.87)	<0.001
			1	

\*: p value for trend across categories

++: OR not computable (too few subjects in category)

#### Associations with surgical resection for NSCLC

The surgical resection rates for confirmed NSCLC were 12.7% (Aberdeen), 11.1% (Dunfermline), 10.5% (Glasgow) and 7.3% (Inverclyde) but the differences were not statistically significant. Given the differences in the patient profiles between centres, multivariable modeling was performed and the results are outlined in Table 4.5 as model 1. After adjustment, there were the expected associations between surgical resection, performance status and tumour stage. Female sex and increasing age were associated with a reduced likelihood of surgical resection. The only comorbidity significantly associated with the reduced chance of surgical resection was COPD. The multivariable analysis indicated that after adjustment for performance status, tumour stage, age, sex and COPD, patients in Inverclyde were more than five times likely to be operated on for NSCLC than patients from Aberdeen. It was also noted however that the number of patients with Stage 1 and II NSCLC who declined surgical treatment differed between centres: 25% (Aberdeen 3/12), 50% (Dunfermline 5/10), 0% Glasgow (0/7) and Inverclyde (0/8) ( $\chi^2$ ,p=0.009) and that women were more likely to decline surgical treatment than men (21.7% vs. 11.5%) although this difference was not statistically significant. Patient refusal was not associated with socio-economic status. If the patients with Stage I or Stage II NSCLC who declined surgery had actually been operated on (Table 4.5 model 2) the sex and age associations with surgical resection were not significant and patients in Dunfermline were more likely to undergo surgical resection.

Table 4.5 Results of multiple logistic regression relating decision to treat non-small cell lung cancer by surgical resection to centre, sex, age, comorbidity, performance status, and tumour stage. Model 1: subjects who declined surgical intervention coded as no surgery, Model 2: subjects who declined surgical intervention coded as surgical intervention.

$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		Model 1		Model 2	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		Odds ratio (95% CI)	p value	Odds ratio (95% CI)	p value
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Aberdeen (n=22)	1		1	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Dunfermline (n=9)	1.19 (0.28-5.00)	0.81	4.39(1.14-16.9)	0.032
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Glasgow (n=16)	2.49 (0.83-7.46)	0.10	1.57(0.55-4.41)	0.40
Female sex $(n=22)$ $0.39 (0.16 \cdot 0.96)$ $0.041$ $0.53 (0.22 \cdot 1.31)$ $0.21$ Age (years) < 60	Inverclyde (n=9)	5.84 (1.50-22.7)	0.011	2.77(0.77-10.0)	0.12
Female sex $(n=22)$ $0.39$ $(0.16-0.96)$ $0.041$ $0.53(0.22-1.31)$ $0.21$ Age (years) < 60					
$\begin{array}{c cccc} Age (years) \\ <60 & (n=9) \\ 60-70 & (n=25) \\ 70-80 & (n=21) \\ >80 & (n=1) \end{array} & \begin{array}{c} 1 \\ 0.56 & (0.13 \cdot 2.47) \\ 0.37 & (0.08 \cdot 1.66) \\ 0.02 & (0.001 \cdot 0.29) \end{array} & \begin{array}{c} 0.007^* & \begin{array}{c} 1 \\ 1.03 & (0.22 \cdot 4.90) \\ 0.60 & (0.13 \cdot 2.82) \\ 0.10 & (0.01 \cdot 0.79) \end{array} & \begin{array}{c} 0.051^* \\ 0.008^* & \begin{array}{c} 1 \\ 0.37 & (0.13 \cdot 1.08) \\ 0.29 & (0.08 \cdot 0.99) \\ 0.14 & (0.02 \cdot 1.21) \end{array} & \begin{array}{c} 0.012^* \\ 0.008^* & \begin{array}{c} 1 \\ 0.37 & (0.13 \cdot 1.08) \\ 0.29 & (0.08 \cdot 0.99) \\ 0.14 & (0.02 \cdot 1.21) \end{array} & \begin{array}{c} 0.012^* \\ 0.018^* \\ 0.018^* \end{array} & \begin{array}{c} 0.008^* \\ 0.002^* & \begin{array}{c} 1 \\ 0.31 & (0.09 \cdot 1.09) \\ 0.18 & (0.04 \cdot 0.76) \\ 0.07 & (0.06 \cdot 0.71) \\ 0.00 & (0.00 \cdot 0) + \end{array} & \begin{array}{c} 0.018^* \\ 0.018^* \end{array} & \begin{array}{c} 0.018^* \\ 0.008^* \\ 0.000 & 0.00 & 0 \end{array} & \begin{array}{c} 0.018^* \\ 0.000 & 0.00 & 0 \\ 0.00 & 0.00 & 0 \\ 0.00 & 0.00 & 0 \\ 0.00 & 0.00 & 0 \\ 0.00 & 0.00 & 0 \\ 0.00 & 0 & 0$	Female sex (n=22)	0.39 (0.16-0.96)	0.041	0.53(0.22-1.31)	0.21
Age (years)1 $0.007^*$ 1 $0.007^*$ 1 $0.051^*$ $60^{-70}$ (n=25) $0.56$ ( $0.13 \cdot 2.47$ ) $0.007^*$ $1$ $0.03(0.22 \cdot 4.90)$ $0.051^*$ $70 \cdot 80$ (n=21) $0.37$ ( $0.08 \cdot 1.66$ ) $0.60(0.13 \cdot 2.82)$ $0.10(0.01 \cdot 0.79)$ $0.10(0.01 \cdot 0.79)$ $>80^{\circ}$ (n=1) $0.02(0.001 \cdot 0.29)$ $0.008^*$ $1$ $0.008^*$ $1$ $60 \cdot 80\%$ (n=17) $0.36$ ( $0.12 \cdot 1.13$ ) $0.008^*$ $1$ $0.37(0.13 \cdot 1.08)$ $0.012^*$ $40\%$ (n=1) $0.26$ ( $0.07 \cdot 1.03$ ) $0.26$ ( $0.07 \cdot 1.03$ ) $0.14(0.02 \cdot 1.21)$ $0.018^*$ $40\%$ (n=1) $0.24$ ( $0.07 \cdot 0.88$ ) $0.002^*$ $1$ $0.018^*$ $0.002^*$ $1$ $0.002^*$ $0.31(0.09 \cdot 1.09)$ $0.018^*$ $1$ (n=27) $0.24$ ( $0.07 \cdot 0.88$ ) $0.31(0.09 \cdot 1.09)$ $0.18(0.04 \cdot 0.76)$ $3$ (n=2) $0.04$ ( $0.01 \cdot 0.53$ ) $0.00(0.00^-$ )+ $0.00(0.00^-$ )+					
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Age (years)				
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	<60 (n=9)	1	0.007*	1	0.051*
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	60-70 (n=25)	0.56 (0.13-2.47)		1.03(0.22-4.90)	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	70-80 (n=21)	0.37 (0.08-1.66)		0.60(0.13-2.82)	
$\begin{array}{c cccc} COPD(FEV_{1\%} \ predicted) \\ >80\% \ (n=32) \\ 60\cdot80\% \ (n=17) \\ 40\cdot60\% \ (n=6) \\ <40\% \ (n=1) \end{array} \begin{array}{c ccccc} 1 & 0.36 \ (0.12\cdot1.13) \\ 0.26 \ (0.07\cdot1.03) \\ 0.10 \ (0.01\cdot1.18) \end{array} \begin{array}{c ccccccccccccccccccccccccccccccccccc$	>80 (n=1)	0.02(0.001-0.29)		0.10(0.01-0.79)	
$\begin{array}{c cccc} \text{COPD}(\text{FEV}_{1\%} \text{ predicted}) \\ >80\% & (n=32) & 1 \\ 60-80\% & (n=17) & 0.36 & (0.12-1.13) \\ 40-60\% & (n=6) & 0.26 & (0.07-1.03) \\ <40\% & (n=1) & 0.10 & (0.01-1.18) \end{array} \\ \begin{array}{c ccccccccccccccccccccccccccccccccccc$					
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	COPD(FEV <sub>1</sub> % predicted)				
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	>80% (n=32)	1	0.008*	1	0.012*
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	60-80% (n=17)	0.36 (0.12-1.13)		0.37(0.13-1.08)	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	40-60% (n=6)	0.26 (0.07-1.03)		0.29(0.08-0.99)	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	<40% (n=1)	0.10 (0.01-1.18)		0.14(0.02-1.21)	
Performance status 0 $(n=21)$ 1 $0.002^*$ 1 $0.018^*$ 1 $(n=27)$ $0.24 (0.07 \cdot 0.88)$ $2 (n=6)$ $0.08 (0.02 \cdot 0.38)$ $0.18 (0.04 \cdot 0.76)$ $0.07 (0.06 \cdot 0.71)$ $4 (n=0)$ $0.00 (0.00^-)^+$ $0.002^*$ $0.0018^*$					
Performance status 010.002*10.018*1 $(n=27)$ $0.24 (0.07-0.88)$ $0.08 (0.02-0.38)$ $0.31(0.09-1.09)$ $0.18(0.04-0.76)$ $0.07(0.06-0.71)$ $0.018^*$ 3 $(n=2)$ $0.04 (0.01-0.53)$ $0.00 (0.00-)+$ $0.00(0.00-)+$					
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $					
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Denfermente				
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Performance status	4	0.002*	4	0.049*
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0 (n=21)		0.002		0.018
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1 (n=27)	0.24 (0.07 - 0.88)		0.31(0.09-1.09)	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	2 (n=0)	0.08 (0.02 - 0.38)		0.18(0.04-0.76)	
4 (n=0)   0.00 (0.00- )+   0.00 (0.00- )+	3 (h=Z)	0.04 (0.01-0.53)		0.07(0.06-0.71)	
	4 (n=0)	0.00 (0.00- )+		0.00(0.00- )+	
Tumour stago	Tumour stage				
1 - (n-38) = 1 -	l (n=38)	1	<0.001*	1	<0.001*
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	(n-30)	1 0 48 (0 18-1 30)	~0.001	0.65(0.25-1.66)	<b>\U.UU</b>
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		0.70(0.10(1.30))		$0.03(0.23^{-1.00})$	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(n-4)			0.03(0.07 - 0.11)	
V  = 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,	$\frac{110}{110} \qquad (n=0)$				

\*: p value for trend across categories

+: upper 95 CI not computable

#### Associations with radical radiotherapy for NSCLC

There were clear between centre differences in the rates of radical radiotherapy (definition as per SIGN guidelines<sup>23</sup>) for NSCLC (Tables 4.2, 4.6) that persisted after adjustment for factors found to be associated with radical radiotherapy for NSCLC on univariable analysis. Multivariable modeling also indicated that the decision to treat NSCLC with radical radiotherapy was more likely if the patient was a woman and in patients with COPD. There was a non-significant associated comorbidity.

Table 4.6 Results of univariable analysis relating decision to treat non-small cell lung cancer by high dose radical radiotherapy to centre (model 1). Model 2 outlines the results of multiple logistic regression modeling relating decision to treat by high dose radical radiotherapy to centre, sex, comorbidity (COPD, alcohol) and tumour stage.

	Model 1		Model 2	
	Odds ratio (95% CI)	p value	Odds ratio (95%	p value
			CI)	
Aberdeen (n=31)	6.75 (0.84-54.0)	0.072	8.52 (0.98-74.3)	0.052
Dunfermline (n=19)	13.5 (1.65-110)	0.015	14.6 (1.54-137)	0.019
Glasgow (n=28)	8.60 (1.09-68.1)	0.042	9.84 (1.14-84.9)	0.038
Inverclyde (n=4)	1		1	
Female sex (n=41)			2.56 (1.05-6.26)	0.040
COPD (FEV <sub>1</sub> % predicted)				
>80% (n=41)			1	0.023*
60-80% (n=19)			3.79 (1.22-11.8)	
40-60% (n=19)			5.41 (1.62-18.0)	
<40% (n=3)			3.68 (0.56-24.2)	
Alconol (severity score)			1	0.0/0*
0 (II=09)				0.063
1 (11=7)			0.40(0.09-2.70)	
2 (11=3)			12.2 (0.04-170)	
J S (∏=∠)			12.2 (0.09-100)	
NSCI C stage $1/11$ (n=24)			1	
NSCLC stage I/II (II=24) $NSCLC$ stage III (II=25)			1 0 24 (0 00-0 61)	0.003
NSCLC stage III (II=55) $NSCLC stage IV (p=0)$			0.24 (0.07-0.01)	0.005
NOCLU SLAYE IN (II=U)			тт	

\*: p value for trend across categories

++: OR not computable (too few subjects in category)

#### Associations with optimal treatment of SCLC

In total 13.4% of the patients were diagnosed with SCLC, with 28.7% staged as limited disease. Optimal treatment for limited stage SCLC was defined as chemotherapy with a platinum doublet along with consolidation radiotherapy to mediastinum (conventionally defined as at least 30Gy (SIGN, 2005) and brain, either sequentially or concurrently (SIGN, 2005). There was evidence of between centre differences in the incidence of SCLC, with Aberdeen highest and Dunfermline lowest (15.2 versus 7.4% of patients respectively) but probably because of small numbers these differences did not reach statistical significance ( $\chi^2$ ,p=0.077) (Table 4.2). However, multivariable modeling suggested that after adjustment there were between centre differences in the decision to treat SCLC with optimal therapy, furthermore there were associations between optimal treatment and performance status, alcohol associated comorbidity and IHD comorbidity. (Table 4.7) Table 4.7 Results of univariable analysis decision to treat small cell lung cancer with combination chemotherapy or concurrent chemoradiotherapy to centre (model 1). Model 2 outlines the results of multiple logistic regression modeling relating decision to treat small cell lung cancer with combination chemotherapy or concurrent chemoradiotherapy to centre to centre, comorbidity (alcohol, COPD), and performance status.

model	
Odds ratio (95% CI) p value Odds ratio (95% CI) p v	alue
Aberdeen (n=5) 1.07 (0.28-4.00) 0.92 0.24 (0.03-1.67) 0.1	5
Dunfermline (n=1) 0.27 (0.05-1.40) 0.12 0.05 (0.04-0.56) 0.0	)16
Glasgow (n=12) 0.83 (0.22-3.07) 0.78 0.30 (0.05-1.95) 0.2	.1
Inverclyde (n=5) 1 1	
Ischaemic heart disease (severity score) 1 0.0   0 (n=14) 1 0.0   1 (n=2) 0.48 (0.10-2.24) 0.06 (0.01-0.54)	)37*
3 (n=2) 1.14 (0.02-62.1)	
Alcohol (severity score) 0 (n=12) 1 (n=3) 2 (n=2) 3 (n=1) 0.02 (0.01-0.39) 0.12 (0.01-2.09) 0.12 (0.01-2.09)	)10*
COPD(FEV1% predicted)   1   0.0     >80% (n=14)   0.09 (0.01-1.15)   0.39 (0.04-3.54)     40-60% (n=3)   0.51 (0.01-2.18)   0.51 (0.01-2.18)	90*
Performance status	004*
$\begin{bmatrix} U & (n=3) \\ 1 & (n=11) \end{bmatrix} = \begin{bmatrix} 1 \\ 1 & F(0,11,2,2) \end{bmatrix}$	001*
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	
$\begin{array}{c} 3 \\ 4 \\ (n=0) \\ \end{array}$	

\*: p value for trend across categories

+: upper 95 CI not computable

#### Associations with best supportive/palliative care

For 182 patients (20.6%) the treatment option was best supportive care and/or referral to specialist palliative care i.e. no primary surgery, radiotherapy, or chemotherapy. There were significant ( $\chi^2$ ,p<0.001) between centre differences in the frequency of supportive/palliative care only with 10.8% (32/265) of patients in Aberdeen, 36.8% (50/136) in Dunfermline, 22.5% (64/285) in Glasgow and 22.0% (36/164) patients in Inverclyde receiving supportive/palliative supportive care (Table 4.8 model 1). Univariable analyses revealed that the likelihood of supportive/palliative care was increased by increasing age (mean 75.1 years 95%CI 73.8-76.4 vs. 69.1 (68.4-69.9), t test p<0.001), a failure to make a histological diagnosis (SCLC 7.8%, NSCLC 13.7%, no histological diagnosis 45.9%), decreasing performance status (active supportive care for PS 0/1 7.0% vs. 32.9% for PS 2/3/4,  $\chi^2$ , p<0.001), increasing serum CRP and creatinine, decreasing albumin and increasing severity scores for congestive cardiac failure, cerebrovascular disease, dementia, renal impairment, and peripheral vascular disease. The decision to manage a patient with best supportive/palliative care was not associated with sex, socio-economic status, COPD, IHD, diabetes nor alcohol history. Multivariable modeling demonstrated that even after adjustment for multiple factors, between centre differences persisted (but not between Inverclyde and Aberdeen) and in each centre the decision to treat with best supportive/palliative care was associated with age, performance status, dementia severity, and extensive small cell lung cancer (Table 4.8, model 2).

Table 4.8 Results of univariable analysis relating decision to treat by best supportive/palliative care to centre (model 1). Model 2 outlines the results of multiple logistic regression modeling relating decision to treat by best supportive/palliative care to centre, age, dementia comorbidity score, performance status, tumour stage, and tumour histology.

	Model 1	Model 2	
	Odds ratio (95% Cl)	p value Odds ratio (95% CI)	p value
Aberdeen (n=3	2) 1	1	
Dunfermline (n=5	) 4.82 (2.90-7.99)	<0.001 3.66(1.97-6.78)	<0.001
Glasgow (n=6	4) 2.40 (1.51-3.80)	<0.001 1.79(1.04-3.09)	0.036
Inverclyde (n=3	5) 2.33 (1.38-3.92)	0.001 1.20(0.66-2.18)	0.55
Age (years)			0.004
<60 (n=19)			<0.001*
60-70 (n=3)		1.24(0.54-2.83)	
/0-80 (n=10	0)	2.14(0.99-4.61)	
>80 (n=72)		3.91(1.73-8.82)	
Dementia (severity sco	.)		
0 (n=192)	,	1	0.003*
1 (n=16)		4 04(1 50-10 9)	0.005
(n = 10)		3,92(1,05-14,6)	
3 (n=10)		3.38(0.77-14.8)	
Performance status			
0 (n=12)		1	<0.001*
1 (n=39)		2.07(0.67-6.38)	
2 (n=62)		4.40(1.44-13.4)	
3 (n=89)		19.2 (6.2-58.8)	
4 (n=26)		60.4 (14.1-259)	
NSCLC stage I/II (n=2	)	1	
NSCLC stage III (n=50		0.65 (0.33-1.28)	0.21
NSCLC stage IV(n=122)		1.23 (0.65-2.30)	0.53
SCLC limited (n=4)		0.44 (0.11-1.79)	0.25
SCLC extensive (n=8		0.32 (0.11-0.91)	0.032
No histology (n=9		0.77 (0.29-1.99)	0.58

\*: p value for trend across categories
#### 4.4 **Discussion**

In this study we have prospectively collected data from 882 patients newly diagnosed with lung cancer in four Scottish centres. Within each centre the patients were consecutive and thus representative of the local population and medical practice. Our hypothesis was that variations in population demographics such as socio-economic status, performance status, stage at presentation, comorbidity and age determined the ability to deliver optimal standard therapy. The aim was to assess variations in practice between the four Scottish centres and possible causes for this such as demographics, age and comorbidity. A number of differences were identified between centres. These included investigation, treatment (surgical rates), patient age, tumour histology, stage, smoking history, socio-economic profile, ventilatory function and performance status. These differences may, in some part, explain the variations in practice seen in Scotland and we have attempted to characterise the very complex relationship between them.

Much of the variation in the initial treatments delivered may be accounted for by differences in comorbidity and patient choice. This is especially the case for those who decline surgical intervention. For supportive/ palliative care (the decision not to offer active anti-cancer therapy) between centre differences persisted after adjustment for tumour factors, demographics and comorbidities. The treatment rates shown in this chapter are generally unexpectedly low with surgical resection rates under 8% and palliative chemotherapy less than 10%. This was noted at the time of this aspect of the work and the data reanalyzed. These rates are inaccurately low due to coding errors, which are rectified in later chapters of the thesis.

Declining PS has been shown to correlate well with poorer survival (Radzikowska et al., 2002; Ludbrook et al., 2003). It is difficult to assess if comorbidity represents a separate predictive entity for treatment choice

72

or is simply a surrogate for performance status. The multivariable modeling indicated that whilst performance status is strongly associated with treatment delivered, the presence of co-existing comorbidities was additionally associated with treatment differences. Thus co-morbidities not only contribute to performance status but also have additional associations with treatment selection. In our study, predictors of declining performance status included increasing severity of a number of comorbidities, age, lower socio-economic status and specific geography (primarily Glasgow and Inverclyde).

The rate of decline in PS has never previously been explored. In multivariable analysis this rate has been shown to be related to increasing age, lower socio-economic group, several comorbid factors and tumour stage. However this assessment was retrospective and it has not been validated. The decline in PS, independently associated with socioeconomic status within individual cities, probably reflects social trends in psychosocial factors influencing health e.g. diet, housing and social support. This decline in performance status also highlights the need for patients, particularly in lower socio-economic classes, to be seen and assessed early on in their disease before they deteriorate to a level where radical treatment is inappropriate. The results of this study suggest that the striking differences in deprivation seen between centres and generally within Scotland, in comparison to the remainder of the United Kingdom (Office for National Statistics, 2006; ISD, 2009; Gregor et al., 2001; Janssen-Heijnen et al., 1998) may be a contributory factor to the observation that Scottish patients with lung cancer have a poorer outcome in a global setting.

Surgical resection rates are closely associated with cure for NSCLC (SIGN, 2005), and the identification of factors influencing this, especially comorbidities, were a primary study aim. The mean surgical resection rate in this study was 6.3% with a range from 5.5% to 7.4%, these figures are similar to those reported by a Cancer Registry in the South East of England

that reported a median resection rate of 6% with a range of 4 - 10% (Jack et al., 2003). In the present study socio-economic status was not associated with the likelihood of patients declining surgery. This contrasts with published studies that suggest lower socio-economic status patients are more likely to decline surgery (Mitchell et al., 2003). COPD was the only comorbid factor to persist once multi-variable analysis, including performance status, had been performed. The available literature supports the assertion that comorbidity is detrimental to outcome. However these studies have not assessed comorbidity in any significant detail (Birim et al., 2005; Imperatori et al., 2006)

The decision to offer best supportive/palliative care rather than active anticancer therapy clearly differed between centres, even after multivariable analysis. This is likely to reflect physician's choice using different criteria to determine treatment choice for a patient and it is notable that in this study the centre with the lowest rate of initial decision to offer best supportive/palliative care had the highest rate of chemotherapy for locally advanced/metastatic NSCLC. The present study is consistent with previous studies suggesting that comorbidity appears to influence treatment choice and potentially affect outcome, but adds to the literature by identifying relevant individual comorbidities and demonstrating dose response associations (Potosky et al., 2004; Earle et al., 2000; Extermann, 2000; Charlson et al., 1987; Tammemagi et al., 2004; Janssen-Heijnen et al., 2004; Colinet et al., 2005).

In an attempt to try to assess the reasons for differences in treatment and outcome within lung cancer in Scotland, this early data has shown some interesting differences between centres in relation to patient characteristics and treatment. We believe that one of the strengths of this study is that the study population is representative of the local population of people presenting with lung cancer because the centralised system of free universal healthcare in Scotland ensures that most people with suspected lung cancer are referred directly to Respiratory Physicians, in addition comparisons of data submitted centrally from lung cancer MDTs with Cancer Registry data confirms that Respiratory Physicians investigate in excess of 90% of people with lung cancer in Scotland (MCN, Personal Communication). We suspect that the 10% of patients not investigated by Respiratory Physicians are relatively elderly, frail with advanced cancer diagnosed and managed at home by General Practitioners and those diagnosed at post-mortem. The primary outcome of interest was the initial treatment option decided by the MDT and given that, the clinician responsible for instigating the treatment agrees upon this decision, then it is almost certain that the treatment option was initiated. A limitation of this study is that because of relative small patient numbers it was not possible to analyse the factors influencing some of the more complex combined modality treatment decisions e.g. neo-adjuvant chemotherapy, sequential chemotherapy and radical radiotherapy, furthermore for pragmatic reasons it was not possible to ascertain whether the second part of these treatment options were actually commenced and if not why not. Such considerations may have contributed to the low rates of radical radiotherapy reported. A further limitation of this study is that we do not yet have survival data on the subjects and therefore cannot relate our observations on comorbidity to survival. We are currently investigating the possibility of using centrally collected data to address this.

The important message of this study is that is scientifically unsound to compare crude data from centres or countries and conclude that variations entirely reflect differences in practice. Whilst it is common practice to adjust such comparative data for sex, age, tumour and in some instances indirect measures of comorbidity (days in hospital in previous five years) our study highlights the need to adjust for other factors such as patient choice, comorbidity and performance status. The present study suggests that adjustment of comparative data for variation in performance status is insufficient and further adjustment for specific and quantified comorbidities should be carried out, as comorbidity appears to have an independent impact upon between centre differences. Even so, whilst the present study highlights the importance of variation in comorbidity, performance status, and tumour stage there is still evidence of between centre differences in practice suggesting variation in clinical practice.

Previous studies have shown that Scotland (in particular the West of Scotland) suffers from significantly higher rates of many comorbid diseases such as Ischaemic Heart Disease (Mitchell et al., 2005), COPD (Thorax, 2004) and Alcoholism (Emslie et al., 2009). The present study suggests that the combination of these comorbidities and social deprivation is especially disadvantageous, not only influencing the rate at which performance status deteriorates in the six months prior to presentation, but additionally affecting the treatment options available to the assessing physician. The particularly adverse combination of chronic ill-health and social deprivation might explain the significantly poorer performance status, lower active treatment rates and poorer survival seen for certain centres in Scotland.

This study has identified many significant between centre differences within Scotland. We believe this to be the first study to identify non-tumour factors independent of performance status that together limit the ability to deliver radical, possibly curative, therapy to our lung cancer population. It is only by identifying such factors that we can hope to address the significant health inequalities seen across the four centres and in the wider population and begin to improve upon the relatively poor outlook for the majority of Scottish lung cancer patients.

76

# CHAPTER 5: COMORBIDITIES IN LUNG CANCER: PREVALENCE, SEVERITY AND LINKS WITH SOCIO-ECONOMIC STATUS AND TREATMENT.

## 5.1 Introduction

Lung cancer remains the commonest cause of cancer related death in Scotland (Gregor et al., 2001). Survival lags significantly behind much of Western Europe and the United States (Gregor et al., 2001; Janssen-Heijnen et al., 1998). Many causes for this have been suggested including late presentation, the impact of comorbidity and lower treatment rates (Berrino et al., 2007; Horner et al., 2009). Surgical resection rates within Scotland are around 12% compared to figures often quoted as above 20% within Europe and the USA (Fry et al., 1999; Laroche et al., 1998; Cartman et al., 2002; Damhuis et al, 1996). Scotland frequently has lower recorded rates for both chemotherapy and radiotherapy. Patients in Scotland often present late (around 40 -50% will present with stage IV disease (Birring et al., 2005) and have significant comorbidity.

In our own previously published work, the prevalence of comorbidity was significant with 44% of patients having Chronic Obstructive Pulmonary Disease and 27% Ischaemic Heart Disease (Grose et al., 2011). Comorbidity has previously been shown to have an adverse impact upon available treatment options and subsequent outcome both in cancer in general and specifically within lung cancer (Potosky et al., 2004; Earle et al., 2000; Extermann, 2000; Charlson et al., 1987; Tammemagi et al., 2004; Janssen-Heijnen et al., 2004; Colinet et al., 2005; Birim et al., 2005; Imperatori et al., 2006; Asmis et al., 2008; Erridge et al., 2009; Grose et al., 2011) and it is likely that poorer outcomes in Scotland are due to a complex mixture of these factors.

There are few studies on the impact of comorbidity in lung cancer (Grose et al., 2011). Of these only one study used a tool specifically designed for lung cancer(Colinet et al., 2005). Most studies use the Charlson Co-morbidity Index, which was designed to assess comorbidity in patients with breast cancer (Charlson et al., 1987). In none of these studies is the grade or severity of disease fully assessed and, because of improved treatment options for some of the comorbidities may be out dated. Potential selection bias is also an issue as some of these studies only deal with operable patients. In addition, information on comorbidities is poorly recorded in routinely collected audit and cancer registry data and thus little is known about its characteristics among lung cancer patients in Scotland.

We know that treatments offered to patients with lung cancer in Scotland show inter-centre variability(Grose et al., 2011) (Tables4.1+4.2). However we do not know the reason(s) for this. The aim of our study was to prospectively describe the prevalence and severity of comorbidities in patients newly diagnosed with lung cancer. In addition we wished to assess the relationship between presence and severity of comorbidity with patient demographics, including a proxy for socioeconomic status, and the treatment offered.

## 5.2 <u>Methods</u>

Previously described in chapter 3

# 5.3 Statistical Analysis

All statistical testing was conducted at the 5% level. Unless otherwise stated, medians and interquartile range (IQR) are used. Where appropriate the chi-squared test for trend was used to assess for statistically significant differences, otherwise the Pearson chi-squared test was used. Statistically significant differences were set at  $p \le 0.05$ .

Statistical analyses were performed using SPSS v19.0 (SPSS Inc., Chicago, IL)

### 5.4 <u>Results</u>

### Demographics

882 patients were included in the study, comprising 297 from Aberdeen, 136 from Dunfermline, 285 from Glasgow, and 164 from Inverclyde. The ages ranged from 31 to 94 (median 72) and 487 (55%) were male. Baseline characteristics are previously described in Chapter 4.

24.5% of patients were diagnosed on the basis of clinical examination and radiological investigations alone, without histological evidence. This compares favourably with the National Lung Cancer Audit data, which had a median rate of 37% (Rich et al., 2011) of patients who did not have histological confirmation of lung cancer. We did not look at sub-groups of NSCLC histology.

### Comorbidity

In total 868 patients had their full comorbidity recorded. Comorbidities were found in 758 (87.3%) of patients at the time of diagnosis with lung cancer (Table 5.2). Only 110 (12.7%) patients had no comorbidity. Two or more comorbidities were found in 557 (64.2%) patients; 347 (40.0%) had three or more comorbidities and 4 (0.5%) had as many as 8 (Figure 5.1). The most frequent and most severe comorbidity was weight loss, which occurred in 53% of patients and had a mean score of 1.20.COPD was the next commonest comorbidity, present in 43% of patients and with a mean severity score of 0.76, followed by renal impairment (28%, mean 0.40) and IHD (27%, mean 0.36) (Table 5.1). The cumulative severity score in those with comorbidities ranged from 1-15 (mean 3.8 and median 3.0)

The most common co-existing diseases were weight loss and COPD (215 cases, 24% of 882 patients), weight loss with renal impairment (164 cases, 19%), COPD with renal impairment (154 cases, 17%), weight loss with IHD

(144 cases, 16%) and COPD with IHD (124 cases, 14%). The numbers of patients with aetiologically similar cardiovascular morbidities was low, with IHD and CVA concurring in 32 cases (4%), IHD and PVD in 37 cases (4%) and PVD and CVA in 14 cases (2%).The commonest co-existing diseases which both had grade 3 severity were severe weight loss with severe COPD (20 cases, 2%), severe renal impairment (9 cases, 1%) or severe IHD (9 cases, 1%) (Table 5.2).

Table 5.1. Distribution of comorbidities among 868 newly diagnosedlung cancer patients.

Comorbidity (missing data / %)	0	1	2	3	Mean
Weight loss (3/ 0.3%)	416 (47%)	131 (15%)	74 (8%)	258 (29%)	1.20
COPD (9/ 1%)	494 (57%)	146 (17%)	180 (21%)	53 (6%)	0.76
Renal impairment (7/ 0.8%)	634 (72%)	145 (17%)	79 (9%)	17 (2%)	0.40
IHD (4/ 0.5%)	645 (73%)	165 (19%)	50 (6%)	18 (2%)	0.36
CCF (4/ 0.5%)	759 (86%)	84 (10%)	31 (4%)	4 (0.5%)	0.18
Other Malignancy (3/ 0.3%)	755 (86%)	95 (11%)	28 (3%)	1 (0.1%)	0.18
Alcohol xs (5/ 0.6%)	774 (88%)	65 (7%)	20 (2%)	18 (2%)	0.18
Diabetes (5/ 0.6%)	776 (88%)	66 (8%)	32 (4%)	3 (0.3%)	0.16
CVA (4/ 0.5%)	800 (91%)	52 (6%)	18 (2%)	8 (1%)	0.13
PVD (4/ 0.5%)	800 (91%)	45 (5%)	27 (3%)	6 (0.7%)	0.13
Dementia (4/ 0,5%)	827 (94%)	24 (3%)	14 (2%)	13 (1%)	0.10



Figure 5.1. Histogram showing number of comorbid conditions per patient. 868 lung cancer patients in 4 Scottish centres, 2005-08.

# Table 5.2. Concurrence between individual comorbid conditions.868lung cancer patients in 4 Scottish centres, 2005-08.

(Left of diagonal for any grade. Right of diagonal for grade 3 of both conditions)

omorbidity	Comorbio	dity									
	Woight		Popal			Othor	Alcohol	Diabotos	CVA		Domonti
	weight	COPD	Renal	עחו	CCF	Other	Alcohol	Diabetes	CVA	PVD	Dementi
	loss		impairment			malignancy	XS				
Veight loss		20	9	9	2	0	5	0	4	2	4
COPD	215		0	0	0	0	1	0	0	0	0
Renal mpairment	164	154		0	0	0	0	0	0	0	0
HD	144	124	83		3	0	0	0	0	0	0
CF	76	62	43	90		0	0	0	0	0	0
Other	73	61	46	39	22		0	0	0	0	0
Nalignancy											
Alcohol xs	64	66	61	25	18	18		0	0	0	1
Jiabetes	56	47	47	44	26	24	24		0	0	0
ZVA	50	37	32	32	17	14	13	13		0	0
<b>'VD</b>	43	46	35	37	20	16	16	18	14		0
)ementia	42	33	33	24	12	12	17	15	18	10	

Using the cumulative comorbidity scoring index (Table 5.3) we found that 411 (47.4%) of patients with lung cancer had moderate (278 (32.0%) or severe comorbidities (133 (15.3%).

Table 5.3. Cumulative comorbidity scoring index. 868 lung cancer patients in 4 Scottish centres, 2005-08.

Range of	Group	Number of	Percentage of
cumulative		patients	patients
comorbidity			
score			
0	Nil	110	12.7%
1-3	Low	347	40.0%
4-6	Moderate	278	32.0%
≥ 7	Severe	133	15.3%

### Relationship between demographics and comorbidity (Table 5.4)

The presence of comorbidity in patients with lung cancer varied significantly between the 4 centres. The prevalence of severe comorbidity (index score of 3) was three times higher in patients treated in Dunfermline than the other three centres (p <0.001 Pearson chi-squared test)

### Relationship between socio-economic status and comorbidity (Table 5.4)

Significant variations in the level of co-morbidity between the relative socio-economic groups were seen. The most deprived patients had increased levels of comorbidity in comparison with those from more affluent backgrounds (p=0.026 Pearson chi-squared test)

### Relationship between cancer stage and comorbidity (Table 5.5)

The presence of comorbidity did not appear to be associated with stage of disease for either NSCLC (p=0.406) or SCLC (p=0.348)

<u>Relationship</u> between treatments offered for NSCLC patients and comorbidity (Table 5.4)

Among patients with NSCLC, there were significant associations between comorbidities and treatment. For NSCLC with early stage disease disease, patients offered surgery had lowest comorbidities scores, indicating fewer and less severe comorbidities, while those not offered curative treatment had most severe comorbidity scores (p = 0.006 Chi-square test for trend). Similarly, the least active palliative treatments were offered to those patients with most severe comorbidities (p<0.001 Pearson chi-squared test).

# Table5.4.The relationship between comorbidity grouping,demographics, socio-economic status, stage and treatment.

	Comorbid group % (n)					
	Nil	Low	Moderate	Severe		
Centre (n = 882) (p<0.0	01)					
Aberdeen (297)	21%	41%	27%	11%		
Glasgow (285)	8%	44%	35%	13%		
Inverclyde (164)	14%	44%	30%	12%		
Dunfermline (136)	3%	21%	41%	35%		
Socio-economic status (	(quintile) (n=	868) (p=0.026)				
1 (Most deprived) (265)	11%	42%	33%	14%		
2 (126)	14%	33%	36%	17%		
3 (248)	8%	43%	34%	15%		
4 (169)	18%	35%	31%	16%		
5 (60)	22%	48%	18%	12%		
Stage NSCLC (n = 525)	(p=0.406)					
l (65)	17%	<b>49</b> %	26%	8%		
II (43)	7%	44%	33%	16%		
Illa (48)	6%	48%	35%	11%		
IIIb (138)	14%	46%	31%	9%		
IV (231)	15%	44%	25%	16%		
Stage SCLC (n = 115)	(p=0.34	8)				
Limited (33)	25%	39%	21%	15%		
Extensive (82)	15%	32%	35%	18%		
Early Stage NSCLC (n=1	56) (p=0.006	5)				
Surgery (55)	18%	49%	29%	4%		
Radical Radiotherapy (32)	<b>9</b> %	47%	34%	10%		
No curative Option (69)	6%	46%	30%	18%		
Advanced Stage NSCLC (n = 362) (p=0.001)						
Palliative Chemotherapy (176)	18%	52%	25%	5%		
Palliative radiotherapy (88)	9%	36%	36%	19%		
Supportive care (98)	11%	39%	24%	26%		

#### 5.5 Discussion

We have found a high prevalence of comorbidity at presentation in patients with lung cancer. Comorbidities most frequently identified are weight loss (which was found in over 50% of patients), COPD, renal impairment and IHD. Most patients have at least one comorbidity and two thirds had two or more Given that most patients were current or ex smokers, fewer than expected had cardiovascular comorbidities. However, our findings are consistent with recent British Heart Foundation reports (British Heart Foundation, 2009). The overall severity of comorbidity was associated with socio-economic status and treatment centre. Importantly comorbidity status at presentation was associated with the treatment offered.

Treatment options for lung cancer, in particularly for advanced NSCLC, have significantly increased over the past decade (NICE 121, 2011) with a resultant improvement in survival. The trial data to support the use of such treatments is based on patients with good performance status. Significant comorbidity usually excludes a patient from enrolment into therapeutic clinical trials. We have demonstrated that worsening comorbidity detrimentally affects the likelihood of a patient being offered active treatment. It would seem logical to suggest that this is highly likely to have a negative impact upon survival.

We are not aware of any published studies that combine both the number and severity of comorbidities into a simple index. In addition no similar work has so clearly demonstrated a link between comorbidity, socioeconomic status and treatment offered. A recent study by Wang et al (2012) using the Charlson Comorbidity Index suggested that age rather than comorbidity was the most significant negative predictor of treatment. It did not look at outcomes or survival. The Adult Comorbidity Evaluation 27 (ACE-27) tool, initially validated in head and neck cancer was used in a prospective study including over 17,000 patients a large cohort of whom had lung cancer (Piccirillo et al., 2004). This showed that worsening co-morbidity significantly impacted negatively on survival, however sub-group analysis identified comorbidity as having the least prognostic information in lung cancer patients.

The major limitation of this study is the lack of survival data currently available. Although the cumulative comorbidity index was found to be associated with, centre, socio-economic status and active treatment rates, suggesting it has some validity, clearly significantly more work will be required to demonstrate a formal link with survival. This work is currently underway.

The important role that comorbidity has in affecting outcomes in lung cancer is beginning to be noted and the inclusion of basic data within the English National Lung Cancer Audit (NLCA) is welcomed. However we believe that more detailed recording is imperative to improve the understanding we have for comorbidity and the impact it has on lung cancer outcomes. Our study goes someway towards addressing this, currently unmet, need.

# CHAPTER 6: SIMPLE AND OBJECTIVE PREDICTION OF SURVIVAL IN PATIENTS WITH LUNG CANCER; STAGING THE HOST SYSTEMIC INFLAMMATORY RESPONSE.

### 6.1 <u>Introduction</u>

Within Scotland lung cancer remains the commonest cause of cancer related death (Berino et al., 2007). The prognosis is bleak with the median survival in advanced disease around four months from diagnosis (Berino et al., 2007). Survival compares unfavourably with other European countries and USA (Berino et al., 2007; Janssen-Heijnen et al. 1998). It has often been felt that the Scottish lung cancer population has more comorbidity with poorer performance status thus presenting fewer tolerable therapeutic options.

It is difficult to quantify the complex nature of patient frailty to provide a degree of objective assessment of fitness (Maltoni et al., 2005) and as a result prediction of survival in patients diagnosed with lung cancer remains problematical. Currently, prognosis is based upon a combination of stage of disease and performance status although other factors such as weight loss have been identified in the advanced cancer setting (Martin et al., 2010; Glare et al., 2004; Gripp et al., 2007). However, these host factors (weight loss and performance status) included in clinical decisions are recognized to be subjective in nature.

Recent work shows that the effect of systemic inflammation is detrimental in terms of outcome in cancer in general (McMillan, 2009; Proctor et al., 2011) and in lung cancer specifically (Wilop et al., 2006; Kock et al., 2009; Gagnon et al., 2010; Arrieta et al., 2010; Forrest et al., 2004; Forrest et al., 2005; Brown et al., 2007; Leung et al., 2012). The combination of C reactive protein and albumin when combined to calculate the modified Glasgow Prognostic Score (Table 6.1) has been previously been validated as an independent predictor of survival (McMillan. 2014).

The aim of the present study was to examine the clinical utility of this established objective marker of the systemic inflammatory response, the modified Glasgow Prognostic Score (mGPS), as the basis of risk stratification in patients with lung cancer.

## Table 6.1 The Glasgow Prognostic Scores

	Score
Glasgow Prognostic score (GPS)	
$CRP \leq 10mg/l and albumin \geq 35g/l$	0
CRP > 10mg/l or albumin < 35g/l	1
CRP > 10mg/l and albumin < 35g/l	2
modified Glasgow Prognostic Score	
(mGPS)	
$CRP \leq 10mg/l$ and albumin $\geq 35g/l$	0
CRP ≤ 10mg/l and albumin < 35g/l	0
CRP > 10mg/l	1
CRP > 10mg/l and albumin < 35g/l	2

## 6.2 <u>Methods</u>

### Previously discussed in chapter 3

### Survival data

Information on date of death was obtained via survival analysis undertaken by the Information Service Division (ISD) of NHS Scotland. Death records were complete until 1 June 2011, which served as the censor date for those alive.

### GPS/ mGPS

A venous blood sample was obtained at diagnosis for measurement of CRP concentration and albumin. The coefficient of variation for these methods over the range of measurement was less than 5%, as established by routine quality control procedures. The GPS was constructed as previously described (Table 6.1) (Forrest et al., 2003; McMillan et al., 2007). In brief, CRP more than 10 mg/L and albumin less than 35 g/dl were each given a score of 1. The GPS was calculated as 0, 1, or 2. Since hypoalbuminaemia alone in the absence of an increased CRP level did not confer a poorer cancer-specific survival in all patients with cancer (McMillan et al., 2007), the GPS was modified to assign a score of 0 in patients with hypoalbuminaemia alone (Table 6.1) (Crumley et al., 2008).

A number of recent studies have supported the use of mGPS in predicting outcome both in lung cancer and other tumour types (Wilop et al., 2006; Kock et al., 2009; Gagnon et al., 2010; Arrieta et al., 2010; Forrest et al., 2004; Forrest et al., 2005; Brown et al., 2007; Leung et al., 2012). It was our intention to stratify the group by mGPS and then analyse the impact of more conventional staging methods such as TNM stage and performance status.

## 6.3 <u>Statistics</u>

All statistical testing was conducted at the 5% level so 95% confidence intervals (CI) are reported throughout. Unless otherwise stated, medians and interquartile range (IQR) are used. The survival time defined as the number of months from study entry until death or if alive at follow-up date, was calculated. Univariate survival analysis was carried out using Kaplan-Meier method and the log rank test. Survival analysis was carried out using the Cox's proportional-hazards model and hazard ratios (HR) were calculated. Multivariate survival analysis was performed using a stepwise backward procedure to derive a final model of the variables that had a significant independent relationship with survival. To remove a variable from the model, the corresponding P-value had to be >0.10.

Statistical analyses were performed using SPSS v19.0 (SPSS Inc., Chicago, IL).

## 6.4 <u>Results</u>

In total, 882 patients from a number of different treatment groups were included in the study, comprising 297 from Aberdeen, 136 from Dunfermline, 285 from Glasgow, and 164 from Inverclyde composing; 59 patients were excluded from the study due to missing survival data (Figure 6.1). Baseline characteristics are previously described in chapter 4. The median age of participants was 72 years old. The majority was male, current or ex-smokers, of good performance status with advanced disease and had treatment with palliative intent.



Most had an elevated mGPS. The median follow up for survivors was 24.5 months (4.6 - 40.8). The median overall survival was 5.6 months (4.8 - 6.5). The 12-month survival rate was 30% (SE 2%).

Survival analysis using both GPS and mGPS was undertaken (Figures 6.2 + 6.3). Both were highly significantly associated with survival. Since the mGPS has been most extensively validated and readily extrapolated from previous work using C-reactive protein alone (McMillan et al., 2009; McMillan, 2014) it was used in the remainder of the analysis and to stratify the three groups.



Figure 6.2 - The relationship between the Glasgow Prognostic Score (GPS) (0 - 2, from top to bottom) and survival. GPS 0 vs. 1 (log rank p<0.001), GPS 1 vs. 2 (log rank p<0.001)



Figure 6.3 - The relationship between the modified Glasgow Prognostic Score (mGPS) (0 - 2, from top to bottom) and survival. mGPS 0 vs 1(log rank p<0.001), mGPS 1 vs 2 (log rank p<0.001)

The relationship between the mGPS and clinicopathological characteristics is shown in Table 6.2. There were 213 patients in the mGPS score of 0 group, 290 patients in the mGPS score of 1 group and 218 patients in the mGPS score of 2 group. The mGPS was associated with increasing deprivation (p<0.001), pack years smoking (p<0.001), poorer performance status (p<0.001), more weight loss (p<0.001), more advanced disease (p<0.001), less radical treatment (p<0.001) and poorer survival (p<0.001).

Table 6.2	The	relationship	between	the	mGPS	and	clinicopathological
characteri	stics	in patients w	ith lung c	ance	er		

Demographic		mGPS		P value
				(chi-square test)
	0 (n= 213)	1 (n= 290)	2 (n= 218)	_
Age (<60/60-69/70-	33/71/89/28	44/87/119/54	34/57/91/49	0.064
79/≥80 years)				
Sex Male/Female	115/108	174/130	133/98	0.201
Centre	98/10/77/38	146/25/66/67	43/16/126/46	<0.001
Aber/Dunf/Gla/Inver				
Deprivation	68/24/71/39/21	75/47/75/77/30	111/21/70/20/9	<0.001
Most - Least (quintile)				
Smoke (pack years)	21/24/126/37	11/33/177/69	13/21/128/57	0.014
NS/<20/20-60/>60				
Performance Status	42/96/60/22/3	26/130/105/40/3	5/51/88/66/21	<0.001
0/1/2/3/4				
Weight loss (%)	147/24/12/38	146/36/13/109	79/45/33/73	<0.001
0/<5/5-10/>10				
FEV1 (%)	138/31/45/7	181/48/53/22	130/33/47/17	0.126
<80/61-80/40-60/<40				
Local Symptoms	36/127	18/222	29/137	0.246
No /Yes				
Tumour Stage NSCLC	32/11/7/38/38	15/19/20/56/84	7/5/12/36/79	<0.001
I/II/IIIa/IIIb/IV				
Tumour Stage SCLC	13/21	10/32	8/18	0.471
Limited/Extensive				
Treatment				
Early Stage				
(Surgery/RT/no	26/9/38	17/15/43	4/2/18	<0.001
active)				
Advanced Stage				
(Chemo/RT/No	63/27/49	107/58/55	50/49/75	<0.001
active)				
Survival	1	1	1	
Alive/dead	34/181	16/274	8/211	0.003
12 month survival %	46 (4)	16 (2)	14 (3)	<0.001
(SE)				

The relationship between the clinico-pathological characteristics and survival in patients with an mGPS of 0 is shown in Table 6.3. The median survival was 13.2 (11.2 - 18.9) months. On univariate survival analysis, performance status (p<0.001), weight loss (p<0.01), stage of NSCLC (p<0.001), radical treatment offered (p<0.01) and palliative treatment offered (p<0.05) were all significantly associated with survival. On multivariate analysis, performance status (HR 1.69, 95% CI 1.39-2.06, p<0.001), weight loss (HR 1.18, 95% CI 1.04-1.33, p=0.009), stage of NSCLC (HR 1.06, 95% CI 1.01-1.23, p=0.017) and palliative treatment offered (HR 1.30, 95% CI 1.08-1.55, p=0.004) were independently associated with survival.

# Table 6.3 The relationship between parameters and survival in patients with mGPS = 0 (n=213)

Parameter	Patients	Univariate		Multivariate	
	Ν	HR (95%CI)	p-value	HR (95%CI)	p-value
	·				
Age (<60/ 60-69/70-					
79/≥80 years)	33/71/89/28	1.16(0.98-1.37)	0.077		
Sex Male	115	1			
Female	108	0.81(0.61-1.09)	0.169		
Centre Aberdeen	98	1	0.068		
Dunfermline	10	2.72(1.40-5.29)			
Glasgow	77	1.00(0.72-1.41)			
Inverclyde	38	1.13(0.75-1.70)			
-					
Deprivation (Most					
deprived = 1, Least					
Deprived = 5)					
1/2/3/4/5	68/24/71/39/21	0.94(0.85-1.05)	0.299		
Smoking history (NS = n	ever smoker, other	wise pack years)			
NS/<20/20-60/>60	21/24/126/37	0.93(0.77-1.12)	0.467		
Performance Status	-	-	_		
0/1/2/3/4	42/96/60/22/3	1.76(1.47-2.11)	<0.001	1.69(1.39-2.06)	<0.001
Weight loss (%Body wei	ght)			1	
0/<5/5-10/>10	147/24/12/38	1.21(1.08-1.37)	0.002	1.18(1.04-1.33)	0.009
FEV1 (%)					
>80/80-60/59-40/<40	138/31/45/7	0.94(0.80-1.12)	0.515		
Local Symptoms	1	•	1	1	
No/Yes	36/127	1.38(0.91-2.09)	0.124		
Stage	1	r	1	ſ	
NSCLC	32/11/7/38/38	1.38(1.25-1.53)	<0.001	1.06(1.01-1.23)	0.017
1/11/111a/111b/1V					
SCLC	13/21	1.05(0.97-1.14)	0.228		
Limited/Extensive					
Treatment					
Early Stage					
BSC/Surgery/RT	37/26/8	0.61(0.48-0.78)	0.001		
Advanced Stage					
Chemo/RT/BSC	59/27/47	1.22(1.01-1.46)	0.039	1.30(1.08-1.55)	0.004

The relationship between the clinico-pathological characteristics and survival in patients with an mGPS of 1 is shown in Table 6.4. The median survival was 6.1 (4.9 - 7.3) months. On univariate survival analysis, decreased age (p<0.01), performance status (p<0.001), weight loss (p<0.01), stage of NSCLC (p<0.001) and radical treatment offered (p<0.001) were significantly associated with survival. On multivariate analysis, performance status (HR 1.81, 95% CI 1.55-2.13, p<0.001), stage of NSCLC (HR 1.08, 95% CI 1.03-1.13, p<0.01) and radical treatment offered (HR 0.70, 95% CI 0.52-0.94, p<0.05) were independently associated with survival.

# Table 6.4 The relationship between parameters and survival in patients with mGPS = 1 (n=290)

Parameter	Patients	Univariate		Multivariate	
	N	HR (95%CI)	p-value	HR (95%CI)	-value
Age			p · alla c	···· (· •/•••)	
<60/ 60-69/70-79/≥80	33/71/89/28	1.24(1.09-1.40)	0.001		
Sex					
Male	174	1	0.171		
Female	130	0.84(0.66-1.08)			
Centre					
Aberdeen	145	1	0.425		
Dunfermline	23	0.89(0.55-1.43)			
Glasgow	55	0.99(0.72-1.36)			
Inverclyde	67	1.23(0.92-1.66)			
Deprivation(Quintile) (Most deprived = 1, Least Deprived = 5)					
1/2/3/4/5	75/47/75/77/30	0.99(0.90-1.08)	0.777		
Smoking history (NS - nev	er smoker otherwise r	ack years)			
$\frac{1}{10000000000000000000000000000000000$	11/33/177/60	1.02(0.86-1.22)	0 702		
1137 < 207 20-007 > 00	11/33/1//07	1.02(0.00-1.22)	0.792		
Performance Status					
0/1/2/3/4	26/130/105/40/3	1 83(1 57-2 14)	<0.001	1 81(1 55-2 13)	<0.001
0, 11 2, 3, 1	20/100/100/10/0	1.05(1.07 2.11)	0.001	1.01(1.00 2.10)	0.001
Weight loss (%Body weight	t)				
0/<5/5-10/>10	181/48/53/22	1.15(1.06-1.25)	0.001		
		· · · ·		·	•
FEV1 (%)		-	-		-
>80/80-60/59-40/<40	138/31/45/7	0.94(0.83-1.06)	0.313		
Local Symptoms	10/222	1 (0(0 01 2 90)	0.105		
No/ res	10/222	1.00(0.91-2.00)	0.105		
Stage					
NSCIC	15/10/20/56/84	1 31(1 10-1 45)	<0.001	1.08(1.03-1.13)	0.002
	13/17/20/30/04	1.51(1.19-1.45)	-0.001	1.00(1.05-1.15)	0.002
SCLC	10/32	0.51(0.23-1.11)	0.091		
Limited/Extensive					
Treatment					
Early Stage	43/17/15	0.55(0.41-0.72)	<0.001	0.70(0.52-0.94)	0.017
Best Supportive Care					
(BSC)/Surgery/RT					
Advanced Stage	107/58/55	1.07(0.90-1.26)	0.436		
Chemo/RT/BSC					

The relationship between the clinico-pathological characteristics and survival in patients with an mGPS of 2 is shown in Table 6.5. The median survival was 2.1 (1.5 - 2.7) months. On univariate survival analysis, centre (p<0.01), performance status (p<0.001), weight loss (p<0.001), stage of NSCLC (p<0.001) and radical treatment offered (p<0.01) were significantly associated with survival. On multivariate analysis, only performance status (HR 1.44, 95% CI 1.21-1.71, p<0.001) and weight loss (HR 1.13, 95% CI 1.00-1.28, p<0.05) were independently associated with survival.

# Table 6.5 The relationship between parameters and survival in patients with mGPS = 2 (n=218)

Parameter	Patients	Univariate		Multivariate	
	N	HR (95%CI)	p-value	HR (95%CI) r	o-value
Age			<b>P</b>		
<60/60-69/70-79/≥80	34/57/91/49	1.13(0.98-1.30)	0.084		
Sex					
Male	133	1	0.136		
Female	98	0.81(0.61-1.07)			
Centre	1			-	
Aberdeen	43	1	0.002		
Dunfermline	16	1.71(0.95-3.06)			
Glasgow	114	0.64(0.45-0.92)			
Inverclyde	45	0.88(0.57-1.35)			
Deprivation(Quintile)					
(Most deprived = 1)					
Least Deprived = $5$ )					
1/2/3/4/5	111/21/70/20/9	1.05(0.94-1.17)	0.408		
		, , , , , , , , , , , , , , , , , , ,			
Smoking history (NS = ney	ver smoker otherwise	nack years)			
NS/<20/20-60/>60	13/21/128/57	0.83(0.69-1.00)	0.062		
			01002		
Performance Status					
0/1/2/3/4	5/51/88/66/21	1.51(1.30-1.76)	<0.001	1.44(1.21-1.71)	<0.001
	4		•		
Weight loss (%Body weigh	nt)				
0/<5/5-10/>10	79/45/33/73	1.25(1.11-1.40)	< 0.001	1.13(1.00-1.28)	0.047
FEV1 (%)		-			
>80/80-60/59-40/<40	130/33/47/17	0.88(0.77-1.01)	0.073		
Local Symptoms					
No/Yes	29/137	1.51(0.98-2.33)	0.063		
Stage					
Slage			0.001		
NSCLC	7/5/12/36/79	1.32(1.15-1.50)	<0.001		
I/II/IIIa/IIIb/IV					
SCLC	8/18	1.01(0.91-1.12)	0.826		
Limited/Extensive					
Treatment					
I reatment	40/2/4		0.00(		
Early Stage	18/2/4	0.52(0.33-0.83)	0.006		
DSC/SURGERY/KI	E0/40/7E		0.264		
Auvanced Stage	50/49/75	0.92(0.78-1.10)	0.364		
CHEIHO/KI/BSC			1		

The relationship between mGPS, Performance status and survival at 1 year is shown in Table 6.6. When used in combination survival at 1 year varied from 72% (mGPS 0, PS 0) to 6% (mGPS 2, PS 3). The numbers in the PS 4 subgroup were too small to calculate accurately a survival rate.

Table 6.6 The relationship between m	nGPS and PS and	12 month survival
rate (%, SE).		

		Total number		
				of patients
PS	0	1	2	
0	72% (7)	50% (10)	20% (18)	71
1	65% (5)	31% (4)	30% (7)	259
2	49% (7)	19% (4)	16% (4)	245
3	9% (6)	5% (4)	6% (3)	122
4	NC	NC	NC	20
Total number of patients	212	290	218	

NC not calculated where N<10.

The relationship between mGPS, TNM stage (NSCLC patients only) and survival at 1 year is shown in Table 6.7. Survival varied from 69% (mGPS 0, Stage I NSCLC) to 2% (mGPS 2, Stage IV NSCLC).

Table 6.7 The relationship between mGPS and TNM Stage (NSCLC only) and 12 month survival rate (%, SE).

	mGPS			Total number
		of patients		
Stage	0	1	2	
1	69% (19)	28% (12)	NC	62
11	37% (14)	17% (9)	NC	47
Illa	20% (13)	42% (10)	0%	53
IIIb	20% (6)	6% (3)	14% (5)	154
IV	5% (3)	4% (2)	2% (2)	261
Total number of patients	153	225	173	

NC not calculated where N<10.

To stratify for stage, the relationship between mGPS and PS and survival at 3 months for those patients with advanced NSCLC (stage IIIb/IV) is shown in Table 6.8. Survival varied from 100% (mGPS 0, PS 0) to 23% (mGPS 2, PS3). The number of patients in the PS 4 group was too small to accurately calculate survival.

Table 6.8. The Relationship between mGPS and PS and 3 month survival rate (%, SE) in patients with TNM stage IIIb/IV NSCLC (n = 374).

	mGPS			Total
		number of		
		patients		
PS	0	1	2	
0	100% (0)	92% (8)	100% (0)	28
1	94% (4)	73% (6)	55% (9)	119
2	65% (11)	62% (7)	43% (8)	105
3	NC	29% (11)	23% (8)	55
4	NC	NC	NC	7
Total	61	134	111	
number of				
patients				

NC not calculated when N <10
This group was then further stratified to take into account the treatment offered. The relationship between mGPS and PS at 3 months for those patients with advanced NSCLC (stage IIIb/IV) undergoing palliative chemotherapy is shown in Table 6.9. Survival varied from 92% (mGPS 0, PS 1) to 50% (mGPS 2, PS 2). The numbers of patients in the PS 0 and 4 groups were too small to accurately calculate survival.

Table 6.9. The Relationship between mGPS and PS and 3 month survival rate (%, SE) in patients with TNM stage IIIb/IV NSCLC and receiving palliative chemotherapy (n = 138).

		mGPS		Total
				number of
				patients
PS	0	1	2	
0	NC	NC	NC	13
1	92% (7)	74% (8)	54% (14)	60
2	NC	69% (12)	50% (16)	33
3	NC	NC	NC	12
4	NC	NC	NC	0
Total	27	61	30	
number of				
patients				

NC not calculated when N <10

#### 6.5 Discussion

The results of the present study show for the first time that, in a large cohort of patients with lung cancer and using the mGPS as an objective basis for the prediction of survival, significant factors associated with survival varied significantly. Only performance status and to a lesser extent tumour stage were consistently shown to have independent prognostic value. Furthermore, the combination of the mGPS with either performance status or tumour stage effectively stratified the likely outcome in these patients. Therefore, this simple scheme based on objective criteria provides a new readily applicable approach to the routine clinical evaluation of patients with lung cancer.

In the present study it was of interest that weight loss, a well recognised poor prognostic factor, was inconsistently prognostic when included in the analysis with mGPS and performance status. This may suggest that much of the prognostic value of weight loss is attributable to the activation of the systemic inflammatory response and to the progressive loss of lean tissue leading to nutritional and functional decline (McMillan, 2009). Indeed, activation of the systemic inflammatory response resulted in a marked reduction in median survival of 13 months (mGPS 0) to 2 months (mGPS 2) independent of treatment received. This would suggest that the allocation of treatment was suboptimal and it may be that treatment allocated on a more objective scheme as proposed above will result in improved outcomes in all patients. For example, in those patients with mGPS of 2, neither stage nor treatment had independent prognostic value and therefore it would appear that such poor prognosis patients derive little benefit from standard anti-cancer. In particular a very honest appraisal of both benefits and toxicities of any treatment should be made with the patient irrespective of their tumour stage (MacDonald, 2012). However it must be noted that the very small numbers of patients in these groups (e.g. only 2 patients underwent surgery and 4 underwent radical radiotherapy) makes it very difficult to interpret and further studies looking only at radically treatable patients are advised.

The relationship between poor survival and systemic inflammation (the mGPS) remains poorly understood but it is likely to represent an objective marker of the chronic activation of the innate immune response with the consequent up-regulation of pro-inflammatory cytokines and growth factors and the resultant cancer cachexia (Scagliotti et al., 2008; Du Clos et al., 2004; Nozoe et al., 2000; Coussens et al., 2002; Abramovitch et al., 1999; Canna et al., 2008).

It is clear that, in Scotland, lung cancer continues to confer a very poor outcome with a median life expectancy of approximately 5 months. Even in early disease (TNM stage I/II NSCLC), with patients undergoing radical treatment with an expectation of cure (10 -15% of total number of patients within Scotland (Edge et al., 2009)) the 5-year survival is only around 30-60% (Edge et al., 2009). The advent of more advanced imaging modalities such as PET-CT (SIGN, 2005) has improved detection of occult metastasis leading to stage migration and less patients undergoing futile radical local treatment. Nevertheless, the present results highlight the importance of also staging the host systemic inflammatory response. The mGPS is a simple, cheap, and reproducible prognostic tool that has been shown to be a rational starting point for such work.

Since the initial work, a decade ago the combination of C-reactive protein and albumin, the Glasgow Prognostic Score (GPS/ mGPS), has been shown to have independent prognostic value in more than 60 studies (>30,000 patients with cancer). This prognostic value has been demonstrated in a variety of clinical scenarios, in particular primary operable cancer (McMillan, 2014) More recently in approximately 2,500 patients (Laird et al., 2013) and this present study have demonstrated that the mGPS has also clinical utility, together with performance status, in patients with advanced cancer.

In conclusion, the results of the present study confirm the independent prognostic value of the mGPS. In addition, it demonstrates the clinical utility of the mGPS combined with performance status to provide more objective risk stratification in patients diagnosed with lung cancer.

## CHAPTER 7: THE IMPACT OF COMORBIDITY UPON DETERMINANTS OF OUTCOME IN PATIENTS WITH LUNG CANCER.

#### 7.1 Introduction

Within Scotland lung cancer remains the commonest cause of cancer related death (Gregor et al., 2001). Survival lags significantly behind much of Western Europe and the United States (Gregor et al., 2001; Janssen-Heijnen et al., 1998). The cause(s) for this are not fully understood but are likely to include late presentation, the impact of comorbidity and lower treatment rates (Berrino et al., 2007; Horner et al., 2009; Fry et al., 1999; Laroche et al., 1998; Cartman et al., 2002; Damhuis et al., 1996; Birring et al., 2005). Our own, previously published, work has identified very high levels of comorbidity within an unselected lung cancer population within Scotland (Grose et al., 2011). It has also indicated that such comorbidity may play an important part in the decision to offer active treatment both in the radical and palliative setting (Grose et al., 2014). These findings have been supported by a number of studies, which have demonstrated the prognostic significance of comorbidities in many different types of cancer (Potosky et al., 2004; Earle et al. 2000; Extermann, 2000; Charlson et al., 1987; Tammemagi et al., 2004; Janssen-Heijnen et al., 2004; Colinet et al., 2005; Birim et al., 2005; Imperatori et al., 2006; Asmis et al., 2008; Erridge et al., 2009; Grose et al., 2011). However the mechanism of comorbidity upon cancer outcome is not clear. In addition there are often significant differences between the methods used to document and grade the severity of comorbidity (Grose et al., 2011)

The most widely quoted tool to assess comorbidity is the Charlson Comorbidity Index (Charlson et al., 1987) (CCI). This was designed in 1987

and assigned 19 conditions with a weighting index of 1 - 6 in attempt to quantify the likelihood of impact upon survival. Data was acquired via patients being admitted with medical conditions to a Washington Hospital, USA. This tool was initially validated in breast cancer patients with 10year mortality as an endpoint. It has also been validated in predicting progression-free survival (Charlson et al., 1987) in a variety of diseases such as breast and prostate cancer. However limitations with using this tool for lung cancer patients include absence of some potentially relevant diseases such as pulmonary fibrosis, the weighting of HIV is probably now less significant due to improved treatment and the lack of grading of severity of the specific disease(s) (Charlson et al., 1987; Grose et al., 2011)

In addition to stage of disease (Detterbeck et al., 2009) it is clear that the prognosis in lung cancer is determined by more than just comorbidity. Performance status is widely recognised as a very accurate predictor of survival in cancer patient (Martin et al., 2010; Mor et al., 1984; Yates et al., 1980). Recent work shows that the effect of systemic inflammation is detrimental in terms of outcome in cancer in general (McMillan, 2009; Proctor et al., 2011) and in lung cancer specifically (Wilop et al.; Koch et al., 2009; Gagnon et al., 2010; Arrietta et al., 2010; Forrest et al., 2004; Forrest et al., 2005; Brown et al., 2007; Leung et al., 2012). The combination of C reactive protein and albumin when combined to calculate the modified Glasgow Prognostic Score has previously been validated as an independent predictor of survival (McMillan, 2013). Two recent publications by Laird et al., 2013 and Bozzetti et al, 2014 have demonstrated that a combination of mGPS and PS is predictive in determining survival in advanced cancer patients and it is likely that a common pathophysiological association between all these factors determines outcome.

We prospectively investigated the survival of a large unselected lung cancer population assessing the impact of comorbidity along with more standard prognostic determinants. The goal of this study was to determine the role of a novel comorbidity scoring system (SCSS) and to compare it with the already established Charlson Comorbidity Index and the modified Glasgow Prognostic Score (mGPS). We also wished to explore the relationship between comorbidity, mGPS and PS. In addition we investigated a number of standard prognostic markers and demographics. This study aimed to determine which of these factors provided the most accurate information on survival.

#### 7.2 <u>Methods</u>

Previously discussed in chapter 3

#### 7.3 <u>Statistics</u>

All statistical testing was conducted at the 5% level so 95% confidence intervals (CI) are reported throughout. Unless otherwise stated, medians and interquartile range (IQR) are used. The survival time defined as the number of months from study entry until death or if alive at follow-up date, was calculated. Univariate survival analysis was carried out using Kaplan-Meier method and the log rank test. Survival analysis was carried out using the Cox's proportional-hazards model and hazard ratios (HR) were calculated. Multivariate survival analysis was performed using a stepwise backward procedure to derive a final model of the variables that had a significant independent relationship with survival. To remove a variable from the multivariate model, the corresponding P-value had to be >0.20.

Statistical analyses were performed using SPSS v21.0 (SPSS Inc., Chicago, IL).

#### 7.4 <u>Results</u>

In total, 882 patients from a number of different treatment groups were initially included in the study, comprising 297 from Aberdeen, 136 from Dunfermline, 285 from Glasgow, and 164 from Inverclyde. Baseline characteristics are previously described in chapter 4. The median age of participants was 72 years old. The majority of patients were male, current or ex-smokers, of good performance status with advanced disease and had treatment with palliative intent.

The median follow up for survivors was 24.5 months (4.6 - 40.8). The median overall survival was 5.6 months (4.8 - 6.5). The 12-month survival rate was 30% (SE 2%).

#### Impact of Comorbidity

The cumulative comorbidity score, SCSS, was highly significant (p<0.001). Survival with low levels of comorbidity had a hazard ratio of 1.29 (1.02-1.6) while for high levels of comorbidity it was 2.05 (1.55-2.70). The survival varied at 6 months from 60.2% for those patients with no comorbidity to 32.9% in those patients with high levels of comorbidity, while at 24 months the survival varied from 22.2% to 6.4%. In addition, the cumulative comorbidity score clearly increased as both PS and mGPS increased. These relationships are shown in Table 7.1 for the Early Stage group and Table 7.2 for the Advanced Stage group.

Table 7.1 - Median Cumulative Comorbidity Score (Range) as determined by PS and mGPS in the Early Stage group (Stage I-IIIa NSCLC). (no of patients). Missing data = 30 patients

	mGPS			
PS	0	1	2	
	1(5)	2.5 (3)	3.5 (5)	
0	14 patients	6 patients	2 patients	
	3 (5)	3 (9)	2 (4)	
1	23 patients	26 patients	9 patients	
	3.5 (8)	4 (9)	3 (4)	
2	12 patients	18 patients	10 patients	
	-	3.5 (6)	7 (0)	
3		4 patients	2 patients	
	-	-	9 (0)	
4			1 patients	

Table 7.2 - Median Cumulative Comorbidity Score (Range) as determined by PS and mGPS in the Advanced Stage group. (no of patients). Missing data = 135 patients

	mGPS				
PS	0	1	2		
	1(10)	1 (9)	2 (1)		
0	27 patients	19 patients	3 patients		
	2 (7)	3 (9)	3 (10)		
1	63 patients	95 patients	36 patients		
	3 (9)	4 (14)	5 (13)		
2	42 patients	82 patients	70 patients		
	4 (10)	4 (15)	4 (12)		
3	17 patients	35 patients	52 patients		
	1 (5)	3 (6)	6 (13)		
4	3 patients	3 patients	17 patients		

#### Early Stage group

The relationship between the clinico-pathological characteristics and survival in patients with a Early Stage cancer is shown in Table 7.3. This group included patients with NSCLC Stage I - IIIa , the intention in such patients is for potential cure of cancer. Patients who were not offered a curative treatment were still included in this group. The median survival was 16.2 (11.7-20.7) months. On univariate survival analysis, age (p=0.009), sex (p<0.001), Performance status (p<0.001), SCSS (p=0.017), mGPS (p<0.001) and treatment (p<0.001) were significantly associated with survival. On multivariate analysis, only sex (p<0.001), SCSS (p=0.002), mGPS (p<0.001) and treatment (p<0.001) were independently associated with survival.

## Table 7.3 - The relationship between parameters and survival in the Early Stage group (Stage I, II and IIIa NSCLC) (missing data refers to those without a survival endpoint)

Parameter	Patients		Univariate		Multivariate	
	Ν		HR (95%CI)	p-value	HR (95%CI)	p-value
Age (missing data = 10) Over	all p =0.009		• • •			
<60	13	1				
60-69	56	1.54	(0.79-2.99)	0.205		
70-79	60	2.09	(1.09-4.00)	0.026		
≥80	18	3.23	(1.62-6.43)	<0.001		
Cov (missing data 0)	-		(			
Sex (missing data =9)	07	4		0.001		.0.001
Male	8/		(0. 20. 0. 77)	<0.001	0.22 (0.20.0.52)	<0.001
Female	61	0.55	0(0.39-0.77)		0.32 (0.20-0.52)	
Centre (missing = 12) overall	p=0.559			1		
Aberdeen	59	1				
Dunfermline	27	1.22	.(0.77-1.96)	0.397		
Glasgow	39	1.31	(0.89-1.94)	0.175		
Inverclyde	23	1.24	(0.75-2.05)	0.395		
Deprivation (Quintile) (missing =9) p =0.584					1	<u> </u>
Most	38	1				
2	25	0.69	0(0.42-1.15)	0.154		1
3	43	0.09	0 49-1 16)	0.208		
	75	0.70	(0.17 1.10)	0.200		
4	31	0.76	0(0.47-1.20)	0.240		
Least	11	0.72	2(0.36-1.42)	0.341		
Smoking history (missing = 15	5) p=0.545			•		
NS	4	1				
<20	19	0.61	(0.24-1.53)	0.290		
20.60	97	0 53	(0 22 4 22)	0.125	-	
20-60	8/	0.53	(0.23 - 1.22)	0.135		
>60	32	0.59	(0.25-1.43)	0.247		
Performance Status (missing	9) p<0.001					1
0	33	1				
	35 (2		0.04.2.20	0.007		
1	62	1.38	6(0.84-2.28)	0.207		-
2	44	1./1	(1.02-2.85)	0.041		
3	8	4.38	8(2.29-8.40)	<0.001		
4	1	21.1	(2.67-167)	0.004		
			· · · ·			
Scottish Comorbidity Score (	missing = 10)					
p=0.017					F	
Nil	17	1				0.005
Low (1-6)	113	1.89	0(0.92-3.92)	0.084	1.66 (0.73-3.78)	0.227
High(7+)	17	3.66	(1.54-8.66)	0.001	5.10 (1.84-14.1)	0.002
3 ( )			(			
Charleson Comorbidity Index	c missing =10					
Overall p =0.341						
Nil	38	1				
Low (1-2)	83	1.21	(0.79-1.85)	0.391		
Mod (3-4)	23	1 45	0.87-2.41)	0.153		1
High (>5)	3	2 01	(0.83-4.88)	0.121		
		2.01	(0.00 1.00)			
mGPS (missing -37)	1	I		1		1
Overall $n < 0.001$						
	50	4				0.002
0	50	1				0.002
1	49	1.70	(1.12-2.58)	0.013	1.53 (0.92-2.54)	0.102
2	21	2 87	(1 60-4 99)	<0.001	3 24 (1 70-6 15)	<0.001
-	<u></u>	2.02			3.2 (1.70 0.13)	-0.001
Treatment (missing 0) Overa	l ll n <0 001	I		1	<u> </u>	1
	54	1				0.002
Juigery	J <del>4</del>	'				0.002
Radical RT	29	1.61	(0.92-2.81)	0.094	1.36 (0.70-2.66)	0.361
Pall Chemo	9	1.97	(0.91-4.11)	0.086	1.34 (0.58-3.08)	0.486
Pall RT	18	3.86	(2.06-7.23)	<0.001	3.55 (1.77-7.10)	< 0.001
BSC	38	2.43	6(1.44-4.09)	0.001	2.90 (1.49-5.66)	0.002

#### Advanced Stage group

The relationship between the clinico-pathological characteristics and survival in patients with an Advanced Stage cancer is shown in Table 7.4. This group included patients with advanced NSCLC (Stage IIIB/IV) and SCLC where the intention of treatment is almost always palliative. The median survival was 4.1 (3.5-4.8) months. On univariate survival analysis, age (p<0.001), centre (p=0.003), PS (p<0.001), SCSS (p<0.001), mGPS (p<0.001), and treatment (p<0.001) were significantly associated with survival. On multivariate analysis, only PS (p<0.001) and mGPS (p<0.001) were independently associated with survival.

Parameter	Patients	Univariate		Multivariate	
	Ν	HR (95%CI)	p-value	HR (95%CI)	p-value
Age (missing data = 25) Over	all p <0.001				
<60	105	1			
60-69	171	1.27(0.98-1.63)	0.067		
70-79	238	1.39(1.09-1.77)	0.008		
≥80	114	1.79(1.36-2.36)	<0.001		
Sex (missing data=24)					
Male	353	1	0.370		
Female	276	0.93(0.79-1.09)			
Centre (miss = 24) overall p=	0.003				
Aberdeen	225	1			
Dunfermline	92	1.52(1.18-1.94)	0.001		
Glasgow	172	0.99(0.81-1.22)	0.938		
Inverclyde	140	1.25(1.00-1.55)	0.046		
Deprivation (Quintile) (missing =24) p =0.545		1			
Most	184	1			
2	91	1.03(0.80-1.33)	0.820		
3	175	0.96(0.76-1.21)	0.633		
1	120		0 727		
4	129	0.89(0.76-1.21)	0.737		
Least	50	0.80(0.58-1.11)	0.182		
Smoking history (missing = 6)	1) n=0.902				
NS	37	1			
<20	57	1 11(0 73 1 60)	0.617		
<20	00	1.11(0.73-1.09)	0.017		
20-60	359	1.00(0.71-1.41)	0.984		
>60	136	1.04(0.72-1.51)	0.824		
Performance Status (missing	24) p<0.001			1.	
0	60	1		1	<0.001
1	211	1.57(1.15-2.14)	0.004	1.70 (1.19-2.42)	0.004
2	212	2.55(1.87-3.48)	< 0.001	2.57 (1.78-3.71)	< 0.001
3	119	5 32(3 80-7 45)	<0.001	5 03 (3 34-7 58)	<0.001
 Д	27	9.08(5.66-14.56)	<0.001	8 71 (4 86-15 6)	<0.001
	27	7.00(5.00 11.50)	10.001	0.71 (1.00 15.0)	-0.001
Scottish Comorbidity Score ( p<0.001	missing = 35)				
Nil	79	1			
Low (1-6)	430	1.33(1.03-1.72)	0.028		
High (7)	100		<0.001		
(1)gii(7+)	109	1.95(1.45-2.02)	<0.001		
Charleson Comorbidity Index Overall p =0.271	c missing =30.				
Nil	219	1			
Low (1-2)	294	0.99(0.83-1.19)	0.933		
Hed (2.4)	02	1 02(0 70 4 20)	0.001		
MOU (3-4)	92	1.02(0.79-1.30)	0.901		
	10	1.00(1.03-2.09)	0.039		
mGPS (missing =103) Overall p <0.001		•			•
0	148	1		1	<0.001
1	227	1.60(1.28-1.98)	<0.001	1.51 (1.21-1.89)	<0.001
2	175	2 20(1 00 2 01)	-0.004		0.001
∠ Treatment (missing 24) Over	all p <0.001	2.37(1.70-3.01)	<0.001	1.54 (1.20-1.99)	0.001
				1	
Radical RT	41	1			
Pall Chemo	224	1.65(1.15-2.36)	0.006		
		. ,			
Pall RT	143	2.58(1.78-3.76)	<0.001		
BSC	221	2.18(1.52-3.12)	<0.001		

Table 7.4 - The relationship between parameters and survival in the Advanced Stage group (Stage IIIb/IV NSCLC, SCLC and not staged) (missing data refers to those without a survival endpoint)

#### 6.5 **DISCUSSION**

This large unselected population based cohort study of lung cancer patients has demonstrated that a number of important factors have significant impact in terms of survival. It has gone further by showing that the factors, which influence survival, are different, depending upon the stage of cancer at diagnosis and the potential treatment strategy. The novel comorbidity scoring system, the SCSS has compared very favourably with the more established CCI. In addition, it has confirmed the very poor, general outlook for lung cancer patients with a median survival of only 5.6 months.

Our results have shown that in those patients whose cancers can be considered radically treatable female sex, a normal mGPS, low levels of comorbidity based upon the SCSS and being offered surgery were all associated with better survival. Female sex and being offered surgery have both previously been demonstrated to improve outcome in localized NSCLC (Rich et al., 2011; NICE, 2011). Since the initial work, a decade ago the combination of C-reactive protein and albumin, the Glasgow Prognostic Score (GPS/ mGPS), has been shown to have independent prognostic value in more than 60 studies (>30,000 patients with cancer). This prognostic value has been demonstrated in a variety of clinical scenarios, in particular primary operable cancer (McMillan, 2013; Grose et al., 2014). Comorbidity has often been identified as a cause of poor survival in lung cancer and the finding of the high levels of comorbidity and the impact upon survival of the SCSS is supported by many previous studies (Deleuran et al., 2013; Luchtenborg et al., 2012; Ahn et al., 2013).

In those patients whose cancer was suitable only for palliative treatment options only a higher PS and elevated mGPS, indicating systemic inflammation, were associated with poorer outcome. Both of these have been demonstrated previously to have a significant impact upon outcome (Laird et al., 2013; Grose et al., 2014). The very poor outcome of the palliative patients, with a median survival of only 4.1 months, may be the reason why factors such as comorbidity did not appear to associated with survival. Even active palliative treatment such as chemotherapy did not appear to benefit survival across the population when other factors were accounted for in the multivariate analysis. Although chemotherapy has been shown to confer a survival benefit in large randomized trials (SIGN, 2005; NICE, 2011), this is obviously in good PS patients with low levels of comorbidity and this paper supports most clinicians assertions that palliative chemotherapy should only be offered to the fitter group of patients with advanced lung cancer.

It is perhaps surprising that the CCI was not associated with survival in either cohort. However several recent studies have also found no or little relationship (Wang et al., 2012; Ganti et al., 2011; Girones et al., 2011). The reason(s) for this are not particularly clear. However it is important to note that the CCI is now nearly 30 years old and its validation came primarily in a North American breast cancer cohort and thus may not be relevant to the Scottish lung cancer population. The weighting of some diseases e.g. AIDS, indicating the previous lack of treatment and poor survival, have significantly changed and it may be that novel scoring systems such as the one described above may have more validity in the current setting.

In addition we have demonstrated a clear trend in rising comorbidity score as both PS and mGPS rise. As demonstrated above all three of these factors appear to be determinants in patient survival. This suggests that there is a complex relationship between such factors and to the best of our knowledge this relationship has not previously been demonstrated in the literature. One limitation of this current study is the lack of reason for the treatment offered. However we intend to address this in a future paper. In conclusion we have identified that a variety of factors are independent prognostic determinants of outcome in lung cancer. There appear to be clear differences between the radical and palliative groups. Especially in the select group of patients in whom cure of cancer is being attempted it is clear that a more detailed assessment of fitness including comorbidity and evidence of systemic inflammation is required before embarking upon a treatment plan. As the percentage of elderly patients who by definition may have more comorbidity increases then this will become even more important in the future.

### CHAPTER 8: IMPACT OF THE TREATMENT DECISION PROCESS UPON SURVIVAL IN LUNG CANCER

#### 8.1 Introduction

Lung cancer is the commonest cancer in Scotland and the second commonest in the UK (Office for National Statistics, 2006; ISD, 2009). Although one and two year survival has improved over the past decade, five year survival remains significantly poorer in Scotland than in comparable Western European countries or the USA, with 5-year relative survivals of 8.0%, 10.2% and 16.3%, respectively. Variations in treatment may explain some of the observed differences in lung cancer survival between countries (Berrino et al., 2007; Horner et al., 2009).

Treatment options are dependent upon pathology, TNM staging, patient fitness and choice. Radical (i.e. potentially curable) treatment options are surgery or radical radiotherapy (+/- chemotherapy). Patients with Stage I + II Non Small Cell Lung Cancer (NSCLC) should be considered for curative surgery whenever possible as it confers the highest chance of cure. 5 Year survival of 54-80% for patients with St 1a and 38-65% for patients with Stage 1b has been reported (Suzuki et al., 1999; Makitaro et al., 2002; Inoue et al., 1998). In patients with localized NSCLC not suitable for surgery then radical radiotherapy (+/- chemotherapy) is an appropriate option. Meta-analysis of retrospective studies in patients with Stage 1/II has shown overall survival from 50-93% at one year and 0-42% at 5 years (Rowell et al., 2004). In recent years the option of Stereotactic Ablative Radiotherapy (SABR) has become a very reasonable alternative for those with early stage disease and in many centers' has replaced conventional radiotherapy in this setting. (Palma et al., 2011; Lagerwald FJ, 2012)

In patients with advanced disease and a Performance Status (PS) <2 then systemic chemotherapy with a platinum based doublet has been shown to confer a survival benefit of approximately 2 - 4 months (Goffin et al., 2010). Recent therapeutic advances including tyrosine kinase inhibitors (in those with an EGFR mutation) (Rosell et al., 2012) and pemetrexed/platinum doublets in Adenocarcinoma (Scagliotti et al., 2008) have conferred further improvement in survival but these treatments were not available at the time of this study.

A significant proportion of lung cancer patients are not suitable for more aggressive therapies such as resection or radical radiotherapy in localized disease or indeed palliative chemotherapy in the more advanced setting (Grose et al., 2014; Grose et al., 2015; Havlik et al., 1994; NLCA 2013). Clearly this will significantly impact upon survival. Guidelines such as the National Institute for Clinical Excellence (NICE Guideline 121) and the Scottish Intercollegiate Guidelines Network (SIGN Guideline 137) have stated that a patient's suitability for a particular treatment option should be based upon an assessment including fitness, comorbidity and PS and that options such as surgery or radical radiotherapy in the curative setting or chemotherapy in the palliative setting are only suitable for those with an adequate level of fitness and PS.

Previous work undertaken by this group has supported the available guidelines in suggesting that PS, comorbidity and late stage at time of presentation all significantly impact upon survival (Grose et al., 2014; Grose et al., 2015).

The purpose of this study was to determine the reasons for clinician's decision-making process and if these reasons did indeed mirror the individual patient's demographics, fitness and stage. We therefore evaluated demographics, PS and comorbidity in both the palliative and radically treatable patient groups. Our aim was to determine if treatment offered and the evidence available substantiated the reason for that treatment being offered.

#### 8.2 <u>Methods</u>

Previously discussed in chapter 3

#### 8.3 <u>Statistics</u>

All statistical testing was conducted at the 5% level so 95% confidence intervals (CI) are reported throughout. Unless otherwise stated, medians and interquartile range (IQR) are used. The survival time defined as the number of months from study entry until death or if alive at follow-up date, was calculated. Univariate survival analysis was carried out using Kaplan-Meier method and the log rank test.

Statistical analyses were performed using SPSS v21.0 (SPSS Inc., Chicago, IL).

#### 8.4 <u>Results</u>

In total, 646 patients from a number of different treatment groups were initially included in the study, comprising 218 from Aberdeen, 91 from Dunfermline, 194 from Glasgow, and 143 from Inverclyde. Baseline characteristics are shown in Table 8.1. The median age of participants was 70 years old. The majority of patients were male, current or exsmokers, of good performance status with advanced disease and had treatment with palliative intent.

The median duration of follow up of survivors was 52.6 months (43.2 - 58.9). The median overall survival was 6.1 months (95% CI 5.3 - 7.0). The 12 months survival was 32% (SE 2). The median duration of follow up of survivors is significantly increase compared to in previous chapters due to updating of the survival data.

Parameter	
Age (years)	Median 70.0 years (range 31-92)
Sox Malo	277 (59 4%)
Sex Male	377 (30.4%) 260 (41.6%)
Tenlate	
Smoking (missing = $30$ )	
Never smoker	34 (5 3%)
<20 Pack Years	68 (10.5%)
20-60 Pack Years	378 (58.5%)
> 60 Pack Years	136 (21.1%)
Centre	
Aberdeen	218 (33.7 %)
Dunfermline	91 (14.1%)
Glasgow	194 (30.0%)
Inverclyde	143 (22.1%)
Deprivation (Quintile)	
Most deprived	197 (30.5%)
2.00	94 (14.6%)
3.00	183 (28.3%)
4.00	132 (20.4%)
Most affluent	40 (6.2%)
Performance Status	
0	87 (13.5%)
1	252 (39%)
2	204 (31.6%)
3	91 (14.1%)
4	12 (1.9%)
Co morbidity Score (missing = 6)	
Nil	83 (12.8%)
Low	466 (72.1%)
High	91 (14.1%)
Staging	
	66 (10.2%)
	43(0.7%)
	40(7.4%)
	(21.7%)
NSELC IV	
SCI C Limited	33 (5 1%)
	82 (12 7%)
Treatment	
Surgery	56 (8.7%)
Radical Radiotherany	74 (11 5%)
Palliative Chemotherany	232 (35.9%)
Palliative Radiotherapy + BSC	123 (19.0%)
Best Supportive Care (BSC)	161 (24.9%)

# Table 8.1Baselinecharacteristicsofpatientswithlungcancer(n=646)

#### Stage I -IIIa NSCLC

156 patients in total were included in this group of whom a treatment decision is available for 148. 88 patients were offered radical treatment with either surgery or radical radiotherapy and 68 were deemed suitable for palliative treatment, which included chemotherapy, palliative radiotherapy and/or best supportive care (bsc). The median survival was 29.9 months (95% CI 18.2-41.6) and 11.3 months (95% CI 5.9-16.6) respectively (P<0.001).

For those patients not undergoing radical treatment the reasons as recorded at the MDT were assessed. These were then compared with the actual results, for example, in whom the PS was recorded as the reason for suboptimal therapy the actual breakdown of PS was shown. Some patients may have more than one reason recorded.

In the group in which age was a determining factor all were over 70 with the vast majority (69.2%) being 80 or over. In the group in which PS was a factor only 11.1 % had a PS of 1, of the remainder 70.4% had a PS of 2 with 22.5% being PS 3/4. In the group in who comorbidity was a determining factor 77.5% had low levels of comorbidity while 22.5% had high levels. In 7 patients radical therapy was refused (Table 8.2).

Table 8.2Breakdown of determining factors for sub-optimal therapyin Stage I-IIIa NSCLC.

Note - Patients may have more than one reason for sub-optimal therapy.

Reason given	Number	Actual	Grouping	Number (%)
for Sub-				
optimal				
treatment				
	I	I	I	
Age	13	Age	70 - 79	4 (30.8%)
			> 79	9 (69.2%)
	I	I		•
PS	27	PS	1	3 (11.1%)
			2	19 (70.4%)
			3	4 (18.8%)
			4	1 (3.7%)
	I	I	L	
Comorbidity	40	Comorbidity	Low	31 (77.5%)
			High	9 (22.5%)
	1	1	1	1
Patient	7			
refusal				

There was a statistically significant difference in age (p <0.001), PS (p<0.001) and comorbidity (p=0.006) between the group offered radical treatment and those offered palliative therapy with the palliative group being older, having a higher PS and more comorbidity (Table 8.3). Deprivation, as based upon the SIMD quintile, was not statistically significant (p=0.379).

# Table 8.3Comparison of factors between those being offered radicaland palliative therapy in Stage I-IIIa NSCLC. (n =156)

	Radical Treatment offered	Palliative treatment offered	
	(N=88)	(N=68)	
			Р
			value
Age group (%)	• •		<0.001
missing=1			
<60	10 (11.4%)	4 (5.9%)	
60-69	38 (43.2%)	19 (27.9%)	
70-79	37 (42%)	28 (41.2%)	
>80	3 (3.4%)	17 (25%)	
		•	
PS (%)			<0.001
0	27 (30.7%)	7 (10.1%)	
1	48 (54.5%)	19 (27.5%)	
2	11 (12.5%)	36 (52.2%)	
3	2 (2.3%)	6 (8,7%)	
4	0	1 (1.4%)	
Comorbidity	(%)		0.006
missing = 1			
nil	13 (14.9%)	4 (5.8%)	
low	69 (79.3%)	53 (76.8%)	
high	5 (5.7%)	12 (17.4%)	

#### Stage IIIb/IV NSCLC + SCLC

447 patients were included in this group. 447 patients had a treatment decision available. 222 patients were offered chemotherapy while 225 were offered best supportive care (bsc) +/- palliative radiotherapy. The median survival was 6.4 months (95% CI 5.2-7.5) and 2.4 months (95% CI 2.0 -2.9) respectively (P<0.001).

In those patients who were deemed inappropriate for chemotherapy due to age the vast majority were over 80 (65.7%). In the group in which PS was a factor 55.2% had a PS of 3 or 4, 39.2% had a PS of 2 while 5.6% had a PS of 1. No patient had a PS of 0. For those patients in whom comorbidity was a determining factor 4.6% had no recorded comorbidity, 59.3% had low levels of co morbidity identified as a reason and 36.1% had high levels. 16 patients refused palliative chemotherapy. (Table 8.4)

Table 8.4Breakdown of determining factors for sub-optimal therapyin Stage IIIb/IV NSCLC + SCLC

Note - Patients may have more than one reason for sub-optimal therapy.

Reason given	Number	Actual	Grouping	Number (%)
for Sub-				
optimal				
treatment				
	I	1		
Age	35	Age	60-69	1 (2.9%)
			70 -79	11 (31.4%)
	I	I	>79	23 (65.7%)
PS	125	PS	1	7 (5.6%)
			2	49 (39.2%)
			3	58 (46.4%)
			4	11 (8.8%)
		•		
Comorbidity	108	Comorbidity	Nil	5 (4.6%)
			Low	64 (59.3%)
			High	39 (36.1%)
Patient	16			
refusal				

There was a statistically significant difference in age (p <0.001), PS (p<0.001) and comorbidity (p<0.001) between the group offered radical treatment and those offered palliative therapy with the palliative group being older, having a higher PS and more comorbidity (Table 8.5). Deprivation was again not statistically significant (p=0.368)

Table 8.5 Comparison of factors between those being offered chemotherapy and supportive care in Stage IIIb/IV NSCLC + SCLC. (n =447)

	Chemotherapy offered	Supportive care (+/- palliative	
		RT) offered	
	(N=222)	(N=225)	Р
			value
Age group (%	6)		<0.001
<60	53 (23.9%)	28 (12.4%)	
60-69	83 (37.4%)	52 (23.1%)	
70-79	71 (32%)	97 (43.1%)	
>80	15 (6.8%)	48 (21.3%)	
PS (%)			<0.001
0	31 (14%)	13 (5.8%)	
1	109 (49.1%)	50 (22.2%)	
2	66 (29.7%)	84 (37.3%)	
3	16 (7.2%)	67 (29.8%)	
4	0	11 (4.9%)	
Comorbidity	(%)		<0.001
missing = 4			
nil	35 (15.9%)	22 (9.9%)	
low	165 (75%)	147 (65.9%)	
high	20 (9.1%)	54 (24.2%)	

#### 8.5 Discussion

Survival in lung cancer remains very poor; our study population, in which the median survival is 6.1 months, confirms this. This is despite advances in surgical techniques, new radiotherapy options such as SABR and novel chemotherapy agents. In a significant number of patients the treatment options will be reduced due to poor fitness, PS, comorbidity and advanced stage at presentation. However a number of publications (Blum et al., 2014; Forrest LF et al., 2014; Koshy et al., 2015) have suggested that treatment disparities are a complicated and multifactorial issue and that clinician selection bias may have at least some influence. There has also been a recent publication suggesting that social class / deprivation may also influence treatment choice (Forrest LF et al., 2014) but this has not been identified in our study.

We wished to determine the clinician's reason(s) for treatment offered and how that compared with the actual objective fitness, PS, comorbidity and age of the patients.

We have shown that for the majority of patients, both in the early and advanced stage at presentation, the treatment decision appears to be appropriate given the recorded fitness, PS and comorbidity. However in a small but significant number of patients there did appear to be discrepancies between the clinician's reasons for sub-optimal therapy and the recorded objective assessment of the patient in question. We identified low levels of comorbidity as a reason for not offering either radical treatment in the early stage group or chemotherapy in the advanced stage group. A previous publication (Grose et al., 2015) undertaken by ourselves demonstrates that low levels of comorbidity has little impact upon survival and that only high levels of comorbidity influence outcome significantly and it may be that clinicians are being unduly influenced by the clinically irrelevant comorbidity.

It is not clear to what extent such potential under treatment will affect overall survival of this group of patients but as we are faced with the challenges of significant advances in therapy in an increasingly old population it is clear that more detailed objective assessment of patients is required to ensure that all are offered the most appropriate therapy. Significant work (Laird et al., 2013; McMillan. 2013) has been undertaken utilizing markers of systemic inflammation such as the modified Glasgow Prognostic Score (mGPS) along with PS to predict outcome and survival in advanced cancer including lung cancer and this has demonstrated additional prognostic benefits over more conventional assessments. Such tools together with our own extensive work in co morbidity should go some way to addressing this increasing challenge.

#### CHAPTER 9: CONCLUSION AND FUTURE WORK

#### 9.1 <u>Conclusions</u>

The overall aim of the Thesis was to examine the impact of comorbidity in lung cancer, to attempt to quantify the extent and severity of comorbidity and to explore its relationship with treatment and survival.

This work has demonstrated that it is possible to collect very detailed audit data on a large number of patients across several centres in Scotland in real time during the MDT process and to prospectively analyse survival data.

During the course of this work, a number of other publications have explored the role of comorbidity upon outcome in Lung Cancer. A recent prospective observational study (Calvo-Espinos et al., 2015) showed lower survival rates among lung cancer patients with higher age-adjusted Charlson Comorbidity Index, although numbers were small (n=66). A far larger population based cohort study of 5,683 patients captured using the Nebraska Cancer Registry demonstrated that the presence of comorbid conditions was associated with reduced survival, however it did not attempt to grade severity of the conditions (Islam et al., 2015). A specific study of SCLC patients utilizing the Netherlands Cancer Registry also supported the view that increased comorbidity had a negative prognostic effect on survival (Aarts et al, 2015). However a retrospective study in Denmark of 20,552 non-surgically treated patients from 2005 - 2011 (Mellemgaard et al., 2015) appeared to show that comorbidity (utilizing the Charlson Comorbidity Index) had a limited effect upon survival. It is difficult to collect retrospective comorbidity data and the authors acknowledged this. This only serves to highlight the need for comorbidity assessment to be taken prospectively and at the time of diagnosis.

A further retrospective study utilizing the ACE -27 comorbidity scoring system in a variety of tumours including lung cancer, in patients undergoing radiotherapy, (Owen et al., 2014) demonstrated an association between increased comorbidity and reduced active treatment rates. Its main conclusion was that further tumour specific, validated tools would be beneficial. Two recent studies have made use of the extensive Danish Lung Cancer Registry to assess the role of the CCI. lachina et al(2014) explored the impact of the CCI on survival in NSCLC in 3135 patients diagnosed in 2010. It showed that higher levels of comorbidity were associated with reduced survival, however there was significant missing data, which the authors acknowledged, was related to the retrospective nature of the work. The second paper (Deleuran et al., 2013) was a population based cohort study from 2000 to 2011 including 9369 patients. It demonstrated a small but significant increase in survival over this time period except in those patients with high levels of comorbidity. The conclusion for this was that those patients with high levels of comorbidity are unlikely to be offered the newer treatment options such as doublet chemotherapy or radiotherapy due to poorer fitness. The Simplified Comorbidity Score (SCS) had previously been discussed in the initial literature review in chapter 2. A recent paper by Ball et al (2013) retrospectively analysed 921 lung cancer patients to assess the impact of the SCS on survival. However when age, PS and stage was taken into account the SCS had no additional effect upon survival.

The link between comorbidity (utilizing the CCI), socio-economic status and outcome has been assessed in two recent papers. In the first 15,582 patients in the South of England with Lung Cancer (Berglund et al., 2012) were retrospectively analysed. It was demonstrated that levels of comorbidity increased with poorer socioeconomic status. 29.6% of patients had high levels of comorbidity in the most deprived quintile vs 23.8% in the least deprived quintile (p<0.001). It also revealed an associated trend towards survival differences in these groups. The second paper by Woolhouse (2011) also highlighted the previously discussed differences between the UK and the remainder of Europe with regards to lower levels of treatment and poorer outcomes and identified the need for further research to identify reasons for such differences.

As can be seen from the above updated assessment of the literature there remains an unmet need for detailed research into comorbidity in lung cancer and its influence upon outcome. It has been clearly highlighted by much of the research that prospective work to validate a novel tool would be hugely beneficial.

The results of the work presented in this Thesis have demonstrated the significant prevalence and severity of comorbidity through utilization of a novel comorbidity scoring system, the Scottish Comorbidity Scoring System (Grose et al., 2014). In addition it has shown that there appears to be a relationship with both treatment offered and survival, particularly in patients who have potentially curable lung cancer (Grose et al., 2014).

During the course of this work there have been significant developments in our understanding of Lung Cancer. At the time of the work commencing NSCLC was seen as a single entity. However we now see this as several very specific sub-types with particular targeted mutations such as Epidermal Growth Factor (EGFR) and Anaplastic Lymphoma Kinase (ALK) with associated varying treatment options (Rosell et al., 2012; Solomon et al., 2014) leading to potentially different outcomes. Any future work would need to very take into account such advances.

National data has demonstrated that a significant percentage of patients initially present as an emergency admission and that such patients have a poorer outcome (Rich et al., 2011). Unfortunately the data capture in this work did not include a breakdown of mode of presentation. This is a weakness of this work as there would be significant interest to explore if there are differences in comorbidity, socio-economic status, systemic
inflammation and outcomes between those patients who present as an emergency and those who are referred via primary care.

As discussed in Chapter 6, there has been significant research recently undertaken both in cancer in general and lung cancer in particular exploring the role of systemic inflammation upon outcomes and survival (Scagliotti et al., 2008; Du Clos et al., 2004; Nozoe et al., 2000; Coussens et al., 2002; Abramovitch et al., 1999; Canna et al., 2008).

The prognostic value of the combination of CRP and albumin, the (modified) Glasgow Prognostic Score (McMillan, 2014) has been demonstrated across a large number of clinical scenarios including lung cancer. Initially this has been seen as a predictive marker of outcome. However several new studies of a novel targeted agent, Ruxolitinib which is a Janus Kinase Inhibitor, are exploring treating patients with evidence of systemic inflammation indicating a potential role for stratifying patients by mGPS (Hu et al., 2014).

It is of interest that particularly in those patients undergoing radical treatment, both mGPS and SCSS appear to impact upon survival in a multivariate model suggesting they both play an important role in determining outcome (Grose et al., 2014).

Previously the most widely used determinant of comorbidity was the Charlson Comorbidity Index (CCI) and it noteworthy that in this cohort it did not appear to have a statistically significant relationship with outcome. The reasons for this are not clear but as previously discussed the CCI is now almost 30 years old and treatment and prognosis of a number of the diseases has dramatically changed in this period. It is clear that an updated Comorbidity Scoring System taking account of these changes is required and this work has demonstrated that the SCSS is one such potential tool. It is encouraging that within this population it appears that the majority of patients were offered appropriate treatment as per national guidelines. It is likely that the MDT provides a strong role in this and this work supports the assertion that all lung cancer patients should be discussed within a specialist MDT setting with access to both thoracic surgical and clinical oncology input to ensure that radically treatable patients are discussed with the relevant clinicians.

Despite this the outcome for patients generally remains very poor. The median overall survival of 6.1 months (95% CI 5.3 - 7.0) and the 12 months survival of 32% are stark reminders to all clinicians treating this disease of the very poor outcome for the vast majority of patients and their family and carers. As has been shown in Chapter 8 a small but significant number of patients appear to have been offered potentially sub optimal therapy based perhaps upon inaccurate clinical assessments and this reinforces the necessity for development of new clinical tools to help in the decision making process. The SCSS has been validated within this clinical setting to be easily recordable and provide clinically meaningful prognostic information. It is likely that the combination of such a comorbidity tool along with PS, stage and assessment of systemic inflammation (mGPS) will in the future provide increasingly accurate assessment of patients to aid in the increasingly complex and challenging decision making process.

# 9.2 <u>Utilisation of the Scottish Comorbidity Scoring System + Proposed</u> Future Work

The aim of the research leading to this Thesis was to examine the impact of comorbidity in lung cancer, to attempt to quantify the extent and severity of comorbidity and to explore its relationship with treatment and survival. In addition the development and validation of the novel comorbidity scoring system has the potential for usage in future research work.

Three current and future research projects are utilizing and developing upon the work in this Thesis.

The Scottish Lung Cancer Forum has carried out a prospective audit of comorbidity, inflammation and management in small cell lung cancer. It is the hypothesis that there are differences in outcome and treatment delivered that occur between regions within the UK and also internationally. The hypothesis is that this may be due to variations in comorbidity, inflammation and demographics. This study has utilized both the data collection tool developed for this Thesis and the SCSS to analyse comorbidity.

This International study involves 9 Centres from across Scotland, England and also Berlin in Germany. Recruitment lasted from June 2009 until December 2012. So far we have at least 2 year follow up on approximately 700 patients. Data analysis is currently ongoing with preliminary results being anticipated by late 2015.

From the results of the research presented in this Thesis it is clear that a very detailed assessment of comorbidity such as the SCSS would not be practical or indeed useful in all newly diagnosed patients. However it appears to have significant merit particularly in those patients who are of borderline fitness for radical treatment such as surgery or radical radiotherapy. In such patients a very detailed assessment of comorbidity may identify patients who would be suitable for active steps to try to improve their fitness to allow such treatments. I am already beginning work with two groups specifically investigating this premise.

Professor Roma Maguire based at the University of Surrey School of Health Sciences has published extensively in Lung Cancer with her main interest being in attempting to link markers of the inflammatory response to patient outcomes and symptoms with the hypothesis that significant levels of inflammation may correspond to increasing burden of symptoms. We intend to analyse the relationship between symptoms, SCSS and mGPS. In those with high burdens of comorbidity and inflammation the intention is to develop focused supportive care interventions targeting the inflammatory response including pharmacological or non-pharmacological (e.g. exercise) along with interventions to reverse comorbidity whenever possible.

Dr Jo Bowden is a Consultant Palliative Care Physician at Edinburgh University. Her proposed MD will be to explore the relationship between comorbidity, muscle wasting in lung cancer and to assess how this impacts on physical function, treatment completion and survival. She is in the planning stage of an epidemiological study using a database of patients diagnosed with lung cancer pre-2010 in Scotland. The Edinburgh Research Team have developed software that allows the quantification of muscle mass from patients' staging/follow-up CT scans and are looking to combine this with information about patients' cancer characteristics and comorbid illnesses.

Her proposal will include body composition analysis from our patient database and to then combine this with our comorbidity, inflammation and survival data. The hypothesis is that these findings will help to identify patients who are the best (or better at least) candidates for cachexia management whether in anticipation of starting cancer-modifying treatment (a 'prehabilitation' approach), alongside treatment or even in the no treatment group.

As can be seen from the above collaborations there will be significant future work utilizing the work undertaken in this Thesis and it is hoped that both the improved understanding of the role of comorbidity and inflammation in lung cancer along with potential future interventions may improve the poor outcome currently seen in this disease.

# REFERENCES

Aarts MJ, Aerts JG, van den Borne BE, Biesma B, Lemmens VE, Kloover JS. Comorbidity in Patients with Small-Cell Lung Cancer. Trends and Prognostic Impact. Clin Lung Cancer. 2015;16(4):282-91

Abramovitch R, Marikovsky M, Meir G, Neeman M. Stimulation of tumour growth by wound-derived growth factors. Br J Cancer 1999;79:1392-1398.

Ahn DH, Mehta N, Yorio JT, Xie Y, Yan J, Gerber DE. Influence of Medical Comorbidities on the Presentation of Stage I-III NSCLC. Clin Lung Ca 2013:14(6):644-650.

Arriagada R, Le Chevalier T, Riviere A, Chomy P, Monnet I, Bardet E, Santos-Miranda JA, Le Péhoux C, Tarayre M, Benhamou S, Laplanche A. Patterns of failure after prophylactic cranial irradiation in small-cell lung cancer: analysis of 505 randomized patients. AnnOncol 2002;13(5):748-54.

Arrieta O, Rosa RMM, Villanueva-Rodríguez G, Serna-Thomé MG, Flores-Estrada D, Diaz-Romero C, Rodríguez CM, Martínez L, and Sánchez-Lara K. Association of nutritional status and serum albumin levels with development of toxicity in patients with advanced non-small cell lung cancer treated with paclitaxel-cisplatin chemotherapy: a prospective study. BMC Cancer 2010;10:50.

ASA Relative Value Guide 2002. American Society of Anesthesiologists, page xii, Code 99140

Asmis TR, Ding K, Seymour L, Shepherd FA, Leighl NB, Winton TL, Whitehead M, Spaans JN, Graham BC, Goss GD; National Cancer Institute of Canada Clinical Trials Group. Age and Comorbidity as Independent Prognostic Factors in the Treatment of Non-Small-Cell Lung Cancer: A Review of National Cancer Institute of Canada Clinical Trials Group Trials. JCO 2008; 26: 54-59

Auperin A, Arriagada R, Pignon JP, Le Pechoux C, Gregor A, Stephens RJ, Kristjansen PE, Johnson BE, Ueoka H, Wagner H, Aisner J. Prophylactic cranial irradiation for patients with small-cell lung cancer in complete remission. Prophylactic Cranial Irradiation Overview Collaborative Group. N Engl J Med 1999;341(7):476-84.

Ball D, Thursfield V, Irving L, Mitchell P, Richardson G, Torn-Broers Y, Wright G, Giles G. Evaluation of the Simplified Comorbidity Score (Colinet) as a prognostic indicator for patients with lung cancer: a cancer registry study. Lung Cancer. 2013 Nov;82(2):358-61

Bennett CL, Greenfield S, Arnow H, Ganz P, Vogelzang NJ, Elaskoff RM. Patterns of care related to age of men with prostate cancer. Cancer 1991; 67: 2633 - 2641

Berglund A, Lambe M, Luchtenborg M, Linklater K, Peake MD, Holmberg L, Moeller H. Social differences in lung cancer management and survival in South East England: a cohort study. BMJ Open 2012;2:e001048.

Berrino F, De Angelis R, Sant M, Rosso S, Bielska-Lasota M, Coebergh JW, Santaquilani M and the EUROCARE Working group: Survival for eight major cancers and all cancers combined for European adults diagnosed in 1995-99: results of the EUROCARE-4 study. Lancet Oncol 2007;8:773-783.

Birim O, Pieter Kappetein A, Bogers AJCC. Charlson comorbidity index as a predictor of long term outcome after surgery for non-small-cell lung cancer. Eur Jrnl Cardio-Thoracic S 2005; 28: 759-762.

Birring SS, Peake MD. Symptoms and the early diagnosis of lung cancer. Thorax 2005;60:268-269. Blum TG, Rich A, Baldwin D, Beckett P, De Ruysscher D, Faivre-Finn C, Gaga M, Gamarra F, Grigoriu B, Hansen NC, Hubbard R, Huber RM, Jakobsen E, Jovanovic D, Konsoulova A, Kollmeier J, Massard G, McPhelim J, Meert AP, Milroy R, Paesmans M, Peake M, Putora PM, Scherpereel A, Schönfeld N, Sitter H, Skaug K, Spiro S, Strand TE, Taright S, Thomas M, van Schil PE, Vansteenkiste JF, Wiewrodt R, Sculier JP. The European initiative for quality management in lung cancer care.Eur Respir J. 2014 May;43(5):1254-77

Bozzetti F, Santarpia L, Pironi L, Thul P, Klek S, Gavazzi C, Tinivella M, Joly F, Jonkers C, Baxter J, Gramlich L, Chicharro L, Staun M, Van Gossum A, Lo Vullo S, Mariani L. The prognosis of incurable cachectic cancer patients on home parenteral nutrition: a multi-centre observational study with prospective follow up of 414 patients. Ann Oncol 2014; 25(2):487-93.

British Heart Foundation Statistics Database 2009.

Brown DJF, Milroy R, Preston T, McMillan DC. The relationship between an inflammation-based prognostic score (Glasgow Prognostic Score) and changes in serum biochemical variables in patients with advanced lung and gastrointestinal cancer. J ClinPathol 2007;60:705-708.

Brown T, Darnton A, Fortunato L, Rushton L. Occupational cancer in Britain: Respiratory cancer sites: larynx, lung and mesothelioma. Br J Cancer 2012;107(S1):S56-S70.

Calvo-Espinos C, De Ganona-Lana ER, Gonzalez-Anguren C, Lama-Gay M. Assesment of the impact of comorbidity on the survival of cancer patients treated by palliative care teams. Pall Supp Care 2015;13(4):1049-55.

Cancer Research UK and National Cancer Intelligence Network. Cancer by deprivation in England: Incidence, 1996-2010, Mortality, 1997-2011. London: NCIN; 2014.

Canna K, Hilmy M, McMillan DC, Smith GW, McKee RF, McArdle CS, McNicol A. The relationship between tumour proliferative activity, the systemic inflammatory response and survival in patients undergoing curative resection for colorectal cancer. Colorectal Dis 2008;10:663-667.

Cartman ML, Hatfield AC, Muers MF Peake MD, Haward RA, Forman D; Yorkshire Cancer Management Study Group, Northern and Yorkshire Cancer Registry and Information Service (NYCRIS), UK. Lung cancer: district active treatment rates affect survival. J Epidemiol Community Health 2002; 56:424-9.

Charlson ME, Pompei P. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chron Dis 1987; 40: 373-383.

Chen H, Overcash J, Extermann M, Cantor A, Balducci L. Comprehensive geriatric assessment in elderly cancer patients: impact on overall survival and progression-free survival. Am Geriatric Soc Meeting 1999; Poster P76

Clemens J, Feinstein A, Holabird N, Cartwright S. A new clinical-anatomic staging system for evaluating prognosis and treatment of prostatic cancer. J Chron Dis 1986; 39: 913-928

Coleman MP, Rachet B, Woods LM, Mitry E, Riga M, Cooper N, Quinn MJ, Brenner H, Estève J. Trends and socioeconomic inequalities in cancer survival in England and Wales up to 2001. Br J Cancer 2004;90(7):1367-73.

Colinet B, Jacot W, Bertrand D, Lacombe, M-C Bozonnat, J-P Daurès, and J-L Pujol. A new simplified co-morbidity score as a prognostic factor in non-small-cell lung cancer: description and comparison with the Charlson's index. Br Jrnl Ca 2005; 93: 1098-1105.

Coussens LM, Werb Z. Inflammation and cancer. Nature 2002;420: 860 - 867.

Crumley ABC, Stuart RC, McKernan M McDonald, A.C., and McMillan, D.C. Comparison of an inflammation-based prognostic score (GPS) with performance status (ECOG-ps) in patients receiving palliative chemotherapy for gastroesophageal cancer. Jrnl Gastro and Hep 2008; 23: 325-329.

Damhuis RA, Schutte PR. Resection rates and postoperative mortality in 7,899 patients with lung cancer. EurRespir J 1996; 9:7-10.

Deleuran T, Thomsen RW, Norgaard M Jacobsen, JB, Rasmussen, TR, Sogaard, T. Comorbidity and survival of Danish lung cancer patients from 2000-2011: a population-based cohort study. Clinical Epidemiology 2013:5(1):31-38.

De Leyn P, Lardinois D, Van Schil PE, Rami-Porta R, Passlick B, Zielinski M, Waller DA, Lerut T, Weder W. ESTS guidelines for preoperative lymph node staging for non-small cell lung cancer. Eur J Cardiothorac Surg2007;32(1):1-8.

De Ruysscher D, Vansteenkiste J. Chest radiotherapy in limited stage small cell lung cancer: facts, questions, prospects. RadiotherOncol 2000;55(1):1-9.

Detterbeck FC, Boffa DJ, Tanoue LT. The new lung cancer staging system. Chest. 2009;136(1):260-271.

Detterbeck FC, Rivera MP. Diagnosis and treatment of Lung Cancer: an evidence based guide for the practicing clinician. Philadelphia: WB. Saunders; 2001.

Doll R, Peto R, Boreham J, Sutherland I. Mortality from cancer in relation to smoking: 50 years observations on British doctors. Br J Cancer 2005;92(3):426-29.

Du Clos TW, Mold C. C-Reactive protein: an activator of innate immunity and a modulator of adaptive immunity. Immunol Res 2004; 30:261-277.

Earle CC, Venditti LN, Neumann PJ, Gelber RD, Weinstein MC, Potosky AL, Weeks JC. Who Gets Chemotherapy for Metastatic Lung Cancer? Chest 2000; 117:1239-1246

Edge SB, Bird DR, Compton CC. AJCC Cancer staging Manual. 7<sup>th</sup> ed. 2009.

Emslie C, Mitchell R. Are there gender differences in the geography of alcohol-related mortality in Scotland? An ecological study.BMC Public Health. 2009 Feb 16;9:58

Erridge SC, Murray B, Williams L, Brewster D, Black R, Price A, Murray N, Sheehan F. Improved survival from lung cancer in British Columbia compared to Scotland-are different treatment rates the whole story? Lung Cancer 2009;64:358-366.

Extermann M. Measuring co-morbidity in older cancer patients. EurJrnlCa 2000; 36: 453-471.

Ferlay, P. Autier, M. Boniol, Heanue M, Colombet M, Boyle P. Estimates of the cancer incidence and mortality in Europe 2006.Ann Oncol. 2007; Mar;18(3):581-92.

Firat S, Roger M, Byhardt W, Gore E. Comorbidity and Performance Score are Independent Factors in Stage III Non-Small-Cell Lung Cancer. Int J. Radiat Oncol Biol Phys 2002; 54:357-364 Forrest LM, McMillan DC, McArdle CS, Angerson WJ, Dunlop DJ. Evaluation of cumulative prognostic scores based on the systemic inflammatory response in patients with inoperable non-small-cell lung cancer. Br J Cancer 2003;89:1028 -1030

Forrest LM, McMillan DC, McArdle CS, Angerson WJ, Dunlop DJ. Comparison of an inflammation-based prognostic score (GPS) with performance status (ECOG) in patients receiving platinum-based chemotherapy for inoperable non-small-cell lung cancer. Br J Cancer 2004;90:1704-1706.

Forrest LM, McMillan DC, McArdle CS, Angerson WJ, Dagg K and Scott HR. A prospective longitudinal study of performance status, an inflammationbased score (GPS) and survival in patients with inoperable non-small-cell lung cancer. Br J Cancer 2005;92:1834-1836.

Forrest LF, Sowden S, Rubin G, White M, Adams J. Socio-economic inequalities in patient, primary care, referral, diagnostic, and treatment intervals on the lung cancer care pathway: protocol for a systematic review and meta-analysis.Syst Rev. 2014 Mar 25;3:30

Fry WA, Phillips JL, Menck HR. Ten-year survey of lung cancer treatment and survival in hospitals in the United States. Cancer 1999;86: 1867-76.

Gagnon B, Abrahamowicz M, Xiao Y, Beauchamp ME, MacDonald N, Kasymjanova G, Kreisman H and Small Dhttp:// www.nature. com/bjc/ journal/v102 /n7/full /6605603a.html-aff5#aff5. Flexible modeling improves assessment of prognostic value of C-reactive protein in advanced non- small cell lung cancer. Br J Cancer 2010;102:1113-1122.

Ganti AK, Siedlik E, Marr AS Loberiza F.R Jr., Kessinger A. Predictive ability of Charlson comorbidity index on outcomes from lung cancer. Am J ClinOncol 2011;34(6):593-6.

Girling DJ. Comparison of oral etoposide and standard intravenous multidrug chemotherapy for small-cell lung cancer: a stopped multicentre randomised trial. Medical Research Council Lung Cancer Working Party. Lancet 1996; 348 (9027) :563-6.

Girones R, Torregrosa D, Gomez-Codina J Maestu I, Tenias JM, Rosell R. Prognostic impact of comorbidity in elderly lung cancer patients: Use and comparison of two scores. Lung Ca 2011:72;108-113.

Glare PA, Eychmueller S, and McMahon P. Diagnostic accuracy of the palliative prognostic score in hospitalized patients with advanced cancer. J ClinOncol 22: 4823-8, 2004

Goffin J, Lacchetti C, Ellis PM, Ung YC, Evans WK. First-line systemic chemotherapy in the treatment of advanced non-small cell lung cancer: A systematic review. J Thorac Oncol 2010;5(2):260-74.

Greenfield S, Aplone G. Assessment of Comorbidity: the Index of Coexistent Disease (ICED). New England Medical Centre, Boston, MA. 1991

Gregor A, Thomson CS, Brewster DH, Stoner PL, Davidson J, Fergusson RJ, Milroy R on behalf of the Scottish Cancer Trials Lung Group and the Scottish Cancer Therapy Network. Management and survival of patients with lung cancer in Scotland diagnosed in 1995: results of a national population based study. Thorax 2001; 56: 212-217

Gripp S, Moeller S, Bolke E, Schmitt G, Matuschek C, Asgari S, Asgharzadeh F, Roth S, Budach W, Franz M, Willers R. Survival prediction in terminally ill cancer patients by clinical estimates, laboratory tests, and self-rated anxiety and depression. J ClinOncol 25:3313-20, 2007

Grose D, Devereux G, Brown L, Jones R, Sharma D, Selby C, Morrison DS, Docherty K, McIntosh D, Louden G, Downer P, Nicolson M, Milroy R;

Scottish Lung Cancer Forum. Variation in comorbidity and clinical management in patients newly diagnosed with lung cancer in four Scottish centers. J ThoracOncol. 2011;6:500-9.

Grose D, Devereux G, Milroy R. Comorbidity in Lung Cancer. Important but Neglected. A review of the current literature. Clinical Lung Cancer 2011;12:207-211.

Grose D, Morrison DS, Devereux G, Jones R, Sharma D, Selby C, Docherty K, McIntosh D, Louden G, Nicolson M, McMillan DC, Milroy R; Scottish Lung Cancer Forum. Comorbidities in lung cancer: prevalence, severity and links with socioeconomic status and treatment. Postgrad Med J. 2014 Jun;90(1064):305-10.

Grose D, Morrison DS, Devereux G, Jones R, Sharma D, Selby C, Docherty K, McIntosh D, Louden G, Nicolson M, McMillan DC, Milroy R. Simple and Objective Prediction of Survival in Patients with Lung Cancer: Staging the Host Systemic Inflammatory Response. Lung Ca International. Volume 2014 (2014), Article ID 731925,

Grose D, Morrison DS, Devereux G, Jones R, Sharma D, Selby C, Docherty K, McIntosh D, Nicolson M, McMillan DC, Milroy R; Scottish Lung Cancer Forum. The impact of comorbidity upon determinants of outcome in patients with lung cancer. Lung Cancer. 2015 Feb;87(2):186-92.

Guadagnoli E, Shapiro C, Gurwitz JH. Age-related patterns of care: evidence against ageism in the treatment of early stage breast cancer. JCO 1997; 15: 2338 - 2344

Haynes SR, Lawler PG. An assessment of the consistency of ASA physical status classification allocation. Anaesthesia 1995; 50 (3): 195-9.

Havlik RJ, Yancik R, Long S, Ries L, Edwards B. The National Institute on Aging and the National Cancer Institute SEER collaborative study on comorbidity and early diagnosis of cancer in the elderly. Cancer. 1994 Oct 1;74(7 Suppl):2101-6.

Horner MJ, Ries LAG, Krapcho M, Neyman N, Aminou R, Howlader N. SEER Cancer Statistics Review, 1975-2006, National Cancer Institute. Bethesda, MD, http://seer.cancer.gov/csr/1975\_2006/, based on November 2008 SEER data submission, posted to the SEER web site, 2009

http://www.ccs.ca/download/position\_statements/Grading%20of%20Angin a.pdf

Hu Y, Hong Y, Xu Y, Liu P, Guo DH, Chen Y. Inhibition of the JAK/STAT pathway with ruxolitinib overcome cisplatin resisitance in NSCLC. Apoptosis 2014; 19(11):1627-36.

Iachina M, Green A, Jakobsen E. The direct and indirect impact of comorbidity on the survival of patients with non-small cell lung cancer: a combination of survival, staging and resection models with missing measurements in covariates. BMJ Open 2014;4

Ibrahim EM, Kazkaz GA, Abouelkhair KM, Al-Mansour MM, Al-Fayea TM, Al-Foheidi M, Bayer AM, Elmasri OA. Increased risk of second lung cancer in Hodgkin's lymphoma survivors: a meta-analysis. Lung 2013; 191(1):117-34).

Imperatori A, Harrison RN, Leitch DNF Rovera, G Lepore, G Dionigi, P Sutton, L Dominioni. Lung cancer in Teesside (UK) and Varese (Italy): a comparison of management and survival. Thorax 2006; 61: 232-239.

Inoue K, Sato M, Fujimura S, Sakurada A, Takahashi S, Usuda K, Kondo T, Tanita T, Handa M, Saito Y, Sagawa M. Prognostic assessment of 1310 patients with non-small-cell lung cancer who underwent complete resection from 1980 to 1993. JThorac CardiovascSurg 1998; 116(3):407-11.

Information Services Division of the NHS in Scotland. Cancer in Scotland (April 2009). ISD, Edinburgh, 2009.

ISD Scotland May 2013: http://www.isdscotland.org/Health-Topics /Cancer /Publications/index.asp.

Islam KM, Jiang X, Anggondowati T, Lin G, Ganti AK. Comorbidity and Survival in Lung Cancer Patients. Cancer Epidemiol Bio Prev 2015; 24(7):1079-85.

Jack R, Guilford M, Ferguson J, Møller H. Geographical inequalities in lung cancer management and survival in South East England: evidence of variation in access to oncology services?. Br J Cancer 2003; 88(7):1025-31.

Janssen-Heijnen MLG, Gatta G, Forman D. Variation in survival of patients with lung cancer in Europe. Eur J Cancer 1998; 34: 2191-6

Janssen-Heijnen MLG, Smulders S, Lemmens VEPP, Smeenk F, van Geffen HJAA, Coebergh A. Effect of comorbidity on the treatment and prognosis of elderly patients with non-small cell lung cancer. Thorax 2004; 59: 602-607

Kaplan M, Feinstein A. The importance of classifying initial comorbidity in evaluating the outcome of diabetes mellitus. J Chron Dis 1974; 27:387-404.

Kates M, Perez X, Gribetz J, Swanson SJ, McGinn T, Wisnivevsky JP. Validation of a Model to Predict Perioperative Mortality from Lung Cancer Resection in the Elderly.Am.J.Respir. Crit. Care Med. 2009; 179: 390-395 Koch A, Fohlin H, So renson S. Prognostic significance of c-reactive protein and smoking in patients with advanced non-small cell lung cancer treated with first-line palliative chemotherapy. J ThoracOncol 2009;4:326.

Koshy M, Malik R, Spiotto M, Mahmood U, Weichselbaum R, Sher D. Disparities in treatment of patients with inoperable stage I non-small cell lung cancer: a population-based analysis. J ThoracOncol. 2015 Feb;10(2):264-71

Lagerwaard FJ, Verstegen NE, Haasbeek CJ, Slotman BJ, Paul MA, Smit EF, Senan S. Outcomes of stereotactic ablative radiotherapy in patients with potentially operable stage I non-small cell lung cancer.Int J RadiatOncolBiol Phys. 2012 May 1;83(1):348-53

Laird BJ, Kaasa S, McMillan DC, Fallon MT, Hjermstad MJ, Fayers P, Klepstad P. Prognostic factors in patients with advanced cancer: a comparison of clinicopathological factors and the development of an inflammation-based prognostic system. Clin Cancer Res. 2013;19:5456-64

Laroche C, Wells F, Coulden R, Stewart S, Goddard M, Lowry E, Price A, Gilligan D. Improving surgical resection rate in lung cancer. Thorax 1998; 53:445-9.

Laroche C, Fairbairn I, Moss H, Pepke-Zaba J, Sharples L, Flower C, Coulden R. Role of computed tomographic scanning of the thorax prior to bronchoscopy in the investigation of suspected lung cancer. Thorax 2000;55(5):359-63.

Leung EYL, Scott HR, McMillan DC. Clinical Utility of the Pretreatment Glasgow Prognostic Score in Patients with Advanced Inoperable Non-small Cell Lung Cancer. JTO 2012; 7(4):655-662. Linn B, Linn W Gurel L. Cumulative Illness Rating Scale. J Am Geriatr Soc 1968; 16:622-626.

Little, Brown & Co, The Criteria Committee of the New York Heart Association. *Nomenclature and Criteria for Diagnosis of Diseases of the Heart and Great Vessels*. 9th ed. Boston, Mass; 1994:253-256.

Lorigan P, Califano R, Faivre-Finn C, Howell A, Thatcher N. Lung cancer after treatment for breast cancer. Lancet Oncol 2010;11(12):1184-92.

Luchtenborg M, Jakobsen E, Krasnik M, Linklater KM, Mellemgaard A, Møller H. The effect of comorbidity on stage-specific survival in resected NSCLC patients. EJC 2012:48:3386-3395.

Ludbrook JJ, Truong PT, MacNeil MV. Do age and comorbidity impact treatment allocation and outcomes in limited stage small-cell lung cancer? A community-based population. Int Jrnl Radiat Oncol Biol Phys. 2003; 55: 1166-1168.

Lyden D, Lu M, Levine S, Brott TG, Broderick J. Modified National Institutes of Health Stroke Scale for Use in Stroke Clinical Trials. Stroke. 2001;32:1310.

MacDonald N. Terminology in cancer cachexia: importance and status. Current Opinion in Clinical Nutrition & Metabolic Care 2012; 15(3): 220 -225.

Maddams J, Parkin DM, Darby SC. The cancer burden in the United Kingdom in 2007 due to radiotherapy. Int J Cancer 2011;129(12):2885-93.

Makitaro R, Paakko P, Huhti E, Bloigu R, Kinnula VL. Prospective population-based study on the survival of patients with lung cancer. EurRespir J 2002; 19(6):1087-92.

Maltoni M, Caraceni A, Brunelli C, Broeckaert B, Christakis N, Eychmueller S, Glare P, Nabal M, Viganò A, Larkin P, De Conno F, Hanks G, Kaasa S; Steering Committee of the European Association for Palliative Care. Prognostic factors in advanced cancer patients: evidence-based clinical recommendations--a study by the Steering Committee of the European Association for Palliative Care. J ClinOncol 23:6240-8, 2005

Managed Clinical Network Scotland. Unpublished Data (Personal Communication) Management of exacerbations of COPD. Thorax 2004; 59: 131-i156

Martin L, Watanabe S, Fainsinger R, Lau F, Ghosh S, Quan H, Atkins M, Fassbender K, Downing GM, Baracos V. Prognostic factors in patients with advanced cancer: use of the patient-generated subjective global assessment in survival prediction. J ClinOncol 28:4376-83, 2010

McMillan DC. Systemic inflammation, nutritional status and survival inpatients with cancer. Curr Opin Clin Nutr Metab Care 2009;12:223-226.

McMillan DC. The systemic inflammation-based Glasgow Prognostic Score: A decade of experience in patients with cancer. Cancer Treat Rev. 2013 Aug;39(5):534-40

McMillan DC, Crozier JE, Canna K, Angerson WJ, McArdle CS. Evaluation of an inflammation based prognostic score (GPS) in patients undergoing resection for colon and rectal cancer. Int J Colorectal Dis 2007;22:881-886.

Mellemgaard A, Luchtenborg M, Iachina M, Jakobsen E, Green A, Krasnik M, Moller H. Role of comorbidity on survival after radiotherapy and chemotherapy for nonsurgically treated lung cancer. JTO 2015; 10(2):272-9.

Milroy R, Grose D, McAlpine L, Fergusson R, Brown P. Impact of Comorbidity on Outcome in Lung Cancer. Lung Cancer 2005; 49:204.

Milroy R, Grose D, Louden G, Downer P, Jones R. Assessment of comorbidity in newly diagnosed lung cancer patients. Eur Resp Jrnl 2009; 34: Suppl 53: 479S

Mitchell M, Christie J, Silvestri G. Racial Differences Pertaining to a Belief about Lung Cancer Surgery: Results of a Multicentre Survey. Ann Intern Med 2003; 139:558-563

Mitchell R, Fowkes G, Blane D, Bartley M. High rates of ischaemic heart disease in Scotland are not explained by conventional risk factors. J Epidemiol Community Health 2005;**59**:565-567

Mor V, Laliberte L, Morris JN, Wiemann M. The Karnofsky Performance Status Scale. An examination of its reliability and validity in a research setting. Cancer 1984; 53:2002-7.

Morris J: The Clinical Dementia Rating (CDR): Current version and scoring rules. Neurology; 43: 2412-2414

Mosenifar Z. Population issues in clincial trials. PATS 2007; 4: 185-188

Mountain CF. A new International staging system for lung cancer. Chest 1986; 89 (suppl 4):225S-33S

National Collaborating Centre for Clinical Conditions. Management of exacerbations of COPD. Thorax 2004;59:i131-i156.

NICE Guideline CG 121. 2011. The Diagnosis and Treatment of Lung Cancer

Nozoe T, Matsumata T, Sugimachi K. Preoperative elevation of serum creactive protein is related to impaired immunity in patients with colorectal cancer. Am J ClinOncol 2000;23:263.

NSCLC Meta-analyses Collaborative Group. Adjuvant chemotherapy, with or without postoperative radiotherapy, in operable non-small cell lung cancer: two meta-analyses of individual patient data. Lancet 2010;375(9722):1267-77.

Office for national Statistics (2006) Mortality statistics cause. Review of the registrar General on deaths by cause, sex and age, in England and Wales 2005. London: Office for national Statistics.

Office of the Chief Statistician, Scottish Executive. Scottish Index of Multiple Deprivation 2006: technical report

Office for National Statistics, July 2013.:http:// www.ons.gov.uk/ons/rel/ vsob1/cancer-statistics-registrations--england--series -mb1-/index.html.

Owen JB, Khalid N, Ho A, Kachnic LA, Komaki R, Tao ML, Currey A, Wilson JF. Can patient comorbidities be included in clinical performance measures for radiation oncology?. J Oncol Pract. 2014 May;10(3):e175-81

Palma D, Visser O, Lagerwaard FJ, Belderbos J, Slotman BJ, Senan S. Impact of introducing stereotactic lung radiotherapy for elderly patients with stage I non-small-cell lung cancer: a population-based time-trend analysis.J ClinOncol. 2010 Dec 10;28(35):5153-9

Parkin DM. 2.Tobacco-attributable cancer burden in the UK in 2010. Br J Cancer 2011;105(S2):S6-S13.

Parkin DM, Darby SC. 12.Cancers in 2010 attributable to ionising radiation exposure in the UK. Br J Cancer 2011;105(S2):S57-S65.

Parkin DM, Whelan SL, Ferlay J. Cancer Incidence in Five Continents Volume VIII.IARC Scientific Publications.Vol. 155. Lyon, France: International Agency for Research on Cancer, 2002.

Piccirillo J. Inclusion of comorbidity in a staging system for head and neck cancer. Oncology 1995; 9:831-836

Piccirillo J, Tierney R, Coastas I, Grove L, Spitznagel EL. Prognostic importance of comorbidity in a hospital-based cancer registry. JAMA 2004; 291: 2441-2447

Pieterman RM, van Putten JW, Meuzelaar JJ, Mooyaart EL, VaalburgW, Koëter GH, et al. Preoperative staging of non-small-cell lung cancer with positron-emission tomography. N Engl J Med2000;343(4):254-61.

Potosky AL, Saxman S, Wallace RB, Lynch CF. Population variations in the Initial Treatment of Non-Small Cell Lung Cancer. JCO 2004; 22: 3261-3268

Proctor MJ, Morrison DS, Talwar D, Balmer SM, O'Reilly DSJ, Foulis AK, Horgan PJ, McMillan DC. An inflammation-based prognostic score (mgps) predicts cancer survival independent of tumour site: a Glasgow Inflammation Outcome Study. Br J Cancer 2011;104:726-734.

Radzikowska E, Glaz P and Roszkowski K. Lung cancer in women: age, smoking, histology, performance status, stage, initial treatment and survival. Population-based study of 20 561 cases. Anns Oncology 2002; 13:1087 - 1093

Read WL, Tierney RM, Page NC, Costas I, Govindan R, Spitznagel EL, Piccirillo JF. Differential Prognostic Impact of Comorbidity. J ClinOncol 2004; 15:3099-3103 Rich A, Tata LJ, Stanley RA, Free CM, Peake MD, Baldwin DR, Hubbard RB. Lung cancer in England: Information from the National Lung Cancer Audit (LUCADA). Lung Cancer 2011; 72:16-22.

Riechelmann H, Neagos A, Netzer-Yilmaz U, Gronau S, Scheithauer M, Rockemann MG. The ASA-score as a comorbidity index in patients with cancer of the oral cavity and oropharynx. Laryngorhinootologie 2006; 85(2):99-104.

Rosell R, Carcereny E, Gervais R, Vergnenegre A, Massuti B, Felip E, Palmero R, Garcia-Gomez R, Pallares C, Sanchez JM, Porta R, Cobo M, Garrido P, Longo F, Moran T, Insa A, De Marinis F, Corre R, Bover I, Illiano A, Dansin E, de Castro J, Milella M, Reguart N, Altavilla G, Jimenez U, Provencio M, Moreno MA, Terrasa J, Muñoz-Langa J, Valdivia J, Isla D, Domine M, Molinier O, Mazieres J, Baize N, Garcia-Campelo R, Robinet G, Rodriguez-Abreu D, Lopez-Vivanco G, Gebbia V, Ferrera-Delgado L, Bombaron P, Bernabe R, Bearz A, Artal A, Cortesi E, Rolfo C, Sanchez-Ronco M, Drozdowskyj A, Queralt C, de Aguirre I, Ramirez JL, Sanchez JJ, Molina MA, Taron M, Paz-Ares L; Spanish Lung Cancer Group in collaboration with Groupe Français de Pneumo-Cancérologie and Associazione Italiana Oncologia Toracica. Erlotinib versus standard chemotherapy as first-line treatment for European patients with advanced mutation-positive non-small-cell lung cancer EGFR (EURTAC): A multicentre, open-label, randomised phase 3 trial. Lancet Oncol 2012;13(3):239-46.

Rowell N, Williams CJ. Radical radiotherapy for stage I/II non-small cell lung cancer in patients not sufficiently fit for or declining surgery (medically inoperable). Cochrane Database of Systematic Reviews 2004, Issue 2.

Scagliotti GV, Parikh P, von Pawel J, Biesma B, VansteenkisteJ, Manegold C, Serwatowski P, Gatzemeier U, Digumarti R, Zukin M, Lee JS,

Mellemgaard A, Park K, Patil S, Rolski J, Goksel T, de Marinis F, Simms L, Sugarman KP, Gandara D. Phase III study comparing cisplatin plus gemcitabine with cisplatin plus pemetrexed in chemotherapy naïve patients with advanced-stage non-small-cell lung cancer. J ClinOncol 2008;26(21):3541-51.

Schreiber G, McCrory DC. Performance characteristics of different modalities for diagnosis of suspected lung cancer: summary of published evidence. Chest 2003; 123(1 Suppl):115S-28S.

Scottish Intercollegiate Guidelines Network (SIGN). Management of patients with lung cancer 2005.http://www.sign.ac.uk/pdf/sign80.pdf

SIGN 137. The Management of Lung Cancer.http://www.sign.ac.uk/pdf/ SIGN 137.pdf

Simon GR, Wagner H. Small cell lung cancer. Chest 2003;123(1Suppl):259S-71S.

Solomon BJ, Mok T, Kim DW, Wu YL, Nakagawa K, Mekhail T, Felip E, Cappuzzo F, Paolini J, Usari T, Lyer S, Reisman A, Wilner KD, Tursi J, Blackhall F. First Line Crizotinib vs Chemotherapy in ALK-Positive Lung Cancer. NEJM 2014; 371:2167-2177.

Spiro SG, Gould MK, Colice GL. Initial evaluation of the patient with lung cancer: symptoms, signs, laboratory tests, and paraneoplastic syndromes: ACCP evidenced-based clinical practice guidelines (2nd edition). Chest. Sep 2007; 132 (3 Suppl):149S-160S.

Suzuki K, Nagai K, Yoshida J, Moriyama E, Nishimura M, Takahashi K, Nishiwaki Y. Prognostic factors in clinical stage I non-small cell lung cancer. Ann ThoracSurg 1999;67(4):927-32. Tammemagi MC, Neslund-Dudas CMA, Simoff M, Kvale P. Smoking and Lung Cancer Survival: The role of Co-morbidity and Treatment. Chest 2004; 125: 27-37.

Temel JS, Greer JA, Muzikansky A, Gallagher ER, Admane S, Jackson VA, Dahlin CM, Blinderman CD, Jacobsen J, Pirl WF, Billings JA, Lynch TJ. Early Palliaitve Care for Patients with Metastatic Non-Small-Cell Lung Cancer. N Engl J Med 2010 ;363:733-42.

The Criteria Committee of the New York Heart Association. Nomenclature and Criteria for Diagnosis of Diseases of the Heart and Great Vessels. 9th ed. Boston, Mass: Little, Brown & Co; 1994:253-256 The Information Centre for Health and Social Care. National Clinical Audit Support Programme Lung Cancer NLCA Report 2005. Leeds: Information Centre for Health and Social Care; 2006.

Wang S, Wong M, Hamilton N, Davoren J, Jahan TM, Walter LC. Impact of Age and Comorbidity on Non-Small Cell Lung Cancer Treatments in Older Veterans. JCO 2012; 30: 1447-1455.

West of Scotland Cancer Network, 2011 Audit report.www.woscan.scot.nhs.uk

WHO/ECOG Performance status.www.ecog.org/general/perf\_stat.html

Wilop S, Crysandt M, Bendel M, Mahnken AH, Osieka R, Jost, E. Correlation of C-reactive protein with survival and radiographic response to first-line platinum-based chemotherapy in advanced non-small cell lung cancer. Onkologie 2008; 31:665- 670.

Woolhouse I. Variation in lung cancer outcomes in the UK and Europe. Clin Med April 1, 2011 vol. 11 no. 2110-111. Yankelevitz DF, Gupta R, Zhao B, Henschke CI. Small pulmonary nodules: evaluation with repeat CT preliminary experience. Radiology 1999;212(2):561-6.

Yates JW, Chalmer B, McKegney FP. Evaluation of patients with advanced cancer using the Karnofsky performance status. Cancer 1980; 45:2220-4.

## APPENDIX 1: DATA COLLECTION PROFORMA

Patient Details

M / F	Age	Postcode (1 <sup>st</sup> part) Weight				t		
Home circur	Live alone/Independant Requiring support from Partner/Family							
Current Perf	formance Stat	us	0	1	2	3	4	(circle)
Where PS>0, what is deciding reason								
1	Effect of lung cancer (acute poor PS)							
2	General frailness / chronic comorbidity (chronic poor PS)							
3	Combination (acute on chronic poor PS)							
Relative estimate of PS 6 months prior to diagnosis								
Creatinine		Album	in		CRP		_(if me	easured)
FEV1	_% Predicted_			MRC D	yspnoe	ea score	e- Grad	le

Significant comorbidity (ie affecting PS or requiring treatment)

COPD	0	1	2	3
IHD	0	1	2	3
Heart failure	0	1	2	3
CVA	0	1	2	3
Dementia	0	1	2	3
Diabetes Mellitus	0	1	2	3
Renal Failure	0	1	2	3
Other malignancy	0	1	2	3
Weight loss	0	1	2	3
PVD	0	1	2	3
Alcohol Significant other	0 0	1 1	2 2	3 3

Smoking \_\_\_\_\_ pack years

Staging CT performed	Yes	No			
MDT					
Patient not discussed at MDT	yes	no			
Patient discussed at MDT	yes	no			
Patient discussed at MDT then seen at specialist clinic (within 2wo	seen eeks)	not seen			
Please record staff present at MDT					
Respiratory Physician	yes	no			
Clinical Oncologist	yes	no			
Radiologist	yes	no			
Palliative care Physician	ves	no			
Pathologist	yes	no			
Specialist nurse	yes	no			

#### Non-small cell lung cancer Stage I

Local symptoms yes no

Proposed for surgical resection

If not, reason?

Age Poor PS Comorbidity Patient refusal

Alternative treatment recommended

Radical radiotherapy

Chemotherapy + Radical radiotherapy

Chemotherapy

High dose palliative radiotherapy

Palliative radiotherapy

Specialist palliative care referral

Non-small cell lung cancer Stage II

Local symptoms yes no

Proposed for surgical resection

If not, reason?

Age Poor PS Comorbidity Patient refusal

Alternative treatment recommended

Radical radiotherapy

Chemotherapy + Radical radiotherapy

Chemotherapy

High dose palliative radiotherapy

Palliative radiotherapy

Specialist palliative care referral

Non-small cell lung cancer

Stage IIIa

no

Local symptoms yes

Proposed for Radical treatment

Surgery Radical radiotherapy Chemotherapy + Radical radiotherapy Induction chemotherapy

If not, reason?

Age Poor PS Comorbidity Disease volume too great Patient refusal

Alternative treatment recommended

High dose palliative radiotherapy

Palliative radiotherapy

Palliative chemotherapy

Specialist palliative care referral

Non-small cell lung cancer

Stage IIIb

no

Local symptoms yes

Proposed for radical treatment

Radical radiotherapy

Chemotherapy + Radical radiotherapy

Induction chemotherapy

If not, reason?

Age Poor PS Comorbidity Disease volume too great Patient refusal

Alternative treatment recommended

High dose palliative radiotherapy

Palliative radiotherapy

Palliative chemotherapy

Specialist palliative care referral

### Non-small cell lung cancer

Stage IV

Local symptoms yes

no

Proposed for chemotherapy

If not, reason?

Age Poor PS Comorbidity Patient refusal

Alternative treatment recommended

High dose palliative radiotherapy

Palliative radiotherapy

**Biological treatment** 

Specialist palliative care referral

Small cell lung cancer Limited stage

Local symptoms yes no

Proposed for concurrent chemoradiotherapy

Proposed for initial combination chemotherapy

If not, reason?

Age Poor PS Comorbidity Patient refusal

Alternative treatment recommended

Single agent chemotherapy

Palliative radiotherapy

Specialist palliative care referral

Small cell lung cancer

Extensive disease

Local symptoms yes

no

Proposed for combination chemotherapy

If not, reason?

Age Poor PS Comorbidity Patient refusal

Alternative treatment recommended

Single agent chemotherapy

Palliative radiotherapy

Specialist palliative care referral

Clinical/radiographic diagnosis (no histology)

Investigations

Biopsy procedure attempted?	Yes	No
If yes - procedure(s) attempted	Bronchoscopy Percutaneous lung biopsy Transbronchial lung biopsy Mediastinoscopy VATS	r

Neck node FNA/Biopsy

If no, reason?

Age

Poor PS

Comorbidity

Inaccessible

Patient refusal

Clinical/radiological stage

	Т					
	Ν					
	Μ					
	Stage I	lla	llb	Illa	IIIb	IV
	Unknown					
Trea	tmentRecom	mendeo	ł			

Radical radiotherapy

High dose palliative radiotherapy

Palliative radiotherapy

Chemotherapy

Specialist palliative care referral